Some characterizations of function spaces around mixed Morrey spaces

Toru Nogayama

Contents

A	Acknowledgement					
1	Introduction					
	1.1	Mixed	Morrey spaces	5		
	1.2	Three	topics and main theorems	7		
		1.2.1	Main theorem on the boundedness of commutators generated by BMO functions and fractional integral operators on mixed Mor- rey spaces	7		
		1.2.2	Main theorems on the characterization of Morrey spaces associ- ated with Banach lattice in terms of pointwise multiplier spaces	8		
		1.2.3	Main theorems on the atomic decomposition for mixed Morrey spaces	11		
	1.3	Notati	on	13		
2	Preliminaries					
	2.1	Mixed	Morrey spaces and related spaces	15		
		2.1.1	Classical Morrey spaces and fundamental results	15		
		2.1.2	Mixed Lebesgue spaces and fundamental results	18		
		2.1.3	Mixed Morrey spaces and some results	20		
		2.1.4	Predual spaces of mixed Morrey spaces	24		
2.2 Other functi			function spaces	30		
		2.2.1	Besov spaces	30		
		2.2.2	Microlocal Besov spaces	33		
		2.2.3	Lorentz spaces	34		

		2.2.4	Orlicz spaces	36			
3	Bou Moi	Boundedness of commutators of fractional integral operators on mixed Aorrey spaces					
	3.1	Introd	uction and theorems	38			
	3.2	Sharp	maximal inequality	40			
		3.2.1	Preliminaries	40			
		3.2.2	Sharp maximal inequality on mixed Morrey spaces	42			
		3.2.3	The relation to commutators and the sharp maximal function on mixed Morrey spaces	48			
	3.3	Proof	of the main theorem (Theorem 3.1.4) $\ldots \ldots \ldots \ldots \ldots$	54			
4	A c in t	erization of Morrey spaces associated with Banach lattice f pointwise multiplier spaces	56				
	4.1	Introd	uction and theorems	56			
	4.2	Proofs	of the main theorems	59			
		4.2.1	Pointwise multipliers from generalized 2-microlocal Besov spaces to Banach lattices (Theorem 4.1.4)	59			
		4.2.2	Pointwise multipliers from Besov spaces to Lebesgue spaces and Morrey spaces (Theorems 4.1.5 and 4.1.6)	61			
	4.3 Examples		oles	63			
		4.3.1	Pointwise multipliers from Besov spaces to Orlicz spaces	63			
		4.3.2	Pointwise multipliers from Besov spaces to Lorentz spaces	64			
		4.3.3	Pointwise multipliers from Besov spaces to mixed Lebesgue spaces	66			
		4.3.4	Pointwise multipliers from Besov spaces to mixed Morrey spaces	67			
5	Ato	Atomic decomposition for mixed Morrey spaces					
	5.1	Introd	uction and theorems	69			
	5.2	The boundedness of the maximal operator $M^{(\vec{t})}$ (Theorem 5.1.1) 7					
	5.3	Some	observations of Hardy-mixed Morrey spaces	77			
	5.4	Proofs	of the main theorems (Theorems 5.1.2, 5.1.3, 5.1.7, and 5.1.8)	80			
		5.4.1	Reconstruction of mixed Morrey spaces by atoms (Theorem 5.1.2)	80			

Bibliography						
5.5	Application : Olsen's inequality on mixed Morrey spaces $\ldots \ldots \ldots$					
	5.4.3	Atomic decomposition for mixed Morrey spaces and Hardy- mixed Morrey spaces (Theorems $5.1.3$ and $5.1.8$)	86			
	5.4.2	Reconstruction of Hardy-mixed Morrey spaces by atoms (Theo- rem 5.1.7)	81			

Acknowledgement

I would like to express my deep appreciation to my academic advisor Professor Yoshihiro Sawano. Thanks to his constant support and thoughtful guidance, the author could complete the PhD course. The author would also like to be deeply grateful to my supervisor Professor Kazuhiro Kurata. Without his guidance and help, this thesis would not have been possible. I would like to show my appreciation to all the staffs in the Department of Mathematics, Tokyo Metropolitan University for giving me great lectures, warm support and encouragement. I am grateful to Professors Mitsuo Izuki, Takahiro Noi, Hiroki Saito, and Professor Hitoshi Tanaka for many useful discussions.

I would also like to thank Dr. Denny Ivanal Hakim, Dr. Naoya Hatano, Dr. Takahiro Ono, Dr. Daniel Salim and all the members of graduate students in the Department of Mathematics, Tokyo Metropolitan University for many useful discussion and many supports.

Finally, I would like to express my gratitude to my family and friends for their constant support and encouragement.

Chapter 1

Introduction

1.1 Mixed Morrey spaces

Function spaces collect functions having the same properties such as continuity, differentiability, and integrability, and so on. By characterizing the properties of function spaces, many fields of mathematics, for example, Fourier analysis and PDE, etc., have been developed. One of the most fundamental function spaces is the Lebesgue space $L^p(\mathbb{R}^n)$. For $0 , we define the <math>L^p$ norm $\|\cdot\|_{L^p}$ by

$$||f||_{L^p} = ||f||_p = \left(\int_{\mathbb{R}^n} |f(x)|^p \mathrm{d}x\right)^{\frac{1}{p}} \quad (0$$

where f is a measurable function on \mathbb{R}^n . If $p = \infty$, we interpret this expression as

$$||f||_{L^{\infty}} = ||f||_{\infty} = \operatorname{ess\,sup}_{x \in \mathbb{R}^n} |f(x)|.$$

The Lebesgue space $L^p(\mathbb{R}^n)$ is the set of all measurable functions f for which $||f||_p < \infty$. Based on this space, the theory of analysis has advanced remarkably. However, integrability is not enough to describe sufficiently many properties of functions we require. Thus, function spaces having more fine properties were needed, and so during the 20th century many authors introduced a lot of function spaces such as Sobolev spaces, Orlicz spaces, Morrey spaces, Lorentz spaces, Hardy spaces, mixed Lebesgue spaces, Besov spaces, Triebel–Lizorkin spaces, and so on.

One of the important function spaces in this thesis is the Morrey space $\mathcal{M}_q^p(\mathbb{R}^n)$. This is defined as follows: Let $0 < q \leq p < \infty$. Define the *Morrey norm* $\|\cdot\|_{\mathcal{M}_q^p}$ by

$$\|f\|_{\mathcal{M}^p_q} \equiv \sup\left\{|Q|^{\frac{1}{p}-\frac{1}{q}} \left(\int_Q |f(x)|^q \,\mathrm{d}x\right)^{\frac{1}{q}} : Q \text{ is a cube in } \mathbb{R}^n\right\}$$

for a measurable function f. The Morrey space $\mathcal{M}^p_q(\mathbb{R}^n)$ is the set of all measurable functions f for which $||f||_{\mathcal{M}^p_q}$ is finite.

Morrey spaces were introduced by C.B.Morrey Jr. in 1938 to investigate the local behavior of solutions to second order elliptic partial differential equation [97]. Morrey spaces cover Lebesgue spaces so that Morrey spaces describe nice properties more than Lebesgue spaces. For instance, Morrey spaces can handle the function $|x|^{-\frac{n}{p}}$ which fails to belong to Lebesgue spaces. We recall the fundamental properties of this space and the boundedness results for basic operators in harmonic analysis in Subsection 2.1.1.

Meanwhile, in [11], Benedek and Panzone introduced mixed Lebesgue spaces and investigated some properties in 1961. Let $\vec{p} = (p_1, \ldots, p_n) \in (0, \infty]^n$. Then define the mixed Lebesgue norm $\|\cdot\|_{\vec{p}}$ or $\|\cdot\|_{(p_1,\ldots,p_n)}$ by

$$|f||_{L^{\vec{p}}} = ||f||_{\vec{p}} = ||f||_{(p_1,\dots,p_n)}$$
$$\equiv \left(\int_{\mathbb{R}} \cdots \left(\int_{\mathbb{R}} \left(\int_{\mathbb{R}} |f(x_1, x_2, \dots, x_n)|^{p_1} \mathrm{d}x_1\right)^{\frac{p_2}{p_1}} \mathrm{d}x_2\right)^{\frac{p_3}{p_2}} \cdots \mathrm{d}x_n\right)^{\frac{1}{p_n}},$$

where $f : \mathbb{R}^n \to \mathbb{C}$ is a measurable function. If $p_j = \infty$, then we have to make appropriate modifications. We define the *mixed Lebesgue space* $L^{\vec{p}}(\mathbb{R}^n) = L^{(p_1,\ldots,p_n)}(\mathbb{R}^n)$ to be the set of all measurable functions f on \mathbb{R}^n with $||f||_{\vec{p}} < \infty$.

Since functions belonging to mixed Lebesgue spaces have the different integrability for each direction, we expect that they can characterize functions more subtly. In Subsection 2.1.2, we summarize their properties and the classical results for this space.

The author defined the mixed Morrey space $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ combining the Morrey space $\mathcal{M}^p_q(\mathbb{R}^n)$ and the mixed Lebesgue space $L^{\vec{q}}(\mathbb{R}^n)$ in [106].

Let $\vec{q} = (q_1, \ldots, q_n) \in (0, \infty]^n$ and $p \in (0, \infty)$ satisfy

$$\sum_{j=1}^n \frac{1}{q_j} \ge \frac{n}{p}.$$

Then define the mixed Morrey norm $\|\cdot\|_{\mathcal{M}^p_{\vec{\sigma}}(\mathbb{R}^n)}$ by

$$||f||_{\mathcal{M}^{p}_{\vec{q}}} \equiv \sup\left\{ |Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)} ||f\chi_{Q}||_{\vec{q}} : Q \text{ is a cube in } \mathbb{R}^{n} \right\}$$

for all measurable functions f on \mathbb{R}^n . We define the *mixed Morrey space* $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ to be the set of all measurable function f on \mathbb{R}^n satisfying $\|f\|_{\mathcal{M}^p_{\vec{q}}} < \infty$.

In [106], the author investigated the basic properties (for example, embedding, completeness) and the mapping properties of the Hardy–Littlewood maximal operator, fractional integral operators, and singular integral operators. We recall these properties in Subsection 2.1.3.

In this thesis, based on the studies in [106], we summarize the further studies of mixed Morrey spaces and the related spaces in [55, 106, 107, 108]. The next section is devoted to a summary of three topics and main theorems treated in each chapter.

1.2 Three topics and main theorems

From Chapter 3 to Chapter 5, we treat three topics related to mixed Morrey spaces. In this section, we give an overview of these topics and main results for each topic. First, we introduce the idea of commutators generated by functions and operators and consider the boundedness results for commutators on mixed Morrey spaces in Subsection 1.2.1. Subsection 1.2.2 is devoted to giving the concept of pointwise multiplier spaces, and we characterize them in terms of Morrey spaces. Finally, in Subsection 1.2.3, we consider the characterization by atomic decomposition for mixed Morrey spaces.

1.2.1 Main theorem on the boundedness of commutators generated by BMO functions and fractional integral operators on mixed Morrey spaces

The idea of commutators generated by functions and operators was introduced by Coifman, Rochberg, and Weiss [20] and many authors investigated the boundedness and compactness results for these operators. In particular, we treat commutators generated by functions and fractional integral operators. Let $0 < \alpha < n$. Define the fractional integral operator I_{α} of order α by

$$I_{\alpha}f(x) \equiv \int_{\mathbb{R}^n} \frac{f(y)}{|x-y|^{n-\alpha}} \mathrm{d}y$$

for $f \in L^1_{loc}(\mathbb{R}^n)$ as long as the right-hand side makes sense. The commutator $[a, I_\alpha]$ is given by

$$[a, I_{\alpha}](f)(x) = a(x)I_{\alpha}f(x) - I_{\alpha}(af)(x) = \int_{\mathbb{R}^n} \frac{a(x) - a(y)}{|x - y|^{n - \alpha}} f(y) \mathrm{d}y$$

for a measurable function a and $x \in \mathbb{R}^n$ as long as the integral makes sense. Moreover, to describe our main theorem, we recall the BMO class. Define the BMO norm by

$$||f||_{BMO} = \sup_{Q} \frac{1}{|Q|} \int_{Q} |f(y) - f_{Q}| dy$$

for $f \in L^1_{\text{loc}}(\mathbb{R}^n)$, where

$$f_Q = \frac{1}{|Q|} \int_Q f(y) \mathrm{d}y$$

and the supremum is taken over all cubes Q in \mathbb{R}^n . Then, $BMO(\mathbb{R}^n)$ is the set of all functions f modulo constants satisfying $||f||_{BMO} < \infty$.

We recall the classical results on the boundedness of commutators on Morrey spaces. In 1991, Di Fazio and Ragusa [24] gave the necessary and sufficient condition on a function b for the boundedness of commutator $[b, I_{\alpha}]$ on Morrey spaces. Although there was a little additional assumption in their result, Shirai removed it in [140]. Our aim in Chapter 3 is to generalize these results to mixed Morrey spaces. Here we state our main theorem in Chapter 3. **Theorem 1.2.1.** Let $0 < \alpha < n, 1 < p < \frac{n}{\alpha}, \frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$, and $\frac{n}{r} \le \sum_{j=1}^{n} \frac{1}{s_j}$. Also, assume that

 $\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{q_j}{p} = \frac{s_j}{r} \quad (j = 1, \dots, n).$

Then, the following conditions are equivalent:

- (a) $b \in BMO(\mathbb{R}^n)$.
- (b) $[b, I_{\alpha}]$ is bounded from $\mathcal{M}^{p}_{\vec{a}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{\vec{s}}(\mathbb{R}^{n})$.
- (c) $[b, I_{\alpha}]$ is bounded from $\widetilde{\mathcal{M}}^{p}_{\vec{q}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{\vec{s}}(\mathbb{R}^{n})$.
- (d) $[b, I_{\alpha}]$ is bounded from $\widetilde{\mathcal{M}}^{p}_{\vec{a}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{1}(\mathbb{R}^{n})$.

Here, $\widetilde{\mathcal{M}}^{p}_{\vec{a}}(\mathbb{R}^{n})$ is the $\mathcal{M}^{p}_{\vec{a}}(\mathbb{R}^{n})$ -closure of $C^{\infty}_{c}(\mathbb{R}^{n})$.

We compare our results with the classical ones. Usually, when we handle commutators, the sharp maximal operator is a useful tool as was done in [24, 129, 130]. The sharp maximal operator, which is defined in [34], is a good operator to control the singularity of the integral operators. To control the sharp maximal operator, we use the so-called good λ -inequality described in [141]. However, the layer cake formula, which is also described in [141], is not available in the mixed-norm setting. So we make use of the dyadic local sharp maximal operator defined in [84] together with a key formula [84, Theorem 2.2]. By using these ingredients, we established the estimate for the sharp maximal operator on mixed Morrey spaces in Section 3.2. In Section 3.3, we give the proof of Theorem 1.2.1. We note that our method does not employ the predual spaces of Morrey spaces which were used in the previous works ([24, 140]).

1.2.2 Main theorems on the characterization of Morrey spaces associated with Banach lattice in terms of pointwise multiplier spaces

Given Banach spaces $E_1(\mathbb{R}^n)$ and $E_2(\mathbb{R}^n)$ of measurable functions defined on \mathbb{R}^n , we define PWM $(E_1(\mathbb{R}^n), E_2(\mathbb{R}^n))$ as follows: A measurable function g is a *pointwise multiplier* from $E_1(\mathbb{R}^n)$ to $E_2(\mathbb{R}^n)$ if the pointwise product $f \cdot g$ belongs to $E_2(\mathbb{R}^n)$ for each $f \in E_1(\mathbb{R}^n)$ and there exists a constant M > 0 such that

$$\|f \cdot g\|_{E_2(\mathbb{R}^n)} \le M \|f\|_{E_1(\mathbb{R}^n)}.$$
(1.1)

One defines a norm on $\text{PWM}(E_1(\mathbb{R}^n), E_2(\mathbb{R}^n))$ by

$$||g||_{\text{PWM}(E_1, E_2)} \equiv \inf\{M > 0 : (1.1) \text{ holds for all } f \in E_1(\mathbb{R}^n)\}$$

for $g \in \text{PWM}(E_1(\mathbb{R}^n), E_2(\mathbb{R}^n))$.

A typical example is

$$PWM(L^{p}(\mathbb{R}^{n}), L^{1}(\mathbb{R}^{n})) = L^{p'}(\mathbb{R}^{n}),$$

where $p' = \frac{p}{p-1}$ is a conjugate exponent of p. This example is easily obtained from the Hölder inequality.

In [82], to investigate the solution of the Navier–Stokes equation, Lemarié-Rieusset used that pointwise multiplier spaces from Besov spaces to Lebesgue spaces coincide Morrey spaces.

Theorem 1.2.2 (cf. [82]). Let $1 \le p < \infty$ and $0 < s \le \frac{n}{p}$. Then $\operatorname{PWM}(\dot{B}^s_{p1}(\mathbb{R}^n), L^p(\mathbb{R}^n)) \approx \mathcal{M}_p^{\frac{n}{s}}(\mathbb{R}^n)$

with equivalence of norms.

Lemarié-Rieusset obtained Theorem 1.2.2 for n = 3 and p = 2 [82, Lemma 6]. A passage to the general case is a minor modification, so that we give a proof in Subsection 4.2.2.

Our main result generalizes this one by replacing Lebesgue spaces with abstract Banach lattices. Recall that a *Banach* (function) lattice on \mathbb{R}^n is a Banach space $(E, \|\cdot\|_E)$ contained in the linear space of all measurable functions, such that, for all $f, g \in E$, the implication " $|f| \leq |g| \Rightarrow ||f||_E \leq ||g||_E$ " holds. Then, we have also to generalize Morrey spaces and Besov spaces. To simplify the discussion, in this subsection and Chapter 4, we let $E(\mathbb{R}^n)$ be a Banach lattice be such that $||f(\cdot - x)||_E =$ $||f||_E$ for all $f \in E(\mathbb{R}^n)$ and $x \in \mathbb{R}^n$.

Based on [58, Definition 2.6], we define $\mathcal{M}^p_E(\mathbb{R}^n)$ to be the set of all measurable functions f for which

$$\|f\|_{\mathcal{M}_E^p} \equiv \sup_Q |Q|^{\frac{1}{p}} \left(\frac{1}{\|\chi_Q\|_E} \|f\chi_Q\|_E\right)$$

is finite, where Q moves over all cubes whose edges are parallel to the coordinate axes. Note that if we take $E(\mathbb{R}^n) = L^q(\mathbb{R}^n)$, $\mathcal{M}^p_E(\mathbb{R}^n)$ coincides with the classical Morrey space $\mathcal{M}^p_q(\mathbb{R}^n)$.

Next, we define 2-microlocal Besov spaces. First, we recall the class $\mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$:

Definition 1.2.3 (Weight class $\mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$). Let $\alpha_1, \alpha_2, \alpha_3 \in [0, \infty)$. The class $\mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$ of weights is defined as the set of all the sequences of the measurable functions $w = \{w_j\}_{j=-\infty}^{\infty}$ satisfying the following conditions:

1. There exists a constant C > 0 such that for all $x, y \in \mathbb{R}^n$ and $j \in \mathbb{Z}$,

$$0 < w_j(x) \le Cw_j(y)(1+2^j|x-y|)^{\alpha_3}$$

2. For all $x \in \mathbb{R}^n$ and $j \in \mathbb{Z}$,

$$2^{-\alpha_1} w_j(x) \le w_{j+1}(x) \le 2^{\alpha_2} w_j(x).$$

Such a sequence $w \in \mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$ is called an admissible weight sequence.

The (homogeneous) generalized 2-microlocal Besov spaces are usually defined by the use of the Fourier transform as follows. For $f \in L^1(\mathbb{R}^n)$, define its Fourier transform and its inverse Fourier transform by

$$\mathcal{F}f(\xi) \equiv (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} f(x) \mathrm{e}^{-\mathrm{i}x \cdot \xi} \mathrm{d}x, \quad \mathcal{F}^{-1}f(x) \equiv (2\pi)^{-\frac{n}{2}} \int_{\mathbb{R}^n} f(\xi) \mathrm{e}^{\mathrm{i}x \cdot \xi} \mathrm{d}\xi.$$

By a well-known method, we can extend $\mathcal{F}, \mathcal{F}^{-1}$ naturally to the Schwartz distribution space $\mathcal{S}'(\mathbb{R}^n)$.

Definition 1.2.4 (Generalized 2-microlocal Besov spaces). Let $w \in \mathcal{W}^{\alpha_3}_{\alpha_1,\alpha_2}$. Let $\varphi \in C^{\infty}_{c}(\mathbb{R}^n)$ satisfy

$$\chi_{B(4)\setminus B(2)} \le \varphi \le \chi_{B(8)\setminus B(\frac{3}{2})}$$

and define $\varphi_j(x) = \varphi(2^{-j}x)$. Let $0 < p, q \leq \infty$. Then for $f \in \mathcal{S}'(\mathbb{R}^n)$ define

$$\|f\|_{\dot{B}^{s,\mathrm{mloc}}_{pq}(\mathbb{R}^n,w)} \equiv \left(\sum_{j=-\infty}^{\infty} 2^{js} \|w_j \mathcal{F}^{-1}[\varphi_j \mathcal{F}f]\|_p^q\right)^{\frac{1}{q}}.$$

The generalized 2-microlocal Besov space $\dot{B}_{pq}^{s,\text{mloc}}(\mathbb{R}^n, w)$ is the set of all $f \in \mathcal{S}'(\mathbb{R}^n)$ for which $\|f\|_{\dot{B}_{pq}^{s,\text{mloc}}(\mathbb{R}^n,w)}$ is finite.

Note that if we take $w_j(x) = 1$ for all $x \in \mathbb{R}^n$ and $j \in \mathbb{Z}$, we obtain classical Besov spaces.

To state our result, also recall the definition of dyadic cubes. For $j \in \mathbb{Z}$ and $m \equiv (m_1, m_2, \ldots, m_n) \in \mathbb{Z}^n$, we define $Q_{jm} \equiv \prod_{j=1}^n \left[\frac{m_j}{2^j}, \frac{m_j+1}{2^j}\right)$. Denote by $\mathcal{D} = \mathcal{D}(\mathbb{R}^n)$ the set of such cubes. The elements in \mathcal{D} are called *dyadic cubes*.

Our main result in Chapter 4 is the following assertion.

Theorem 1.2.5. Let $E(\mathbb{R}^n)$ and $F(\mathbb{R}^n)$ be Banach lattices such that

$$\|\chi_{Q_{jm}}\|_F \lesssim \|\chi_{Q_{jm}}\|_E |Q_{j0}|^{-\frac{1}{p}} \quad (j \in \mathbb{Z}, m \in \mathbb{Z}^n).$$

Set

$$w_j \equiv \|\chi_{Q_{j0}}\|_E |Q_{j0}|^{-1-\frac{1}{p}} \quad (j \in \mathbb{Z}).$$

Then $\dot{B}_{11}^{0,\mathrm{mloc}}(\mathbb{R}^n,w)$ is continuously embedded into $F(\mathbb{R}^n)$ and

$$\mathrm{PWM}(\dot{B}_{11}^{0,\mathrm{mloc}}(\mathbb{R}^n,w),E(\mathbb{R}^n)) \approx \mathcal{M}_E^p(\mathbb{R}^n)$$

with equivalence of norms.

In Section 4.2, we give the proofs of our results. Lemarié-Rieusset used the wavelet decomposition to show the corresponding result since he considered the assertion based on $L^2(\mathbb{R}^n)$. Meanwhile for the proof of Theorem 1.2.5, we employ the atomic decomposition for classical Besov spaces and 2-microlocal Besov spaces. Section 4.3 is devoted to apply our result for various function spaces, which are Orlicz spaces (Subsection 4.3.1), Lorentz spaces (Subsection 4.3.2), mixed Lebesgue spaces (Subsection 4.3.3), and mixed Morrey spaces (Subsection 4.3.4). In addition, the definition of each function space is given in Section 2.2.

1.2.3 Main theorems on the atomic decomposition for mixed Morrey spaces

One of the characterization methods of function spaces is to decompose functions or distributions into linear sums of elementary ones. In Chapter 5, we discuss the decomposition results for mixed Morrey spaces. In particular, we concentrate on the decomposition by the atom, which is a function with a compact support, a suitable norm estimate, and the moment condition. The decomposition results by atoms for classical Morrey spaces were proved by Iida, Sawano, and Tanaka in [66].

Denote by $\mathcal{Q}(\mathbb{R}^n)$ the set of all cubes in \mathbb{R}^n . Our first result is the following construction result about the functions in mixed Morrey spaces.

Theorem 1.2.6. Suppose that the parameters p, \vec{q}, s, \vec{t} satisfy

$$1$$

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}, \quad \frac{n}{s} \le \sum_{j=1}^{n} \frac{1}{t_j}$$

Assume that $\{a_j\}_{j=1}^{\infty} \subset \mathcal{M}_{\vec{t}}^s(\mathbb{R}^n), \ \{\lambda_j\}_{j=1}^{\infty} \subset [0,\infty), \ and \ \{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n) \ fulfill$

$$||a_j||_{\mathcal{M}^s_{\vec{t}}} \leq |Q_j|^{\frac{1}{s}}, \quad \operatorname{supp}(a_j) \subset Q_j, \quad \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}} < \infty.$$

Then $f = \sum_{j=1}^{\infty} \lambda_j a_j$ converges in $\mathcal{S}'(\mathbb{R}^n) \cap L^{\vec{q}}_{\text{loc}}(\mathbb{R}^n)$ and satisfies

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \le C_{p,\vec{q},s,\vec{t}} \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

The next assertion concerns the decomposition of functions in $\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$. For $d \geq 0$, denote by $\mathcal{P}_{d}(\mathbb{R}^{n})$ the set of all polynomial functions with degree less than or equal

to d, so that $\mathcal{P}(\mathbb{R}^n) \equiv \bigcup_{d=0}^{\infty} \mathcal{P}_d(\mathbb{R}^n)$. It is clear that $\mathcal{P}_{-1}(\mathbb{R}^n) = \{0\}$. The set $\mathcal{P}_K(\mathbb{R}^n)^{\perp}$ denotes the set of measurable function f for which

$$(1+|\cdot|^2)^{\frac{K}{2}}f \in L^1(\mathbb{R}^n)$$
 and $\int_{\mathbb{R}^n} x^{\alpha}f(x)dx = 0$

for any $\alpha \in (\mathbb{N} \cup \{0\})^n$ with $|\alpha| \leq K$. Such a function f is said to satisfy the moment condition of order K. In this case, one also writes $f \perp \mathcal{P}_K(\mathbb{R}^n)$.

Theorem 1.2.7 (cf. [60]). Suppose that the real parameters p, \vec{q}, K satisfy

$$1$$

where $q_0 = \min(q_1, \ldots, q_n)$. Let $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then there exists a triplet $\{a_j\}_{j=1}^{\infty} \subset L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}^{\perp}_K(\mathbb{R}^n)$, $\{\lambda_j\}_{j=1}^{\infty} \subset [0, \infty)$, and $\{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n)$ such that $f = \sum_{j=1}^{\infty} \lambda_j a_j$ in $\mathcal{S}'(\mathbb{R}^n)$ and that, for any v > 0

$$|a_j| \le \chi_{Q_j}, \quad \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^v \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}} \le C_v \|f\|_{\mathcal{M}^p_{\vec{q}}}.$$

Here the constant $C_v > 0$ is independent of f.

Note that applying Theorems 1.2.6 and 1.2.7 for $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, we obtain norm estimate

$$C^{-1} \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}} \le \|f\|_{\mathcal{M}^p_{\vec{q}}} \le C \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}}$$

for some C > 1. Here λ_j and Q_j are same ones which appear in Theorem 1.2.7.

Although Theorem 1.2.7 can be given as a corollary of the abstract results in [60], we can give a direct proof without using Herz spaces which were used in the abstract setting of [60].

Moreover, we can prove the general decomposition theorems using Hardy-mixed Morrey spaces (Theorems 5.1.7 and 5.1.8 to follow). We can show that Hardy-mixed Morrey spaces coincide with mixed Morrey spaces for $\vec{q} > 1$ (Proposition 5.3.1). Therefore, Theorems 5.1.7 and 5.1.8 include Theorems 1.2.6 and 1.2.7, respectively.

As an application, we show the Olsen inequality for the fractional integral operator I_{α} acting on mixed Morrey spaces.

Theorem 1.2.8. Suppose that the parameters α , p, \vec{q} , p^* , \vec{q}^* , s, \vec{t} satisfy

$$1 < p, p^*, s < \infty, \quad 1 < \vec{q}, \vec{q}^*, \vec{t} < \infty$$

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}, \quad \frac{n}{p^*} \le \sum_{j=1}^{n} \frac{1}{q_j^*}, \quad \frac{n}{s} \le \sum_{j=1}^{n} \frac{1}{t_j},$$
$$\max\{t_1, \dots, t_j\} < q_j^*, \quad \frac{1}{p} > \frac{\alpha}{n}, \quad \frac{1}{p^*} \le \frac{\alpha}{n},$$

for each $j = 1, 2, \ldots, n$, and that

$$\frac{1}{s} = \frac{1}{p^*} + \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{t_j}{s} = \frac{q_j}{p} \quad (j = 1, 2, \dots, n).$$

Then for all $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ and $g \in \mathcal{M}^{p^*}_{\vec{q}^*}(\mathbb{R}^n)$

$$\|g \cdot I_{\alpha}f\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \cdot \|f\|_{\mathcal{M}^{p}_{\vec{q}}},$$

where the constant C is independent of f and g.

As the special case of $q_i = q$ and $t_i = t$ for all i = 1, ..., n, this result recaptures the one for classical Morrey spaces [134, Proposition 1.8].

In Section 5.2, we establish the boundedness result for the maximal operator to prove Theorem 1.2.6. We observe the characterization of Hardy-Morrey spaces in Section 5.3. Section 5.4 is devoted to the proof of the main theorems. In Subsections 5.4.1 and 5.4.2, we consider the reconstruction theorems for mixed Morrey spaces and Hardy-mixed Morrey spaces, respectively. Meanwhile in Subsection 5.4.3, we prove the decomposition theorems for mixed Morrey spaces. At last, in Section 5.5, we establish Olsen's inequality for mixed Morrey spaces by applying the atomic decompositions.

1.3 Notation

Throughout this thesis, we use the following notation.

- 1. The letters $\vec{p}, \vec{q}, \vec{r}, \ldots$ denote the *n*-tuples of the numbers in $[0, \infty]$ $(n \ge 1)$, that is, $\vec{p} = (p_1, \ldots, p_n), \vec{q} = (q_1, \ldots, q_n), \vec{r} = (r_1, \ldots, r_n).$
- 2. The inequality, for example, $0 < \vec{p} < \infty$ means that $0 < p_j < \infty$ for each $j = 1, \ldots, n$.
- 3. For $\vec{p} = (p_1, \ldots, p_n)$ and $r \in \mathbb{R} \setminus \{0\}$, let

$$\frac{1}{\vec{p}} = \left(\frac{1}{p_1}, \dots, \frac{1}{p_n}\right), \quad \frac{\vec{p}}{r} = \left(\frac{p_1}{r}, \dots, \frac{p_n}{r}\right), \quad \vec{p}' = (p'_1, \dots, p'_n),$$

where
$$p'_j = \frac{p_j}{p_j - 1}$$
 is the conjugate exponent of p_j $(j = 1, ..., n)$.

- 4. Let Q = Q(x, r) be a cube having center x and radius r, whose sides are parallel to the coordinate axes. In particular, if x = 0, then we write Q(r). The symbol Q denotes the set of all cubes.
- 5. Denote by B(x,r) the ball centered at x and radius r > 0. We shall write B(r) = B(0,r) as before.
- 6. The symbols |Q| denotes the volume of the cube Q and $\ell(Q)$ denotes the side length of the cube Q.
- 7. For given a cube Q and k > 0, kQ means a cube concentric to Q with sidelength $k \ell(Q)$.
- 8. For $j \in \mathbb{Z}$ and $m \equiv (m_1, \ldots, m_n) \in \mathbb{Z}^n$, we define $Q_{jm} \equiv \prod_{k=1}^n \left[\frac{m_k}{2^j}, \frac{m_k+1}{2^j} \right)$. Denote by $\mathcal{D} = \mathcal{D}(\mathbb{R}^n)$ the set of such cubes. The elements in \mathcal{D} are called *dyadic cubes*.
- 9. By $A \lesssim B$, we denote that $A \leq CB$ for some constant C > 0, and $A \sim B$ means that $A \lesssim B$ and $B \lesssim A$.
- 10. When we need to emphasize or keep in mind that the constant C depends on the parameter α, β , etc, we write $C = C_{\alpha,\beta}$.
- 11. We write $\mathbb{N}_0 = \mathbb{N} \cup \{0\}$.
- 12. We use " \cdot " for functions; $f = f(\cdot)$. In particular, when we only use " \cdot " for *j*-th coordinate, we write $f = f(\cdot_j)$.
- 13. Let *E* be a measurable set in \mathbb{R}^n . Then, χ_E denotes the characteristic function for *E*.
- 14. Let E be a measurable set in \mathbb{R}^n and f be a measurable function. Then,

$$||f||_{L^p(E)} \equiv ||f\chi_E||_p.$$

15. Let w be a nonnegative measurable function. Then, $\|\cdot\|_{L^p(w)}$ denote the weighted Lebesgue norm, that is, for a measurable function f,

$$||f||_{L^p(w)} \equiv ||fw||_p.$$

- 16. The norm $\|\cdot\|_*$ denote the operator norm.
- 17. We define $L^0(\mathbb{R}^n)$ as the set of all measurable functions on \mathbb{R}^n .
- 18. $\mathcal{S}(\mathbb{R}^n)$ denote the set of all rapidly decreasing functions on \mathbb{R}^n and $\mathcal{S}'(\mathbb{R}^n)$ its dual (that is, the set of all tempered distributions on \mathbb{R}^n).
- 19. Let $L_{c}^{\vec{q}}(\mathbb{R}^{n})$ denote the set of all $L^{\vec{q}}(\mathbb{R}^{n})$ functions with compact support.

Chapter 2

Preliminaries

In this chapter, we provide the definition and some properties of function spaces which are used in this thesis. In Section 2.1, we recall mixed Morrey spaces and their related spaces. First, we review Morrey spaces and mixed Lebesgue spaces in Subsections 2.1.1 and 2.1.2, respectively. After that we summarize the definition and some results of mixed Morrey spaces investigated in [106]. At the end of this section, we consider the predual spaces of mixed Morrey spaces. We use the predual spaces in Chapter 5. Section 2.2 is devoted to introducing some function spaces which we apply to in Section 4. At first, we recall Besov spaces in Subsection 2.2.1. We also consider generalized 2-microlocal Besov spaces which generalize the weighted Besov spaces in Subsection 2.2.2. In the last two subsections, we recall Lorentz spaces and Orlicz spaces, which are other generalizations of Lebesgue spaces.

2.1 Mixed Morrey spaces and related spaces

2.1.1 Classical Morrey spaces and fundamental results

In this subsection, we recall the definition and some properties of the Morrey space $\mathcal{M}_q^p(\mathbb{R}^n)$. For their proofs, we refer to [117].

Definition 2.1.1. Let $0 < q \le p < \infty$. Define the Morrey norm $\|\cdot\|_{\mathcal{M}^p_q}$ by

$$\|f\|_{\mathcal{M}^p_q} \equiv \sup\left\{|Q|^{\frac{1}{p}-\frac{1}{q}} \left(\int_Q |f(x)|^q \,\mathrm{d}x\right)^{\frac{1}{q}} : Q \text{ is a cube in } \mathbb{R}^n\right\}$$

for a measurable function f. The Morrey space $\mathcal{M}^p_q(\mathbb{R}^n)$ is the set of all measurable functions f for which $\|f\|_{\mathcal{M}^p_q}$ is finite.

Remark 2.1.2. We can also define the following norm:

$$\|f\|_{\mathcal{M}^p_q}^{\text{ball}} \equiv \sup\left\{|B|^{\frac{1}{p}-\frac{1}{q}} \left(\int_B |f(x)|^q \,\mathrm{d}x\right)^{\frac{1}{q}} : B \text{ is a ball in } \mathbb{R}^n\right\}.$$

Then, using the fact

$$B(x,r) \subset Q(x,r) \subset B(x,\sqrt{n}r),$$

we see that the norms $||f||_{\mathcal{M}_q^p}$ and $||f||_{\mathcal{M}_q^p}^{\text{ball}}$ are equivalent. Thus, we will use the suitable one and denote it by the same notation $||\cdot||_{\mathcal{M}_q^p}$.

First of all, we point out the fundamental properties of Morrey spaces.

Theorem 2.1.3. Let $1 \leq q \leq p < \infty$. Then, the Morrey space $\mathcal{M}_q^p(\mathbb{R}^n)$ is a Banach space.

The next proposition suggests that Morrey spaces are generalizations of Lebesgue spaces.

Proposition 2.1.4. Let $0 . Then, <math>\mathcal{M}_p^p(\mathbb{R}^n) = L^p(\mathbb{R}^n)$.

The relation of two different Morrey spaces is as follows.

Proposition 2.1.5. Let $0 < q_1 \le q_2 \le p < \infty$. Then, we have

$$L^{p}(\mathbb{R}^{n}) = \mathcal{M}_{p}^{p}(\mathbb{R}^{n}) \hookrightarrow \mathcal{M}_{q_{2}}^{p}(\mathbb{R}^{n}) \hookrightarrow \mathcal{M}_{q_{1}}^{p}(\mathbb{R}^{n}).$$

Next, we consider the examples of functions belonging to Morrey spaces.

Example 2.1.6 ([126, Exercise 6.17]). Let Q be a cube in \mathbb{R}^n . Then, for $0 < q < p < \infty$

$$\|\chi_Q\|_{\mathcal{M}^p_a} = |Q|^{\frac{1}{p}}.$$

Example 2.1.7 ([79, Lemma 4.1]). Let $0 < q < p < \infty$. Then, $|x|^{-\frac{n}{p}} \in \mathcal{M}_q^p(\mathbb{R}^n)$.

Note that $|x|^{-\frac{n}{p}}$ does not belong to $L^p(\mathbb{R}^n)$ for $0 . Therefore, the Lebesgue space <math>L^p(\mathbb{R}^n)$ is proper subset of the Morrey space $\mathcal{M}^p_q(\mathbb{R}^n)$.

Finally, we recall the studies of Morrey spaces. Morrey spaces were introduced by C.B.Morrey Jr. in 1938 to investigate the local behavior of solutions to second order elliptic partial differential equation [97]. Later, many authors investigated Morrey spaces. In 1960s, Campanato introduced and studied Campanato spaces which coincide with the many function spaces, the BMO space, Lipschitz spaces, Hölder spaces, and Morrey spaces. Peetre [114] gave a survey of Morrey spaces and Campanato spaces in 1969. In this survey, he also investigated the boundedness of the singular integral operators. Singular integral operators on Morrey spaces have several definitions via preduals and weight theory. See [19, 120, 121]. As the last point in the first development, Adams pointed out that the fractional integral operator I_{α} is bounded on Morrey spaces in [1].

Theorem 2.1.8. Let $0 < \alpha < n, 1 < q \le p < \infty$, and $1 < s \le r < \infty$. Assume that

$$\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{p}{q} = \frac{r}{s}$$

Then, for all $f \in \mathcal{M}^p_q(\mathbb{R}^n)$,

 $\|I_{\alpha}f\|_{\mathcal{M}^r_s} \lesssim \|f\|_{\mathcal{M}^p_q}.$

The next turning point is that Chiarenza and Frasca proved the boundedness of the Hardy–Littlewood maximal operator on Morrey spaces in 1987 [19]. The Hardy– Littlewood maximal operator is one of the most important operators in harmonic analysis. This operator is defined for all measurable functions f and $x \in \mathbb{R}^n$ as

$$Mf(x) = \sup_{Q} \frac{\chi_Q(x)}{|Q|} \int_{Q} |f(y)| \mathrm{d}y,$$

where the supremum is taken over all cubes Q in \mathbb{R}^n . The basic idea of this operator was introduced by Hardy and Littlewood [54] in the language of cricket. Frankly speaking, this operator is taking the largest average of a function over all cubes containing x. The technique of averaging naturally arises in many situations, so it is very significant. The boundedness result of the Hardy–Littlewood maximal operator on Morrey spaces is as follows.

Theorem 2.1.9. Let $1 < q \le p < \infty$. Then

$$\|Mf\|_{\mathcal{M}^p_q} \lesssim \|f\|_{\mathcal{M}^p_q}$$

for all $f \in \mathcal{M}^p_q(\mathbb{R}^n)$.

Using the maximal operator, Di Fazio and Ragusa [24] and Shirai [140] proved commutators generated by BMO functions and the fractional integral operator I_{α} are bounded on Morrey spaces (these statements will appear in Theorems 3.1.2 and 3.1.3).

Concerning the duality, Long proved that the block space $\mathcal{H}_{q'}^{p'}(\mathbb{R}^n)$ is a predual space of the Morrey space $\mathcal{M}_q^p(\mathbb{R}^n)$ in [87]. Zorko characterized the predual space by means of the atomic decomposition in [152]. In [73], Kalita constructed another predual space of Morrey spaces which is the same space as Zorko's spaces with norm equivalence. Furthermore, by using the theory of capacities, the third predual space was defined by Adams and Xiao [2]. See also a recent survey by Rosenthal and Triebel [122].

Furthermore, Morrrey spaces were generalized by many authors in various directions. First of all, we take up generalized Morrey spaces. A definition which are still often used goes back to Zorko's paper [152]. Later, Mizuhara [93], Nakai [98], and Guliyev [43] defined and investigated generalized Morrey spaces, respectively, and many authors studied them in [22, 30, 44, 45, 79, 53]. For an application to partial differential equations, we refer to [3, 23, 78, 86]. See also a survey [127]. Next, we turn to the weight theory of Morry spaces. We have two different definitions of weighted Morrey spaces which are Samko type [123] and Komori–Shirai type [80]. We regard the weight as the one for functions in the former, as the one for measures in the latter. Each of them is investigated in [103, 104] and [65, 139, 151]. Furthermore, both of them appear in the study of partial differential equations [32, 51, 52, 138]. In addition, Morrey spaces for non-doubling measures were defined by Sawano and Tanaka and investigated the boundedness property of some operators [124, 130, 132], and variable exponent Morrey spaces were studied in [4, 48, 49, 94, 95].

2.1.2 Mixed Lebesgue spaces and fundamental results

In this subsection, we recall the mixed Lebesgue space $L^{\vec{p}}(\mathbb{R}^n)$ which is introduced by Benedek and Panzone in [11]. Since the proofs are overall elementary, we omit the details; see [11].

Definition 2.1.10. Let $\vec{p} = (p_1, \ldots, p_n) \in (0, \infty]^n$. Then define the *mixed Lebesgue* norm $\|\cdot\|_{\vec{p}}$ or $\|\cdot\|_{(p_1,\ldots,p_n)}$ by

$$\|f\|_{L^{\vec{p}}} = \|f\|_{\vec{p}} = \|f\|_{(p_1,\dots,p_n)}$$
$$\equiv \left(\int_{\mathbb{R}} \cdots \left(\int_{\mathbb{R}} \left(\int_{\mathbb{R}} |f(x_1, x_2, \dots, x_n)|^{p_1} \mathrm{d}x_1\right)^{\frac{p_2}{p_1}} \mathrm{d}x_2\right)^{\frac{p_3}{p_2}} \cdots \mathrm{d}x_n\right)^{\frac{1}{p_n}},$$

where $f : \mathbb{R}^n \to \mathbb{C}$ is a measurable function. If $p_j = \infty$, then we have to make appropriate modifications. We define the *mixed Lebesgue space* $L^{\vec{p}}(\mathbb{R}^n) = L^{(p_1,\ldots,p_n)}(\mathbb{R}^n)$ to be the set of all measurable functions f on \mathbb{R}^n with $||f||_{\vec{p}} < \infty$.

Note that if each $p_i = p$, then $L^{\vec{p}}(\mathbb{R}^n) = L^p(\mathbb{R}^n)$, so mixed Lebesgue spaces generalize classical spaces. This space has properties similar to classical Lebesgue space.

Proposition 2.1.11. For $1 \leq \vec{p} \leq \infty$, $L^{\vec{p}}(\mathbb{R}^n)$ is a Banach space.

Proposition 2.1.12 (Hölder's inequality). Let $1 < \vec{p}, \vec{q} < \infty$ and define \vec{r} so that $\frac{1}{\vec{p}} + \frac{1}{\vec{q}} = \frac{1}{\vec{r}}$. If $f \in L^{\vec{p}}(\mathbb{R}^n), g \in L^{\vec{q}}(\mathbb{R}^n)$, then $fg \in L^{\vec{r}}(\mathbb{R}^n)$, and

$$\|fg\|_{\vec{r}} \le \|f\|_{\vec{p}} \|g\|_{\vec{q}}.$$

Proposition 2.1.13. Let $0 < \vec{p} \leq \infty$. The mixed Lebesgue norm has the dilation relation: for all $f \in L^{\vec{p}}(\mathbb{R}^n)$ and t > 0,

$$||f(t\cdot)||_{\vec{p}} = t^{-\sum_{j=1}^{n} \frac{1}{p_j}} ||f||_{\vec{p}}.$$

The mapping

$$(x_2, \dots, x_n) \in \mathbb{R}^{n-1} \mapsto ||f||_{(p_1)}(x_2, \dots, x_n) \equiv \left(\int_{\mathbb{R}} |f(x_1, \dots, x_n)|^{p_1} \mathrm{d}x_1\right)^{\frac{1}{p_1}}$$

is a measurable function and defined on \mathbb{R}^{n-1} . Moreover, we define

$$\|f\|_{\vec{q}} = \|f\|_{(p_1,\dots,p_j)} \equiv \left\| \left[\|f\|_{(p_1,\dots,p_{j-1})} \right] \right\|_{(p_j)},$$

where $||f||_{(p_1,\ldots,p_{j-1})}$ denotes |f|, if j = 1 and $\vec{q} = (p_1,\ldots,p_j), j \leq n$. Note that $||f||_{\vec{q}}$ is a measurable function of (x_{j+1},\ldots,x_n) for j < n.

Next, we consider the examples of $L^{\vec{p}}(\mathbb{R}^n)$.

Example 2.1.14. Let Q be a cube. Then, for $0 < \vec{p} \le \infty$,

$$\|\chi_Q\|_{\vec{p}} = |Q|^{\frac{1}{n}(\frac{1}{p_1} + \dots + \frac{1}{p_n})}.$$
(2.1)

This identity is important to consider some inequalities of mixed Morrey spaces.

Example 2.1.15. Let $m = (m_1, \ldots, m_n) \in \mathbb{Z}^n$ and $\{a_m\}_{m \in \mathbb{Z}^n} \subset \mathbb{C}$. Define

$$f(x) = \sum_{m \in \mathbb{Z}^n} a_m \chi_{m+[0,1]^n}(x).$$

Then,

$$||f||_{\vec{p}} = \left(\sum_{m_n \in \mathbb{Z}} \cdots \left(\sum_{m_1 \in \mathbb{Z}} |a_{(m_1, \dots, m_n)}|^{p_1}\right)^{\frac{p_2}{p_1}} \cdots \right)^{\frac{1}{p_n}}.$$
 (2.2)

We can consider the right-hand side of (2.2) as a mixed sequence norm, which computes each ℓ^{p_i} -norm with respect to m_i . We denote it by $\|\{a_m\}_{m\in\mathbb{Z}^n}\|_{\ell^{(p_1,\ldots,p_n)}}$:

$$\|\{a_m\}_{m\in\mathbb{Z}^n}\|_{\ell^{(p_1,\dots,p_n)}} = \|a_{(m_1,\dots,m_n)}\|_{\ell^{(p_1,\dots,p_n)}}$$
$$\equiv \left(\sum_{m_n\in\mathbb{Z}}\cdots\left(\sum_{m_2\in\mathbb{Z}}\left(\sum_{m_1\in\mathbb{Z}}|a_{(m_1,\dots,m_n)}|^{p_1}\right)^{\frac{p_2}{p_1}}\right)^{\frac{p_3}{p_2}}\cdots\right)^{\frac{1}{p_n}}$$

Furthermore, this norm is also defined inductively:

$$\|a_{(m_1,\dots,m_n)}\|_{\ell^{(p_1,\dots,p_j)}} \equiv \left\| \left[\|a_{(m_1,\dots,m_n)}\|_{\ell^{(p_1,\dots,p_{j-1})}} \right] \right\|_{\ell^{(p_j)}}$$

where $||a_{(m_1,...,m_n)}||_{\ell^{(p_1,...,p_{j-1})}} = |a_{(m_1,...,m_n)}|$ if j = 1 and

$$\|a_{(m_1,\dots,m_n)}\|_{\ell^{(p_j)}} \equiv \left(\sum_{m_j \in \mathbb{Z}} |a_{(m_1,\dots,m_n)}|^{p_j}\right)^{\frac{1}{p_j}}$$

for j = 1, ..., n.

We survey the studies of mixed Lebesgue spaces. Benedek and Panzone investigated fundamental properties (completeness, duality, reflexivity, etc.), a counterpart to the Riesz–Thorin interpolation theorem, the boundedness of the fractional integral operator, and so on. After this paper, there are a lot of studies for mixed Lebesgue spaces. In 1975, Bagby showed the boundedness of the Hardy–Littlewood maximal operator for the functions taking values in the mixed Lebesgue spaces [10]. In particular, Stöckert considered that the strong maximal operator is bounded on mixed Lebesgue spaces [144]. The author, in [106], gave another simple proof using the above Bagby's result. See also [35, 57] for this boundedness. Singular integral operators were studied in [35, 142] by Fernandez, and by Stevanov and Torres, respectively. Additionally, there exist many remarkable works which concern Hörmander–Mikhlin theorem [7], multivariate rearrangements [15, 37], the inclusion problem [42], the theory of variable exponents [59], and the interpolation theory [63, 91]. Recently, Huang and Yang summarized a recent series of investigation on function spaces with mixed norms [62].

From the viewpoint of applications, this mixed norm serves to describe decay at infinity for each direction. For example, we consider a bounded measurable function f on \mathbb{R}^3 satisfying

$$|f(x)| \le \frac{C}{|x_1|^{1/10}|x'|^{10}}, \quad |x_1| > 1, |x'| > 1, C > 0$$

for $x = (x_1, x') \in \mathbb{R} \times \mathbb{R}^2$, where $x' = (x_2, x_3)$. Then, $f \in L^p(\mathbb{R}^3)$ if p > 10, that is, the fast decaying directions are completely ignored in the classical Lebesgue norm. So, in this framework of Lebesgue spaces, this function f is identified with the same as a function g satisfying

$$|g(x)| \le \frac{C}{|x|^{1/10}}, \quad |x| > 1, C > 0.$$

Meanwhile, using the mixed Lebesgue norm, we can see $f \in L^{\vec{p}}(\mathbb{R}^n)$ for $p_1 > 10$ and $p_2, p_3 > \frac{1}{10}$. Employing this idea, many authors established and analyzed solutions for partial differential equations, abstract elliptic and parabolic equations, the Navier–Stokes equation, and etc. [28, 112, 113].

2.1.3 Mixed Morrey spaces and some results

In this subsection, based on [106], we collect some properties and boundedness results of fundamental operators in harmonic analysis. All the proofs are in [106].

Definition 2.1.16. Let $\vec{q} = (q_1, \ldots, q_n) \in (0, \infty]^n$ and $p \in (0, \infty]$ satisfy

$$\sum_{j=1}^{n} \frac{1}{q_j} \ge \frac{n}{p}$$

Then define the *mixed Morrey norm* $\|\cdot\|_{\mathcal{M}^p_{\vec{\alpha}}}$ by

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \equiv \sup\left\{ |Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{q_j}\right)} \|f\chi_Q\|_{\vec{q}} : Q \text{ is a cube in } \mathbb{R}^n \right\}$$

for all measurable functions f on \mathbb{R}^n . We define the mixed Morrey space $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ to be the set of all measurable functions f on \mathbb{R}^n satisfying $\|f\|_{\mathcal{M}^p_{\vec{q}}} < \infty$.

First, we remark the relations of Morrey spaces and mixed Lebesgue spaces. If each $q_i = q \in (0, \infty)$, then $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) = \mathcal{M}^p_q(\mathbb{R}^n)$ with coincidence of norm. In particular,

choose $\vec{q} \in (0,\infty]^n$ satisfying $\frac{1}{p} = \frac{1}{n} \sum_{i=1}^n \frac{1}{q_i}$ so that $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) = L^{\vec{q}}(\mathbb{R}^n)$ with coincidence of norm.

Note that Scapellato and Ragusa [137] introduced the same named space "Mixed Morrey space", which is different from our space, essentially.

We give the properties of the *mixed Morrey spaces*. Just as with mixed Lebesgue spaces, we see that mixed Morrey norm has the following dilation relation:

$$\|f(t\cdot)\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})} = t^{-\frac{n}{p}} \|f\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})} \quad (f \in L^{0}(\mathbb{R}^{n}), t > 0),$$
(2.3)

for $\vec{q} \in (0,\infty]^n$ and $p \in (0,\infty]$ with $\frac{n}{p} \leq \sum_{i=1}^n \frac{1}{q_i}$. The embedding properties are as

follows:

Proposition 2.1.17. ([106, Proposition 3.2]) Let $0 < \vec{q} \le \vec{r} \le \infty$, $0 , and assume <math>\frac{1}{r_1} + \cdots + \frac{1}{r_n} \ge \frac{n}{p}$. Then,

$$\mathcal{M}^p_{\vec{r}}(\mathbb{R}^n) \subset \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n).$$

Let us give some examples.

Example 2.1.18. By Example 2.1.7, $f(x) = |x|^{-\frac{n}{p}} \in \mathcal{M}^p_q(\mathbb{R}^n)$ if q < p. Let $\vec{q} =$ (q_1, \ldots, q_n) . Using Proposition 2.1.17, we have

$$\mathcal{M}^{p}_{\widetilde{q}}(\mathbb{R}^{n}) = \mathcal{M}^{p}_{(\underbrace{\widetilde{q}, \ldots, \widetilde{q}})}(\mathbb{R}^{n}) \subset \mathcal{M}^{p}_{\widetilde{q}}(\mathbb{R}^{n}),$$

n times

where $\widetilde{q} = \max(q_1, \ldots, q_n)$. Thus, if $\max(q_1, \ldots, q_n) = \widetilde{q} < p$,

$$f(x) = |x|^{-\frac{n}{p}} \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n).$$

However, the condition

$$\max(q_1, \dots, q_n) = \widetilde{q}$$

 $\frac{n}{p}$

is a sufficient condition but is not a necessary condition for $f(x) = |x|^{-\frac{n}{p}} \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. In fact, consider the case $\vec{s} = (s_1, \underbrace{\infty, \dots, \infty}_{(n-1) \text{ times}})$ and $s_1 < \frac{p}{n}$. Then, by Proposition

2.1.13,

$$\begin{split} \|f\|_{\mathcal{M}^{p}_{\vec{s}}(\mathbb{R}^{n})} &= \sup_{Q=Q(x,r)} |Q(x,r)|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{s_{j}}\right)} \|f\chi_{Q(x,r)}\|_{\vec{s}} \\ &= \sup_{r>0} |Q(0,r)|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{s_{j}}\right)} \|f\chi_{Q(0,r)}\|_{\vec{s}} \\ &= \sup_{r>0} |Q(0,r)|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{s_{j}}\right)} \|f\chi_{Q(0,1)}\|_{\vec{s}} \times r^{\sum_{j=1}^{n} \frac{1}{s_{j}}} \times r^{-1} \\ &\leq \left(\int_{-1}^{1} |x_{1}|^{-\frac{n}{p}s_{1}} \mathrm{d}x_{1}\right)^{\frac{1}{s_{1}}}. \end{split}$$

Since $s_1 < \frac{p}{n}$, $||f||_{\mathcal{M}^p_{\vec{s}}(\mathbb{R}^n)} < \infty$ and $f \in \mathcal{M}^p_{\vec{s}}(\mathbb{R}^n)$. But \vec{s} does not satisfy (2.4).

Example 2.1.19. Let $0 < \vec{q} \leq \infty$ and assume that $q_j < p_j$ if $p_j < \infty$ and that $q_j \leq \infty$ if $p_j = \infty$ (j = 1, ..., n). Let

$$\sum_{j=1}^{n} \frac{1}{p_j} = \frac{n}{p}.$$
(2.5)

Then, we have

$$f(x) = \prod_{j=1}^{n} |x_j|^{-\frac{1}{p_j}} \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n).$$

Furthermore, condition (2.5) is a necessary and sufficient condition for $f(x) = \prod_{j=1}^{n} |x_j|^{-\frac{1}{p_j}}$ to be a member in $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. In fact, let $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ and $f \neq 0$. Applying (2.3), we have

$$\|f(t\cdot)\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})} = t^{-\frac{n}{p}} \|f\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})} \quad (t>0).$$
(2.6)

On the other hand, since $f(tx) = t^{-\sum_{j=1}^{n} \frac{1}{p_j}} f(x)$,

$$\|f(t\cdot)\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})} = t^{-\sum_{j=1}^{n} \frac{1}{p_{j}}} \|f\|_{\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})}.$$
(2.7)

By (2.6) and (2.7), for all t > 0,

$$t^{-\sum_{j=1}^{n} \frac{1}{p_j}} = t^{-\frac{n}{p}}.$$

Thus, we obtain (2.5).

Example 2.1.20. Let Q be a cube and $\vec{q} \in (0, \infty]^n$. Then,

$$\|\chi_Q\|_{\mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)} = |Q|^{\frac{1}{p}}.$$

To check this, put $\sum_{j=1}^{n} \frac{1}{q_j} = \bar{q}$. First, using (2.1), we get

$$\|\chi_Q\|_{\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)} = \sup_{R \in \mathcal{Q}} |R|^{\frac{1}{p} - \frac{\bar{q}}{n}} \|\chi_Q \chi_R\|_{\vec{q}} \ge |Q|^{\frac{1}{p} - \frac{\bar{q}}{n}} \|\chi_Q\|_{\vec{q}} = |Q|^{\frac{1}{p} - \frac{\bar{q}}{n}} |Q|^{\frac{\bar{q}}{n}} = |Q|^{\frac{1}{p}}.$$

Meanwhile, thanks to Proposition 2.1.17,

$$\|\chi_Q\|_{\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)} \le \|\chi_Q\|_{\mathcal{M}^p_{\max(q_1,\dots,q_n)}(\mathbb{R}^n)} = |Q|^{\frac{1}{p}}.$$

Combining the above two inequalities, we obtain

$$\|\chi_Q\|_{\mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)} = |Q|^{\frac{1}{p}}.$$

We turn to the boundedness results for some operators on mixed Morrey spaces. First, we consider the Hardy–Littlewood maximal operator. The boundedness of the Hardy–Littlewood maximal operator in classical Morrey spaces is showed by Chiarenza and Frasca in 1987 [19] (see Theorem 2.1.9 above).

Theorem 2.1.21 ([106, Theorem 4.5]). Let $1 < \vec{q} < \infty$ and 1 satisfy $<math>\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$. Then $\|Mf\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|f\|_{\mathcal{M}^p_{\vec{q}}}$

for all $f \in \mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)$.

Next, we give the boundedness result of the fractional integral operator I_{α} . Its boundedness in classical Morrey spaces is proved by Adams [1] (see, Theorem 2.1.8).

Theorem 2.1.22 ([106, Theorem 1.11]). Let
$$0 < \alpha < n, 1 < \vec{q}, \vec{s} < \infty$$
 and $1 < p, r < \infty$. Assume that $\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$ and $\frac{n}{r} \leq \sum_{j=1}^{n} \frac{1}{s_j}$. Also, assume that
 $\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{\vec{q}}{p} = \frac{\vec{s}}{r}.$
Then, for $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$,
 $\|I_{\alpha}f\|_{\mathcal{M}^r_{\vec{s}}} \lesssim \|f\|_{\mathcal{M}^p_{\vec{q}}}.$

Finally, we recall the boundedness results for singular integral operators. A singular integral operator T with a kernel k(x, y) is defined as an L^2 -bounded operator which satisfies the following conditions:

- (1) There exists a constant C > 0 such that $|k(x,y)| \le \frac{C}{|x-y|^n}$.
- (2) There exist $\varepsilon > 0$ and C > 0 such that

$$|k(x,y) - k(z,y)| + |k(y,x) - k(y,z)| \le C \frac{|x-z|^{\varepsilon}}{|x-y|^{n+\varepsilon}},$$

 $\text{if } |x-y| \geq 2|x-z| \text{ with } x \neq y.$

(3) If $f \in L^{\infty}_{c}(\mathbb{R}^{n})$, the set of all compactly supported L^{∞} -functions, then

$$Tf(x) = \int_{\mathbb{R}^n} k(x, y) f(y) dy \quad (x \notin \operatorname{supp}(f)).$$

Keeping in mind that T extends to a bounded linear operator on $\mathcal{M}_q^p(\mathbb{R}^n)$ [19], we obtain the following theorem.

Theorem 2.1.23 ([106, Theorem 1.12]). Let $1 < \vec{q} < \infty$ and 1 satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}.$$

Then, if we restrict T to $\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$, which is initially defined on $\mathcal{M}^{p}_{\min(q_{1},\ldots,q_{n})}(\mathbb{R}^{n})$,

$$\|Tf\|_{\mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)} \lesssim \|f\|_{\mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)}$$

for $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

At last in this subsection, we describe the Fefferman–Stein vector-valued inequality on mixed Morrey spaces. This inequality was first considered by Fefferman and Stein in [33].

Proposition 2.1.24 ([106, Theorem 1.8]). Let $1 < \vec{q}, p < \infty, \frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$, and $1 < r \le \infty$. Then

$$\left\| \left(\sum_{j=1}^{\infty} (Mf_j)^r \right)^{1/r} \right\|_{\mathcal{M}^p_{\vec{\alpha}}} \le C \left\| \left(\sum_{j=1}^{\infty} |f_j|^r \right)^{1/r} \right\|_{\mathcal{M}^p_{\vec{\alpha}}}$$

for all sequences of measurable functions $\{f_j\}_{j=1}^{\infty}$.

See [130, Theorem 2.2] and [146, Lemma 2.5] for the case of classical Morrey spaces.

2.1.4 Predual spaces of mixed Morrey spaces

In this subsection, we introduce the predual spaces of mixed Morrey spaces following the idea of Long [87]. We need these space in Section 5.

Definition 2.1.25. Let $1 \le p < \infty$ and $\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$. A measurable function A is said to be a (p, \vec{q}) -block if there exists a cube Q that supports A such that

$$||A||_{\vec{q}} \le |Q|^{\frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_j}\right) - \frac{1}{p}}.$$

Note that the idea of blocks was introduced by Taibleson and Weiss to investigate the a.e. convergence of the Fourier series in [145]. Based on this functions, we define the following function spaces.

Definition 2.1.26. Let $1 \le p < \infty$ and $\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$. Define the function space $\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$

as the set of all $f \in L^p(\mathbb{R}^n)$ for which f is realized as the sum $f = \sum_{j=0}^{\infty} \lambda_j A_j$ with some

 $\lambda = \{\lambda_j\}_{j \in \mathbb{N}_0} \in \ell^1(\mathbb{N}_0)$ and a sequence $\{A_j\}_{j \in \mathbb{N}_0}$ of (p', \vec{q}') -blocks. The norm $\|f\|_{\mathcal{H}^{p'}_{\vec{q}'}}$ for $f \in \mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^n)$ is defined as

$$\|f\|_{\mathcal{H}^{p'}_{\vec{q}'}} \equiv \inf_{\lambda} \|\lambda\|_{\ell^1},$$

where $\lambda = \{\lambda_j\}_{j \in \mathbb{N}_0}$ runs over all admissible expressions

$$f = \sum_{j=0}^{\infty} \lambda_j A_j, \ \{\lambda_j\}_{j \in \mathbb{N}_0} \in \ell^1, \ A_j \text{ is a } (p', \vec{q}') \text{-block for all } j \in \mathbb{N}_0.$$
(2.8)

Note that if $q_1 = \cdots = q_n$, then the notion of (p, \vec{q}) -block and the one of $\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$ coincide the classical ones.

Remark 2.1.27. As is easily verified by Hölder's inequality, any (p, \vec{q}) -block has L^p norm less than 1;

$$\|A\|_p \le 1$$

for all blocks A. Due to this fact, the series $f = \sum_{j=0}^{\infty} \lambda_j A_j$ in Definition 2.1.26 converges in the topology of $L^p(\mathbb{R}^n)$. In fact, let n < m. Then,

$$\left\|\sum_{j=0}^{m}\lambda_{j}A_{j}-\sum_{j=0}^{n}\lambda_{j}A_{j}\right\|_{p}=\left\|\sum_{j=n+1}^{m}\lambda_{j}A_{j}\right\|_{p}\leq \sum_{j=n+1}^{m}|\lambda_{j}|\|A_{j}\|_{p}\leq \sum_{j=n+1}^{m}|\lambda_{j}|\longrightarrow 0$$

as $n, m \to 0$. Thus, this series converges in the topology of $L^p(\mathbb{R}^n)$.

We shall see some properties of the space $\mathcal{H}_{\vec{q'}}^{p'}(\mathbb{R}^n)$.

Lemma 2.1.28. Let $1 \le p < \infty$ and $\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$. If A is a (p', \vec{q}') -block, then $\|A\|_{\mathcal{H}^{p'}_{\vec{q}'}} \le 1.$

Proof. In (2.8), simply choose

$$A_0 = A, A_1 = A_2 = \dots = 0, \lambda_0 = 1, \lambda_1 = \lambda_2 = \dots = 0$$

Lemma 2.1.29. Let $1 \leq p < \infty$ and $\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$. Let A be an $L^{\vec{q}'}(\mathbb{R}^n)$ function supported on a cube Q. Then

$$\|A\|_{\mathcal{H}_{\vec{q}'}^{p'}} \le \|A\|_{\vec{q}'} |Q|^{\frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_j}\right) - \frac{1}{p}}.$$
(2.9)

Proof. Set $B \equiv \frac{|Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)}{\|A\|_{\vec{q}'}} A$, and assume that A is not zero for almost everywhere. Then B is supported on a cube Q and by virtue of the facts that $\frac{1}{p} + \frac{1}{p'} = 1$ and $\frac{1}{\vec{q}} + \frac{1}{\vec{q}'} = (1, 1, \dots, 1),$

$$||B||_{\vec{q}'} = |Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_j}\right)} = |Q|^{\frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q'_j}\right) - \frac{1}{p'}}.$$

Hence, B is a (p', \vec{q}') -block. By Lemma 2.1.28, $||B||_{\mathcal{H}_{\vec{q}'}^{p'}} \leq 1$. Thus, we obtain (2.9).

Recall that $L_{c}^{\vec{q}}(\mathbb{R}^{n})$ denotes the set of all $L^{\vec{q}}(\mathbb{R}^{n})$ function with compact support. By Lemma 2.1.29, the elements of $L_{c}^{\vec{q}}(\mathbb{R}^{n})$ can be regarded as a (p', \vec{q}') -block modulo multiplicative constants. From this fact, we also consider the density for the space $\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^{n})$.

Lemma 2.1.30. The space $L^{\vec{q}}_{c}(\mathbb{R}^{n})$ is dense in $\mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^{n})$. In particular, the space $\mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^{n})$ is separable.

Proof. We shall verify that for all $g \in \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$, the sequence $\{h_j\}_{j=1}^{\infty} \subset L_c^{\vec{q}}(\mathbb{R}^n)$ exists such that $\lim_{j \to \infty} h_j = g$ in $\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$. Since $g \in \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$, there exist $\{\lambda_j\}_{j=1}^{\infty} \in \ell^1$ and $\{b_j\}_{j=1}^{\infty}$ which is a sequence of (p', \vec{q}') -blocks such that

$$g = \sum_{k=1}^{\infty} \lambda_k b_k.$$

For each $j \in \mathbb{N}$, let

$$h_j \equiv \sum_{k=1}^j \lambda_k b_k.$$

Then $h_j \in L^{\vec{q}}_{c}(\mathbb{R}^n)$ and since

$$|g - h_j| \le \sum_{k=j+1}^{\infty} |\lambda_k b_k|,$$

we have

$$\|g - h_j\|_{\mathcal{H}_{\vec{q}'}^{p'}} \le \sum_{k=j+1}^{\infty} |\lambda_k| \to 0 \quad (j \to \infty).$$

The following theorem is an extension of the result by Long [87] to mixed Morrey spaces.

Theorem 2.1.31. Suppose that $1 \le p < \infty$ and $\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}$.

(i) Any $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ defines a continuous functional L_f by

$$L_f: \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n) \ni g \longmapsto \int_{\mathbb{R}^n} f(x)g(x) \mathrm{d}x \in \mathbb{C}$$

on $\mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^n)$.

- (ii) Conversely, every continuous functional L on $\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$ can be realized with $f \in \mathcal{M}_{\vec{q}}^p(\mathbb{R}^n)$.
- (iii) The correspondence

$$\tau : \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n}) \ni f \longmapsto L_{f} \in \left(\mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^{n})\right)^{*}$$

is an isomorphism. Furthermore,

$$\|f\|_{\mathcal{M}^{p}_{\vec{q}}} = \sup\left\{ \left| \int_{\mathbb{R}^{n}} f(x)g(x) \mathrm{d}x \right| : g \in \mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^{n}), \|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} = 1 \right\}$$
(2.10)

and

$$\|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} = \max\left\{ \left| \int_{\mathbb{R}^n} f(x)g(x) \mathrm{d}x \right| : f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n), \|f\|_{\mathcal{M}^p_{\vec{q}}} = 1 \right\}.$$
 (2.11)

Proof. (i) Since $g \in \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$, for any $\varepsilon > 0$, there exist a non-negative sequence $\{\lambda_j\}_{j\in\mathbb{N}} \in \ell^1(\mathbb{N})$ and a sequence $\{g_j\}_{j\in\mathbb{N}}$ of (p', \vec{q}') -blocks such that

$$g = \sum_{j=1}^{\infty} \lambda_j g_j,$$

and

$$\sum_{j=1}^{\infty} \lambda_j \le \|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} + \varepsilon,$$

where each g_j is supported on Q_j . Then, by Hölder's inequality,

$$|L_f(g)| = \left| \int_{\mathbb{R}^n} f(x) \sum_{j=1}^\infty \lambda_j g_j(x) dx \right| \le \sum_{j=1}^\infty \lambda_j \int_{\mathbb{R}^n} |f(x)g_j(x)| dx$$
$$\le \sum_{j=1}^\infty \lambda_j \|f\chi_{Q_j}\|_{\vec{q}} \|g_j\|_{\vec{q}'}.$$

Since each g_j is a (p', \vec{q}') -block which is supported on Q_j , it follows that

$$|L_f(g)| \leq \sum_{j=1}^{\infty} \lambda_j |Q_j|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{\infty} \frac{1}{q_j} \right)} \|f \chi_{Q_j}\|_{\vec{q}}$$
$$\leq \|f\|_{\mathcal{M}^p_{\vec{q}}} \sum_{j=1}^{\infty} \lambda_j \leq \|f\|_{\mathcal{M}^p_{\vec{q}}} \left(\|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} + \varepsilon \right).$$

Since ε is arbitrary, we have

$$|L_f(g)| \le \|f\|_{\mathcal{M}^p_{\vec{q}}} \|g\|_{\mathcal{H}^{p'}_{\vec{q}'}}.$$
(2.12)

Thus, we conclude

$$\|L_f\|_* \le \|f\|_{\mathcal{M}^p_{\vec{a}}},\tag{2.13}$$

where $\|\cdot\|_*$ denotes the operator norm.

(ii) We take a cube Q_0 and let $Q_j \equiv 2^j Q_0$ for $j \in \mathbb{N}$. For the sake of the simplicity, we write

$$L^{\vec{q}'}(Q_j) \equiv \left\{ f \in L^{\vec{q}'}(\mathbb{R}^n) : f \text{ is supported on } Q_j \right\}.$$

According to the proof of Lemma 2.1.29, since we can regard the element of $L^{\vec{q}'}(Q_j)$ as a (p', \vec{q}') -block modulo multiplicative constant, the functional $g \mapsto L(g)$ is well defined and bounded on $L^{\vec{q}'}(Q_j)$. Thus, by the $L^{\vec{q}} - L^{\vec{q}'}$ duality [11], there exists $f_j \in L^{\vec{q}}(Q_j)$ such that

$$L(g) = \int_{Q_j} f_j(x)g(x) \mathrm{d}x$$

for all $g \in L^{\vec{q}'}(Q_j)$. By the uniqueness of this theorem, we can find $L^{\vec{q}}_{loc}(\mathbb{R}^n)$ -function f such that

$$f|_{Q_j} = f_j$$
 a.e.

for any j. We shall prove $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. For a fixed cube Q and the above f, we set

$$g \equiv (\overline{\operatorname{sgn} f}) |f|^{q_1 - 1} \chi_Q ||f \chi_Q||^{q_2 - q_1}_{(q_1)} ||f \chi_Q||^{q_3 - q_2}_{(q_1, q_2)} \cdots ||f \chi_Q||^{q_n - q_{n-1}}_{(q_1, \dots, q_{n-1})}.$$

A simple calculation shows

$$\int_{Q} f(x)g(x)dx = \|f\chi_{Q}\|_{\vec{q}}^{q_{n}}, \quad \|g\|_{\vec{q}'} = \|f\chi_{Q}\|_{\vec{q}}^{q_{n}-1}$$

Then we can write

$$|Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)} ||f\chi_{Q}||_{\vec{q}} = |Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)} \left(\int_{Q} f(x)g(x)dx\right)^{\frac{1}{q_{n}}} = |Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)} (L(g))^{\frac{1}{q_{n}}}.$$
(2.14)

Meanwhile, thanks to Lemma 2.1.29,

$$|L(g)| \le ||L||_* |Q|^{\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{q_j}\right) - \frac{1}{p}} ||g||_{\vec{q}'} = ||L||_* |Q|^{\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{q_j}\right) - \frac{1}{p}} ||f\chi_Q||_{\vec{q}}^{q_n - 1}.$$
(2.15)

Using (2.14) and (2.15), we obtain

$$||f||_{\mathcal{M}^p_{\vec{a}}} \le ||L||_*,$$
 (2.16)

so that $f \in \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$. Hence, we conclude that L is realized as $L = L_{f}$ for $f \in \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$ at least on $g \in L^{\vec{q}}_{c}(\mathbb{R}^{n})$. Since $L^{\vec{q}'}_{c}(\mathbb{R}^{n})$ is dense in $\mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^{n})$ by Lemma 2.1.30, we can obtain the desired result.

(iii) Thanks to (2.13), (2.16) and (ii), it follows that τ is an isomorphism. We shall check (2.10). By virtue of (2.12), we have

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \ge \sup\left\{ \left| \int_{\mathbb{R}^n} f(x)g(x) \mathrm{d}x \right| : g \in \mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^n), \|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} = 1 \right\}.$$

Fix a cube Q. We can assume that $f \not\equiv 0$ on Q. Then, let

$$g \equiv (\overline{\operatorname{sgn} f}) |f|^{q_1 - 1} \chi_Q ||f \chi_Q||_{(q_1)}^{q_2 - q_1} ||f \chi_Q||_{(q_1, q_2)}^{q_3 - q_2} \cdots ||f \chi_Q||_{(q_1, \dots, q_{n-1})}^{q_n - q_{n-1}},$$

and

$$h \equiv \frac{|Q|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)}}{\|g\|_{\vec{q}'}} g.$$

By Lemma 2.1.29, we see that $\|h\|_{\mathcal{H}^{p'}_{\vec{a}'}} \leq 1$. Therefore,

$$\int f(x)h(x)dx = \frac{|Q|^{\frac{1}{p} - \frac{1}{n}\left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)}}{\|g\|_{\vec{q}'}} \int_{Q} f(x)g(x)dx = \frac{|Q|^{\frac{1}{p} - \frac{1}{n}\left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)}}{\|f\chi_{Q}\|_{\vec{q}}^{q_{n}-1}} \|f\chi_{Q}\|_{\vec{q}}^{q_{n}}$$
$$= |Q|^{\frac{1}{p} - \frac{1}{n}\left(\sum_{j=1}^{n} \frac{1}{q_{j}}\right)} \|f\chi_{Q}\|_{\vec{q}}.$$

Taking the supremum over the all functions $h \in \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$ satisfying $\|h\|_{\mathcal{H}_{\vec{q}'}^{p'}} \leq 1$, we obtain

$$|Q|^{\frac{1}{p}-\frac{1}{n}\left(\sum_{j=1}^{n}\frac{1}{q_{j}}\right)}\|f\chi_{Q}\|_{\vec{q}} \leq \sup\left\{\left|\int_{\mathbb{R}^{n}}f(x)h(x)\mathrm{d}x\right|:h\in\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^{n}),\|h\|_{\mathcal{H}_{\vec{q}'}^{p'}}=1\right\}.$$

Thus, we conclude

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \leq \sup\left\{ \left| \int_{\mathbb{R}^n} f(x)g(x) \mathrm{d}x \right| : g \in \mathcal{H}^{p'}_{\vec{q}'}(\mathbb{R}^n), \|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} = 1 \right\},\$$

so that we have (2.10). Meanwhile, by (2.12), we have

$$\|g\|_{\mathcal{H}^{p'}_{\vec{q}'}} \ge \sup\left\{ \left| \int_{\mathbb{R}^n} f(x)g(x) \mathrm{d}x \right| : f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n), \|f\|_{\mathcal{M}^p_{\vec{q}}} = 1 \right\}.$$

Using the Hahn–Banach theorem, we learn that there exists a functional $\tilde{L} \in \left(\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)\right)^*$ such that $\|\tilde{L}\|_* = 1$ and $\|g\|_{\mathcal{H}_{\vec{q}'}^{p'}} = \tilde{L}(g)$. Since $\tilde{L} \in \left(\mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)\right)^*$, by (ii), there is a function $f \in \mathcal{M}_{\vec{q}}^p(\mathbb{R}^n)$ such that $\tilde{L}(g) = L_f(g)$, and that

$$||f||_{\mathcal{M}^p_{\vec{n}}} = ||L_f||_* = ||L||_* = 1.$$

Thus, we obtain (2.11).

Finally, we give an example of functions in the predual spaces $\mathcal{H}_{\vec{a}'}^{p'}(\mathbb{R}^n)$.

Example 2.1.32. Let $1 and <math>\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$. Then, we have

$$\|\chi_Q\|_{\mathcal{H}^{p'}_{\vec{q}'}} = |Q|^{\frac{1}{p'}}$$

Applying Lemma 2.1.29 to the function χ_Q , we have

$$\|\chi_Q\|_{\mathcal{H}^{p'}_{\vec{q}'}} \le \|\chi_Q\|_{\vec{q}'} |Q|^{\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{q_j}\right) - \frac{1}{p}} = |Q|^{1 - \frac{1}{p}} = |Q|^{\frac{1}{p'}}.$$

Meanwhile, let $f(x) \equiv \frac{\chi_Q(x)}{\|\chi_Q\|_{\mathcal{M}^p_{\vec{q}}}}$. Then, $\|f\|_{\mathcal{M}^p_{\vec{q}}} = 1$. Thanks to Theorem 2.1.31 (iii), we obtain

$$\|\chi_Q\|_{\mathcal{H}^{p'}_{\vec{q}\,'}} \ge \int_{\mathbb{R}^n} f(x)\chi_Q(x)\mathrm{d}x = \frac{1}{\|\chi_Q\|_{\mathcal{M}^p_{\vec{q}}}} \int_{\mathbb{R}^n} \chi_Q(x)\mathrm{d}x = |Q|^{1-\frac{1}{p}} = |Q|^{\frac{1}{p'}}.$$

Thus, we conclude $\|\chi_Q\|_{\mathcal{H}^{p'}_{\vec{q}'}} = |Q|^{\frac{1}{p'}}$.

2.2 Other function spaces

2.2.1 Besov spaces

Besov spaces have a lot of studies. First, Besov introduced this space using differences in 1959 [13]. Peetre characterized this space by using Fourier transform in 1967 [115]. After that Besov spaces have been investigated by many authors until now. As for the relation with Morrey sapces, Kozono and Yamazaki introduced Besov-Morrey spaces which were introduced to apply to Navier–Stokes equations [81]. Furthermore, there are many textbooks including these spaces [116, 126, 147, 148].

In this subsection, we recall the definition and some fundamental properties of Besov spaces. All of the proofs are referred to books [116, 126, 147, 148] or a survey paper [128].

In this thesis, we employ the well-known definition by the Fourier transform. For $\tau \in \mathcal{S}(\mathbb{R}^n)$ and $f \in \mathcal{S}'(\mathbb{R}^n)$, define $\tau(D)f \equiv \mathcal{F}^{-1}[\tau \cdot \mathcal{F}f]$. Next, we define the Littlewood– Paley decomposition. Let $\varphi \in C_c^{\infty}(\mathbb{R}^n)$ satisfy

$$\chi_{B(4)\setminus B(2)} \le \varphi \le \chi_{B(8)\setminus B(\frac{3}{2})}.$$
(2.17)

Then define $\varphi_j \equiv \varphi(2^{-j} \cdot)$. The *j*-th Littlewood–Paley decomposition is the operator $\varphi_j(D)$. Now we are ready to define the (homogeneous) Besov space $\dot{B}_{pr}^s(\mathbb{R}^n)$ for $1 \leq p < \infty, 1 \leq r \leq \infty$ and $s \in \mathbb{R}$ using the polynomial space $\mathcal{P}(\mathbb{R}^n)$.

Definition 2.2.1. Let $s \in \mathbb{R}$, $1 \le p, r \le \infty$. We define

$$\|f\|_{\dot{B}^s_{pr}} \equiv \left(\sum_{j=-\infty}^{\infty} (2^{js} \|\varphi_j(D)f\|_{L^p})^r\right)^{\frac{1}{r}}$$

for $f \in \mathcal{S}'(\mathbb{R}^n)/\mathcal{P}(\mathbb{R}^n)$. The (homogeneous) *Besov space* $\dot{B}^s_{pq}(\mathbb{R}^n)$ is the set of all $f \in \mathcal{S}'(\mathbb{R}^n)/\mathcal{P}(\mathbb{R}^n)$ for which the norm $\|f\|_{\dot{B}^s_{pq}}$ is finite.

Remark that in the above definition, we choose φ so that the norm of $\dot{B}^s_{pq}(\mathbb{R}^n)$ depends on φ . However, we can verify that $\dot{B}^s_{pq}(\mathbb{R}^n)$ is independent of φ as the set.

Theorem 2.2.2 ([126, Theorem 2.1]). In Definition 2.2.1, we obtain the equivalent norms for the admissible choice of φ .

We recall the elementary properties for Besov spaces.

Theorem 2.2.3 ([126, Theorem 2.4]). Let $1 \le p, q \le \infty$ and $s \in \mathbb{R}$. Then, $\dot{B}^s_{pq}(\mathbb{R}^n)$ is complete, that is, $\dot{B}^s_{pq}(\mathbb{R}^n)$ is Banach space.

The embedding properties are as follows.

Proposition 2.2.4.

(1) ([126, Proposition 2.2]) For $s \in \mathbb{R}$, $1 \le p \le \infty$ and $1 \le r_1 \le r_2 \le \infty$, we have

$$\dot{B}^{s}_{pr_1}(\mathbb{R}^n) \hookrightarrow \dot{B}^{s}_{pr_2}(\mathbb{R}^n).$$

(2) ([126, Theorem 2.5]) For $s \in \mathbb{R}, 1 \le p_1 \le p_2 \le \infty$ and $1 \le r_1 \le r_2 \le \infty$,

$$\dot{B}^s_{p_1r_1}(\mathbb{R}^n) \hookrightarrow \dot{B}^{s-(\frac{n}{p_1}-\frac{n}{p_2})}_{p_2r_2}(\mathbb{R}^n).$$

Moreover, we consider the relation between Besov spaces and Lebesgue spaces. **Proposition 2.2.5.**

(1) ([126, Theorem 4.4]) For $1 \le p \le q \le \infty$, $\dot{B}_{p1}^{\frac{n}{p}-\frac{n}{q}}(\mathbb{R}^n) \hookrightarrow L^q(\mathbb{R}^n)$.

- (2) ([126, Proposition 2.1]) For $1 \le p \le \infty$, $\dot{B}^0_{p1}(\mathbb{R}^n) \hookrightarrow L^p(\mathbb{R}^n) \hookrightarrow \dot{B}^0_{p\infty}(\mathbb{R}^n)$.
- (3) ([126, Exercise 3.8]) For $1 , <math>\dot{B}^0_{p2}(\mathbb{R}^n) \hookrightarrow L^p(\mathbb{R}^n)$. Moreover, by duality, for $1 , <math>L^p(\mathbb{R}^n) \hookrightarrow \dot{B}^0_{p2}(\mathbb{R}^n)$.
- (4) ([126, Exercise 3.8]) For $1 \le p \le 2$, $\dot{B}^0_{pp}(\mathbb{R}^n) \hookrightarrow L^p(\mathbb{R}^n)$. Moreover, by duality, for $2 \le p \le \infty$, $L^p(\mathbb{R}^n) \hookrightarrow \dot{B}^0_{pp}(\mathbb{R}^n)$.

In Section 4, we use the Besov space $\dot{B}_{p1}^s(\mathbb{R}^n)$ characterized by atoms. So we only describe the characterization of the Besov space $\dot{B}_{p1}^s(\mathbb{R}^n)$. Full statements for Besov spaces can be found in many books. See [126, 147, 148, 149].

Theorem 2.2.6. Let $1 \le p < \infty$ and $0 < s \le \frac{n}{p}$. Define $q \in (p, \infty)$ by

$$-\frac{n}{q} = s - \frac{n}{p}, \quad that \ is \ q = n\left(\frac{n}{p} - s\right)^{-1}$$

Then, the Besov space $\dot{B}_{p1}^{s}(\mathbb{R}^{n})$ coincides with the set of all $f \in L^{q}(\mathbb{R}^{n})$ for which it can be expressed:

$$f = \sum_{j=-\infty}^{\infty} \left(\sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right)$$

in $L^q(\mathbb{R}^n)$ for some complex sequence $\Lambda \equiv \{\lambda_{jm}\}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ and some sequence $A \equiv \{a_{jm}\}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ of C^{∞} -functions satisfying

$$\sum_{j=-\infty}^{\infty} 2^{js} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^p} < \infty, \quad |\partial^{\alpha} a_{jm}| \le 2^{j|\alpha|} \chi_{3Q_{jm}}$$

for all $(j,m) \in \mathbb{Z} \times \mathbb{Z}^n$ and multiindices α with $|\alpha| \leq [s+1]$. Moreover the Besov norm $||f||_{\dot{B}^s_{p_1}}$ is equivalent to the infimum of

$$\sum_{j=-\infty}^{\infty} 2^{js} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^p} = \sum_{j=-\infty}^{\infty} 2^{js-j\frac{n}{p}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^p \right)^{\frac{1}{p}}$$
$$= \sum_{j=-\infty}^{\infty} 2^{-j\frac{n}{q}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^p \right)^{\frac{1}{p}},$$

where $\Lambda = {\lambda_{jm}}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ moves over all possible expressions.

Remark that we use a different definition of sequence spaces to make a_{jm} behave almost similarly to $\chi_{Q_{jm}}$. See [148] and [149, §13.1].

As is easily seen from Proposition 2.2.5, $B_{p1}^s(\mathbb{R}^n)$ is continuously embedded into $L^q(\mathbb{R}^n)$. We remark that Theorem 2.2.6 is an expression of homogeneous Besov spaces

that differs from the one in [126, 147, 148, 149]. See [126, 148, 149] for the fact that these two definitions are actually the same.

Finally, we give an example of the elements of $\dot{B}_{p1}^{s}(\mathbb{R}^{n})$. This example will use to prove Theorem 4.1.5.

Example 2.2.7. Let $1 and <math>0 < s \leq \frac{n}{p}$. Fix $j \in \mathbb{Z}$. Then, by using the partition of unity subordinate to the covering $\{3Q_{jm}\}_{m\in\mathbb{Z}^n}$, we can show that $\exp(-|2^j \cdot -m|^2) \in \dot{B}^s_{p1}(\mathbb{R}^n)$ satisfies $\|\exp(-|2^j \cdot -m|^2)\|_{\dot{B}^s_{p1}} \lesssim 2^{js-j\frac{n}{p}}$ for all $m \in \mathbb{Z}^n$.

2.2.2 Microlocal Besov spaces

We recall the generalized 2-microlocal Besov space $B_{pq}^{s,\text{mloc}}(\mathbb{R}^n, w)$. The idea of 2microlocal analysis is due to Bony in 1984 [16]. It is an appropriate instrument to describe the local regurality and the oscillatory behavior of functions near singularities. Later many authors investigated the function spaces introduced this idea. In particular, Moritoh and Yamada introduced this idea into Besov spaces and characterized these spaces in 2004 [96]. After that Kempka defined and investigated the generalized 2microlocal Besov space $B_{pq}^{s,\text{mloc}}(\mathbb{R}^n, w)$ [74].

To define 2-microlocal Besov spaces, we recall the class $\mathcal{W}^{\alpha_3}_{\alpha_1,\alpha_2}$:

Definition 2.2.8 (Weight class $\mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$). Let $\alpha_1, \alpha_2, \alpha_3 \in [0, \infty)$. The class $\mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$ of weights is defined as the set of all the sequences of the measurable functions $w = \{w_j\}_{j=-\infty}^{\infty}$ satisfying the following conditions:

1. There exists a constant C > 0 such that for all $x, y \in \mathbb{R}^n$ and $j \in \mathbb{Z}$,

$$0 < w_j(x) \le Cw_j(y)(1+2^j|x-y|)^{\alpha_3}.$$

2. For all $x \in \mathbb{R}^n$ and $j \in \mathbb{Z}$,

$$2^{-\alpha_1} w_j(x) \le w_{j+1}(x) \le 2^{\alpha_2} w_j(x).$$

Such a sequence $w \in \mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$ is called an admissible weight sequence.

The (homogeneous) generalized 2-microlocal Besov spaces are usually defined by the use of the Fourier multipliers as follows.

Definition 2.2.9 (Generalized 2-microlocal Besov spaces). Let $w \in \mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$. Let φ satisfy (2.17) and define $\varphi_j(x) = \varphi(2^{-j}x)$. Let $0 < p, q \leq \infty$. Then for $f \in \mathcal{S}'(\mathbb{R}^n)$ define

$$\|f\|_{\dot{B}^{s,\mathrm{mloc}}_{pq}(\mathbb{R}^n,w)} \equiv \left(\sum_{j=-\infty}^{\infty} 2^{js} \|w_j \mathcal{F}^{-1}[\varphi_j \mathcal{F}f]\|_p^q\right)^{\frac{1}{q}}.$$

The generalized 2-microlocal Besov space $\dot{B}_{pq}^{s,\text{mloc}}(\mathbb{R}^n, w)$ is the set of all $f \in \mathcal{S}'(\mathbb{R}^n)$ for which $\|f\|_{\dot{B}_{pq}^{s,\text{mloc}}(\mathbb{R}^n,w)}$ is finite.

Note that if we take $w_j(x) = 1$ for all $x \in \mathbb{R}^n$ and $j \in \mathbb{Z}$, we obtain classical Besov spaces. Also remark that the norm $\|\cdot\|_{\dot{B}^{s,\mathrm{mloc}}_{pq}(\mathbb{R}^n,w)}$ is independent of the choice of φ . We refer to [74].

Kempka characterized the spaces $B_{pq}^{s,\text{mloc}}(\mathbb{R}^n, w)$ via the atomic decomposition in [74].

Theorem 2.2.10. Let $w = \{w_j\}_{j=-\infty}^{\infty} \in \mathcal{W}_{\alpha_1,\alpha_2}^{\alpha_3}$, $s \in \mathbb{R}$, and $0 < p, q \leq \infty$. Furthermore, let $K, L \in \mathbb{N}_0$ with

$$K > s + \alpha_2$$
, and $L > \max\left\{0, n\left(\frac{1}{p} - 1\right)\right\} - s + \alpha_1$.

For each $f \in \dot{B}_{pq}^{s,\mathrm{mloc}}(\mathbb{R}^n,w)$, there exist sequences

 $\{\lambda_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}\subset\mathbb{C},\ \{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}\subset C^{\infty}(\mathbb{R}^n)\ and\ \{Q_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}\subset\mathcal{D}$

such that the representation

$$f = \sum_{j=-\infty}^{\infty} \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm},$$

holds, where the triplet $\{\lambda_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$, $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ and $\{Q_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ satisfies

$$|\partial^{\alpha} a_{jm}| \le 2^{j|\alpha|} \chi_{3Q_{jm}}, \quad \int_{\mathbb{R}^n} x^{\beta} a_{jm} \mathrm{d}x = 0, \quad \left(\sum_{j \in \mathbb{Z}} 2^{jsq} \left\| \sum_{m \in \mathbb{Z}^n} w_j \lambda_{jm} \chi_{Q_{jm}} \right\|_p^q \right)^{\overline{q}} < \infty$$

for all multiindices α and β with $|\alpha| \leq K$ and $|\beta| \leq L$. Here the convergence is in $\mathcal{S}'(\mathbb{R}^n)$. Moreover, there exists a constant C > 0 such that

$$\left(\sum_{j\in\mathbb{Z}}2^{jsq}\left\|\sum_{m\in\mathbb{Z}^n}w_j\lambda_{jm}\chi_{Q_{jm}}\right\|_p^q\right)^{\frac{1}{q}} \le C\|f\|_{\dot{B}^{s,\mathrm{mloc}}_{pq}(\mathbb{R}^n,w)}$$

for all $f \in \dot{B}^{s,\mathrm{mloc}}_{pq}(\mathbb{R}^n,w)$.

Note that this characterization will be used in Section 4, see Definition 4.1.2.

2.2.3 Lorentz spaces

Next, we turn to Lorentz spaces. Lorentz spaces were introduced by Lorentz in [88, 89]. A general treatment of this space was given in the article of Hunt [61]. The boundedness of the classical results on Lorentz spaces are investigated in [9, 136]. Here, we recall the definitions and elementary facts needed in Subsection 4.3.2. For more details, we refer to [41, Section 1.4] or [143, Chapter V].

To define Lorentz spaces, we prepare some notation. Let $f : \mathbb{R}^n \to \mathbb{C}$ be a measurable function. Then the *distribution function* $\lambda_f : [0, \infty) \to [0, \infty]$ is a function defined by

$$\lambda_f(t) \equiv |\{x \in \mathbb{R}^n : |f(x)| > t\}| \quad (t \ge 0).$$

Definition 2.2.11. Let $f \in L^0(\mathbb{R}^n)$. Then its *decreasing rearrangement* f^* is the function defined on $(0, \infty)$ by

$$f^*(t) \equiv \inf(\{s \in [0, \infty) : \lambda_f(s) \le t\} \cup \{\infty\}) \quad (t > 0).$$

Definition 2.2.12.

1. Let $0 and <math>0 < q < \infty$. Then the Lorentz space $L^{p,q}(\mathbb{R}^n)$ is the set of all $f \in L^0(\mathbb{R}^n)$ for which the quasi-norm

$$||f||_{L^{p,q}} \equiv \left\{ \int_0^\infty (t^{\frac{1}{p}} f^*(t))^q \frac{\mathrm{d}t}{t} \right\}^{\frac{1}{q}}$$

is finite.

- 2. The Lorentz space $L^{\infty,\infty}(\mathbb{R}^n)$ stands for $L^{\infty}(\mathbb{R}^n)$.
- 3. If $0 , then the Lorentz space <math>L^{p,\infty}(\mathbb{R}^n)$ denotes the weak L^p -space: $L^{p,\infty}(\mathbb{R}^n) = WL^p(\mathbb{R}^n).$

By virtue of the definition, we easily show that $L^{p,p}(\mathbb{R}^n) = L^p(\mathbb{R}^n)$ for 0 . $Remark that the space <math>L^{p,q}(\mathbb{R}^n)$ is complete under the above quasi-norm $\|\cdot\|_{L^{p,q}}$, that is, $L^{p,q}(\mathbb{R}^n)$ is a quasi-Banach space. Meanwhile, we can show that the Lorentz space $L^{p,q}(\mathbb{R}^n)$ is normable when p, q > 1. To see this, we shall define the function f^{**} . The idea to use this function is due to Calderón [17]. We now set the maximal function f^{**} by

$$f^{**}(t) \equiv \frac{1}{t} \int_0^t f^*(s) \mathrm{d}s, \quad 0 < t \le \infty.$$

Then define the norm

$$||f||_{L^{p,q}}^* \equiv \begin{cases} \left\{ \int_0^\infty (t^{\frac{1}{p}} f^{**}(t))^q \frac{\mathrm{d}t}{t} \right\}^{\frac{1}{q}}, & 1 0} t^{\frac{1}{p}} f^{**}(t), & 1$$

Thanks to the Hardy inequality, if $1 and <math>1 \le q \le \infty$, then $\|\cdot\|_{L^{p,q}}^* \sim \|\cdot\|_{L^{p,q}}$ holds. Furthermore, $L^{p,q}(\mathbb{R}^n)$ is a Banach space with the norm $\|\cdot\|_{L^{p,q}}^*$.

The following result shows that the scale of Lorentz spaces is monotone for the parameter q increases for any fixed p.

Proposition 2.2.13. Suppose that $0 and <math>0 < q < r \le \infty$. Then, the embedding

$$L^{p,q}(\mathbb{R}^n) \hookrightarrow L^{p,r}(\mathbb{R}^n)$$

holds.

At last, we consider the quasi-triangle inequality for the infinite sum. The following lemma is somehow well known. But it seems that its proof is missing in the literature. So, we give a proof.
Lemma 2.2.14. Let 0 . If we decompose a measurable function <math>f by $f = \sum_{j=1}^{\infty} f_j$ such that $\{\operatorname{supp}(f_j)\}_{j=1}^{\infty}$ are pairwise disjoint, then we have

$$\|f\|_{L^{p,q}}^p \lesssim \sum_{j=1}^{\infty} \|f_j\|_{L^{p,q}}^p$$

Proof. We will use the property of the Lorentz norm (see [41, Proposition 1.4.9]):

$$\|f\|_{L^{p,q}} \sim \begin{cases} \left\{ \int_0^\infty (t\lambda_f(t)^{\frac{1}{p}})^q \frac{\mathrm{d}t}{t} \right\}^{\frac{1}{q}}, & q < \infty, \\ \sup_{t>0} t\lambda_f(t)^{\frac{1}{p}}, & q = \infty. \end{cases}$$

Note that we calculate

$$\lambda_f(t) = \left| \bigsqcup_{j=1}^{\infty} \{ x \in \mathbb{R}^n : |f_j(x)| > t \} \right| = \sum_{j=1}^{\infty} \lambda_{f_j}(t).$$

Then by using the triangle inequality, we have

$$\|f\|_{L^{p,q}}^{p} \sim \left\|\sum_{j=1}^{\infty} \lambda_{f_{j}}\right\|_{L^{\frac{q}{p}}(t^{q-1} \mathrm{d}t)} \leq \sum_{j=1}^{\infty} \|\lambda_{f_{j}}\|_{L^{\frac{q}{p}}(t^{q-1} \mathrm{d}t)} = \sum_{j=1}^{\infty} \|f_{j}\|_{L^{p,q}}^{p},$$

where $\|\cdot\|_{L^{\frac{q}{p}}(t^{q-1}\mathrm{d}t)}$ denotes the $L^{\frac{q}{p}}$ -norm with respect to the measure $t^{q-1}\mathrm{d}t$. This is a desired result.

2.2.4 Orlicz spaces

Orlicz spaces initially appeared in 1930's. Birnbaum–Orlicz [14], Orlicz [109, 110, 111], and Nakano [105] investigated Orlicz spaces. Kita investigated the boundedness property of the Hardy–Littlewood maximal operator on Orlicz spaces in [75, 76, 77]. In this thesis, we only recall fundamental facts used in Subsection 4.3.1. For the proof and more details of Orlicz spaces, we refer to [12, 90, 119].

To define Orlicz spaces, we recall the definition of Young functions. A function $\Phi: [0, \infty) \to [0, \infty)$ is a Young function, if it satisfies the following conditions:

- 1. $\Phi(0) = 0$.
- 2. Φ is convex. That is, $\Phi((1-\theta)t_1+\theta t_2) \leq (1-\theta)\Phi(t_1)+\theta\Phi(t_2)$ for all $t_1, t_2 \in (0,\infty)$ and $0 < \theta < 1$.
- 3. $\lim_{t \to 0} \Phi(t) = \Phi(0), \quad \lim_{t \to \infty} \Phi(t) = \infty.$

So, we define the Orlicz space $L^{\Phi}(\mathbb{R}^n)$.

Definition 2.2.15 (Orlicz space). Let $\Phi : [0, \infty) \to [0, \infty)$ be a Young function. Then define the Luxemburg-Nakano norm $\|\cdot\|_{L^{\Phi}}$ by

$$\|f\|_{L^{\Phi}} \equiv \inf\left(\left\{\lambda \in (0,\infty) \, : \, \int_{\mathbb{R}^n} \Phi\left(\frac{|f(x)|}{\lambda}\right) \mathrm{d}x \le 1\right\} \cup \{\infty\}\right)$$

for $f \in L^0(\mathbb{R}^n)$. The Orlicz space $L^{\Phi}(\mathbb{R}^n)$ is the set of all $f \in L^0(\mathbb{R}^n)$ for which $||f||_{L^{\Phi}}$ is finite.

Orlicz spaces extend Lebesgue spaces in the following sense:

Example 2.2.16. If $\Phi(t) = t^p$ for $1 \le p < \infty$, then $L^{\Phi}(\mathbb{R}^n) = L^p(\mathbb{R}^n)$ with coincidence of norms.

Here we content ourselves with the completeness of $L^{\Phi}(\mathbb{R}^n)$ without the proof.

Theorem 2.2.17. Let Φ be a Young function. Then, the Orlicz space $L^{\Phi}(\mathbb{R}^n)$ is a Banach space with respect to $\|\cdot\|_{L^{\Phi}}$.

If we suppose bijective for Young functions Φ , we can calculate $\|\chi_E\|_{L^{\Phi}}$.

Lemma 2.2.18. Assume that Φ is a Young function. Also let Φ be bijective. Then, for all measurable set E with $0 < |E| < \infty$,

$$\|\chi_E\|_{L^{\Phi}} = \left\{ \Phi^{-1} \left(\frac{1}{|E|} \right) \right\}^{-1}.$$

Proof. We write the norm $\|\chi_E\|_{L^{\Phi}}$ in full:

$$\|\chi_E\|_{L^{\Phi}} \equiv \inf \left\{ \lambda \in (0,\infty) : \int_{\mathbb{R}^n} \Phi\left(\frac{\chi_E(x)}{\lambda}\right) \mathrm{d}x \le 1 \right\}.$$

Since $\Phi(0) = 0$, we have

$$1 \ge \int_{\mathbb{R}^n} \Phi\left(\frac{\chi_E}{\lambda}\right) \mathrm{d}x = \int_E \Phi\left(\frac{1}{\lambda}\right) \mathrm{d}x = \Phi\left(\frac{1}{\lambda}\right) \times |E|.$$

By virtue of the bijection of Φ ,

$$\lambda \ge \left\{ \Phi^{-1} \left(\frac{1}{|E|} \right) \right\}^{-1}.$$

Thus,

$$\|\chi_E\|_{L^{\Phi}} = \left\{ \Phi^{-1} \left(\frac{1}{|E|} \right) \right\}^{-1}.$$

Chapter 3

Boundedness of commutators of fractional integral operators on mixed Morrey spaces

3.1 Introduction and theorems

In this section we consider the necessary and sufficient conditions for the boundedness of commutators generated by BMO functions and the fractional integral operator I_{α} on mixed Morrey spaces.

First, we look back on the background and classical results. The idea of commutators for functions and operators appeared first in Coifman-Rocheberg-Weiss's paper [20]. In this paper, they gave the necessary and sufficient condition for the boundedness of commutators generated by functions and singular integral operators on Lebesgue spaces. Meanwhile, Chanillo obtained the following boundedness results for $[b, I_{\alpha}]$ [18].

Theorem 3.1.1. Let $1 < p, q < \infty$ and $0 < \alpha < n$. Assume that

$$\frac{1}{p} = \frac{\alpha}{n} + \frac{1}{q}.$$

If $b \in BMO(\mathbb{R}^n)$, then the commutator $[b, I_\alpha]$ is bounded from $L^p(\mathbb{R}^n)$ to $L^q(\mathbb{R}^n)$. Conversely, if $n - \alpha$ is even and $[b, I_{\alpha}]$ is bounded from $L^{p}(\mathbb{R}^{n})$ to $L^{q}(\mathbb{R}^{n})$, then we have $b \in BMO(\mathbb{R}^n)$.

These results were extended to Morrey spaces. In 1991, Di Fazio and Ragusa gave the necessary and sufficient condition for the boundedness of commutator $[b, I_{\alpha}]$ on Morrey spaces [24].

Theorem 3.1.2. Let $0 < \alpha < n, 1 < q \le p < \frac{n}{\alpha}$. Assume that $\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{q}{p} = \frac{s}{r}.$

$$r = -\frac{1}{p} - -\frac{1}{n}, \quad -\frac{1}{p} = -\frac{1}{n}$$

If $b \in BMO(\mathbb{R}^n)$, then

$$[b, I_{\alpha}] : \mathcal{M}^p_q(\mathbb{R}^n) \to \mathcal{M}^r_s(\mathbb{R}^n).$$

Conversely, if $n - \alpha$ is even and $[b, I_{\alpha}] : \mathcal{M}^p_q(\mathbb{R}^n) \to \mathcal{M}^r_s(\mathbb{R}^n)$, then $b \in BMO(\mathbb{R}^n)$.

Shirai removed the condition for $n - \alpha$ in [140].

Theorem 3.1.3. Let $0 < \alpha < n, 1 < q \le p < \frac{n}{\alpha}$. Also, assume that

$$\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{q}{p} = \frac{s}{r}.$$

Then, the following conditions are equivalent:

- (a) $b \in BMO(\mathbb{R}^n)$.
- (b) $[b, I_{\alpha}]$ is bounded from $\mathcal{M}^{p}_{q}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{s}(\mathbb{R}^{n})$.

Our main theorem extends these results to mixed Morrey spaces. We state our result (This is a same one as Theorem 1.2.1).

$$\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{q_j}{p} = \frac{s_j}{r} \quad (j = 1, \dots, n).$$

Then, the following conditions are equivalent:

- (a) $b \in BMO(\mathbb{R}^n)$.
- (b) $[b, I_{\alpha}]$ is bounded from $\mathcal{M}^{p}_{\vec{a}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{\vec{s}}(\mathbb{R}^{n})$.
- (c) $[b, I_{\alpha}]$ is bounded from $\widetilde{\mathcal{M}}^{p}_{\vec{a}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{\vec{s}}(\mathbb{R}^{n})$.
- (d) $[b, I_{\alpha}]$ is bounded from $\widetilde{\mathcal{M}}^{p}_{\vec{q}}(\mathbb{R}^{n})$ to $\mathcal{M}^{r}_{1}(\mathbb{R}^{n})$.

Here, $\widetilde{\mathcal{M}}^p_{\vec{q}}(\mathbb{R}^n)$ is the $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ -closure of $C^{\infty}_{\mathrm{c}}(\mathbb{R}^n)$.

The remaining parts of this chapter are as follows. In Section 3.2, we establish the sharp maximal inequality on mixed Morrey spaces (Theorem 3.2.6). In Subsection 3.2.1, we prepare some ingredients to show the sharp maximal inequality. Its proof is given in Subsection 3.2.2. Additionally, we check the action of the commutator $[b, I_{\alpha}]$ on mixed Morrey spaces in Subsection 3.2.3. To prove Theorem 3.1.4, Lemma 3.2.15 is significant. Finally, we prove the main theorem in Section 3.3.

3.2 Sharp maximal inequality

We next consider the sharp maximal inequality on mixed Morrey spaces and the relation of the sharp maximal operator and commutators on mixed Morrey spaces to prove the main theorem.

3.2.1 Preliminaries

As we said in Subsection 1.2.1, we cannot apply the layer cake formula and good- λ inequality for mixed norm setting. So instead of these tools, we employ the dyadic local maximal operator $M_{\lambda;Q_0}^{\#,d}$ and the concept of the sparse family. We follow the definition in [84].

Definition 3.2.1. Let $f \in L^0(\mathbb{R}^n)$ and $Q \in \mathcal{Q}$.

1. The decreasing rearrangement of f on \mathbb{R}^n is defined by

$$f^*(t) \equiv |\{\rho > 0 : \mu_f(\rho) > t\}| \quad (0 < t < \infty),$$

where μ_f is a distribution of f. That is, $\mu_f(\rho) = |\{x \in \mathbb{R}^n : |f(x)| > \rho\}|.$

2. The local mean oscillation of f on Q is defined by

$$\omega_{\lambda}(f; Q) \equiv \inf_{c \in \mathbb{C}} \left((f - c) \chi_Q \right)^* (\lambda |Q|) \quad (0 < \lambda < 2^{-1}).$$

3. Assume that the function f is real-valued. Then, the *median* of f over Q, which is denoted by $m_f(Q)$, is a real number satisfying

$$|\{x \in Q : |f(x)| > m_f(Q)\}|, |\{x \in Q : |f(x)| < m_f(Q)\}| \le \frac{1}{2}|Q|.$$

Note that the median $m_f(Q)$ is possibly non-unique.

The symbol $\mathcal{D}(Q_0)$ denotes a set of all cubes with respect to the cube Q_0 , that is, $\mathcal{D}(Q_0)$ is the set of the form

$$\prod_{j=1}^{n} \left[x_j + \frac{(m_j - 1)\ell(Q_0)}{2^k}, x_j + \frac{m_j\ell(Q_0)}{2^k} \right)$$

for all $k \in \mathbb{N}_0$ and $m_j = 1, \ldots, 2^k$ $(j = 1, \ldots, n)$, where (x_1, \ldots, x_n) denotes the left corner of the cube Q_0 . For $0 < \lambda < 2^{-1}$ and $Q_0 \in \mathcal{Q}$, the dyadic local sharp maximal operator $M_{\lambda;Q_0}^{\#,d}$ is defined by

$$M_{\lambda;Q_0}^{\#,d}f(x) \equiv \sup_{Q \in \mathcal{D}(Q_0)} \omega_{\lambda}(f;Q)\chi_Q(x) \quad \left(x \in \mathbb{R}^n, f \in L^0(\mathbb{R}^n)\right).$$

Moreover, we use the following sharp maximal operator

$$M_{\lambda}^{\#,d}f(x) \equiv \sup_{Q_0 \in \mathcal{Q}} \sup_{Q \in \mathcal{D}(Q_0)} \omega_{\lambda}(f;Q)\chi_Q(x) \quad \left(x \in \mathbb{R}^n, f \in L^0(\mathbb{R}^n)\right).$$

Let $f \in L^1_{loc}(\mathbb{R}^n)$. The Fefferman-Stein sharp maximal operator is defined by

$$f^{\#,\eta}(x) \equiv \sup_{x \in Q} \left(\frac{1}{|Q|} \int_Q |f(y) - f_Q|^{\eta} \mathrm{d}y \right)^{\frac{1}{\eta}} \quad (x \in \mathbb{R}^n),$$

where the supremum is taken over all cubes Q in \mathbb{R}^n containing x. When $\eta = 1, f^{\#,\eta}$ equals to $f^{\#}$:

$$f^{\#}(x) = \sup_{x \in Q} \frac{1}{|Q|} \int_{Q} |f(y) - f_Q| dy \quad (x \in \mathbb{R}^n).$$

Jawerth and Torchinsky proved a pointwise equivalence between these two types of the sharp maximal operators in [71] :

$$M^{(\eta)}\left[M_{\lambda}^{\#,d}f\right](x) \sim f^{\#,\eta}(x) \quad (x \in \mathbb{R}^n)$$
(3.1)

for sufficiently small λ , where $M^{(\eta)}$ denotes the *powered Hardy–Littlewood maximal* operator defined by

$$M^{(\eta)}f(x) \equiv \sup_{x \in Q} \left(\frac{1}{|Q|} \int_Q |f(y)|^{\eta} \mathrm{d}y\right)^{\frac{1}{\eta}} \quad (x \in \mathbb{R}^n),$$

where the supremum is taken over all cubes Q in \mathbb{R}^n containing x.

The fractional maximal operator M_{α} is defined by

$$M_{\alpha}f(x) = \sup_{x \in Q} \frac{1}{|Q|^{1-\frac{\alpha}{n}}} \int_{Q} |f(y)| \mathrm{d}y \quad (x \in \mathbb{R}^{n}),$$

where the supremum is taken over all cubes Q in \mathbb{R}^n containing x. Note that the pointwise inequality $M_{\alpha}f(x) \leq I_{\alpha}(|f|)(x), x \in \mathbb{R}^n$ holds.

Moreover we also employ the following ingredient.

Definition 3.2.2. We say that the family of dyadic cubes $\{Q_j^k\}_{k \in \mathbb{N}_0, j \in J_k}$ is a sparse family if the following properties hold:

- 1. For each fixed $k \in \mathbb{N}_0$, the cubes $\{Q_j^k\}_{j \in J_k}$ are disjoint;
- 2. If $\Omega_k \equiv \bigcup_{j \in J_k} Q_j^k$, then $\Omega_{k+1} \subset \Omega_k$;
- 3. $|\Omega_{k+1} \cap Q_j^k| \le \frac{1}{2} |Q_j^k|$ for all $j \in J_k$.

Remark 3.2.3. Recently, we substitute the definition of the sparse for the above: Let $0 < \eta < 1$. Then, $S \subset Q$ is η -sparse if for each $Q \in Q$, there exist pairwise disjoint measurable subsets $\{E_Q\}_{Q \in S}$ such that $E_Q \subset Q$ and $|E_Q| \ge \eta |Q|$. For more details, we refer to [85].

The following lemma is used in the estimate of the dyadic local maximal operator.

Lemma 3.2.4. Let $\{Q_j^k\}_{k \in \mathbb{N}_0, j \in J_k}$ be a sparse family. Then,

$$|Q_j^k| \le 2|Q_j^k \cap \Omega_{k+1}^c|$$

holds for all $k \in \mathbb{N}_0$ and $j \in J_k$.

Proof. For $k \in \mathbb{N}_0$ and $j \in J_k$,

$$|Q_{j}^{k}| \leq |Q_{j}^{k} \cap \Omega_{k+1}^{c}| + |Q_{j}^{k} \cap \Omega_{k+1}| \leq |Q_{j}^{k} \cap \Omega_{k+1}^{c}| + \frac{1}{2}|Q_{j}^{k}|$$

by the condition 3 in Definition 3.2.2. Thus, we obtain the result.

Note that, thanks to this lemma, we can see that the sparse family $\{Q_j^k\}_{k\in\mathbb{N}_0, j\in J_k}$ in Definition 3.2.2 is $\frac{1}{2}$ -sparse. Namely, the idea of η -sparse is generalization of Definition 3.2.2.

To prove the Theorems 3.2.7 and 3.2.8, we invoke the following inequality.

Theorem 3.2.5 ([84]). Let $f \in L^0(\mathbb{R}^n)$ and $Q_0 \in \mathcal{Q}$. Then, there exists a sparse family of $\{Q_j^k\}_{k\in\mathbb{N}_0, j\in J_k} \subset \mathcal{D}(Q_0)$ such that for a.e. $x \in Q_0$,

$$|f(x) - m_f(Q_0)| \le 4M_{\lambda_n;Q_0}^{\#,d} f(x) + 2\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f;Q_j^k) \chi_{Q_j^k}(x).$$

Here, $\lambda_n \equiv 2^{-n-2}$.

3.2.2 Sharp maximal inequality on mixed Morrey spaces

Our aim in this subsection is to show the following sharp maximal inequality for mixed Morrey spaces.

Theorem 3.2.6. Let $0 < \vec{q} < \infty$ and 0 satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}.$$

Then, for any $f \in L^0(\mathbb{R}^n)$ satisfying $Mf \in \mathcal{M}^{p_0}_{\vec{q_0}}(\mathbb{R}^n)$ for some $0 < p_0 < \infty$ and $\vec{q_0} = (q_{0,1}, \ldots, q_{0,n}) \in (0, \infty)^n$ with

$$\frac{n}{p_0} \le \sum_{j=1}^n \frac{1}{q_{0,j}},$$

we have

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \sim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \left\|f^{\#}\right\|_{\mathcal{M}^p_{\vec{q}}}.$$
(3.2)

To obtain the above theorem, we have to consider the following norm equivalence similar to [104, 132].

Theorem 3.2.7. Let $0 < \vec{q} < \infty$ and $0 < p, s < \infty$ satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}, \quad s \le \min(q_1, \dots, q_n, p).$$

For all $f \in L^0(\mathbb{R}^n)$, it holds that

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \sim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s}$$

The term $||f||_{\mathcal{M}_s^p}$ in Theorem 3.2.7 is an auxiliary one although this explains how Morrey spaces can be used to control operators acting on Lebesgue spaces. We can remove this term under a reasonable condition using the idea by Fujii [38].

Theorem 3.2.8. Let $0 < \vec{s} \le \vec{q} < \infty$ and 0 satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}.$$

Assume that $f \in L^0(\mathbb{R}^n)$ satisfies

$$m_f(2^\ell Q) \to 0$$

as $\ell \to \infty$ for any $Q \in \mathcal{Q}$ and for some medians $\{m_f(2^\ell Q)\}_{\ell \in \mathbb{N}_0}$. Then we have

$$\|f\|_{\mathcal{M}^p_{\vec{s}}} \lesssim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{s}}} \le \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Meanwhile, the condition proposed by Fujii [38] can be verified as follows.

Lemma 3.2.9. Let $f \in L^0(\mathbb{R}^n)$. Assume that $Mf \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ for some $0 < \vec{q} < \infty$ and 0 satisfying

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}.$$

For any $Q \in \mathcal{Q}$ and any medians $\{m_f(2^{\ell}Q)\}_{\ell \in \mathbb{N}_0}$, it holds that

$$\lim_{\ell \to \infty} m_f(2^\ell Q) = 0$$

Proof. For $0 < \lambda < 2^{-1}$, we have

$$|m_f(2^{\ell}Q)| \le (f \cdot \chi_{2^{\ell}Q})^* (\lambda |2^{\ell}Q|) \le \frac{1}{\lambda} \inf_{x \in 2^{\ell}Q} Mf(x).$$

Then,

$$|m_f(2^{\ell}Q)| \lesssim |2^{\ell}Q|^{-\frac{1}{n}\sum_{j=1}^{n}\frac{1}{q_j}} ||Mf\chi_{2^{\ell}Q}||_{\vec{q}} \le |2^{\ell}Q|^{-\frac{1}{p}} ||Mf||_{\mathcal{M}_{\vec{q}}^p}$$

Thus, we obtain $\lim_{\ell \to \infty} m_f(2^{\ell}Q) = 0$.

At first, we show Theorem 3.2.6 applying Theorems 3.2.7 and 3.2.8.

Proof of Theorem 3.2.6. By Theorem 3.2.7, we easily know that

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \sim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s} \ge \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Since $Mf \in \mathcal{M}_{q\bar{q}}^{p_0}(\mathbb{R}^n)$ for some $0 < p_0 < \infty$ and $q\bar{q} \in (0,\infty)^n$, we have $\lim_{\ell \to \infty} m_f(2^\ell Q) = 0$. Thus, we are in position to use Theorem 3.2.8. Since $s \leq \min(q_1,\ldots,q_n,p)$, combining Theorems 3.2.7 and 3.2.8, we have

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \sim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s} = \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \left\|M^{\#,d}_{\lambda}f\right\|_{\mathcal{M}^p_{\vec{q}}}$$

Hence we obtain the left equivalence in (3.2). Meanwhile, the right inequality in (3.2) follows from the pointwise estimate (3.1).

We move on to the proofs of Theorems 3.2.7 and 3.2.8. First, we prepare some lemmas to show Theorem 3.2.7 and give its proof. After that, we prove Theorem 3.2.8.

The following estimates are significant for the proof of Theorem 3.2.7. First, we show that the dyadic local sharp maximal operator is bounded on mixed Morrey spaces.

Proposition 3.2.10. Let $0 < \vec{q} < \infty$ and 0 satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j},$$

and $0 < \lambda < 2^{-1}$. Then, for $f \in L^0(\mathbb{R}^n)$, we have

$$\left\|M_{\lambda}^{\#,d}f\right\|_{\mathcal{M}^{p}_{\vec{q}}} \lesssim \|f\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

Proof. In [104], it is known that

$$M_{\lambda}^{\#,d}f(x) \lesssim M^{(\eta)}f(x) \quad (x \in \mathbb{R}^n)$$

for any $\eta > 0$. By virtue of Theorem 2.1.21, taking $\eta < \min(q_1, \ldots, q_n, p)$, we have

$$\left\| M_{\lambda}^{\#,d} f \right\|_{\mathcal{M}^{p}_{\vec{q}}} \lesssim \left\| M^{(\eta)} f \right\|_{\mathcal{M}^{p}_{\vec{q}}} = \left\| M[|f|^{\eta}] \right\|_{\mathcal{M}^{\frac{p}{\eta}}_{\frac{q}{\eta}}}^{\frac{1}{\eta}} \lesssim \left\| |f|^{\eta} \right\|_{\mathcal{M}^{\frac{p}{\eta}}_{\frac{q}{\eta}}}^{\frac{1}{\eta}} = \left\| f \right\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

To prove " \leq " part of Theorem 3.2.7, we evaluate $||f\chi_Q||_{\vec{q}}$ using the dyadic local sharp maximal operator. Here and below, let $\lambda_n = 2^{-n-2}$.

Theorem 3.2.11. Let $0 < \vec{q} < \infty$ and $0 < p, r < \infty$ satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j},$$

and $Q_0 \in \mathcal{Q}$. Then, for all $f \in L^0(\mathbb{R}^n)$, we have

$$\|f\chi_{Q_0}\|_{\vec{q}} \lesssim \left\| \left(M_{\lambda_n;Q_0}^{\#,d} f \right) \chi_{Q_0} \right\|_{\vec{q}} + |Q_0|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{q_j}} \left(\frac{1}{|Q_0|} \int_{Q_0} |f(x)|^r \mathrm{d}x \right)^{\frac{1}{r}}.$$

Proof. We take a median $m_f(Q_0)$ and use the quasi-triangle inequality to get

$$\|f\chi_{Q_0}\|_{\vec{q}} \lesssim \|[f - m_f(Q_0)]\chi_{Q_0}\|_{\vec{q}} + |Q_0|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{q_j}} |m_f(Q_0)|.$$

First, we estimate the first term. Applying Theorem 3.2.5, we obtain

$$\|[f - m_f(Q_0)]\chi_{Q_0}\|_{\vec{q}} \lesssim \left\| \left(M_{\lambda_n;Q_0}^{\#,d} f \right) \chi_{Q_0} \right\|_{\vec{q}} + \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f;Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{q}}.$$

Since $\{Q_j^k\}_{k\in\mathbb{N}_0, j\in J_k}$ is a sparse family, we have $|Q_j^k| \leq 2|Q_j^k \cap \Omega_{k+1}^c|$ by Lemma 3.2.4. This implies a pointwise estimate

$$\chi_{Q_j^k}(x) \le 2M[\chi_{Q_j^k \cap \Omega_{k+1}^c}](x).$$

Let $\eta > \max(1, q_1^{-1}, \dots, q_n^{-1})$. By Proposition 2.1.24, it follows that

$$\left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f; Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{q}} \lesssim \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f; Q_j^k) M[\chi_{Q_j^k \cap \Omega_{k+1}^c}]^\eta \right) \chi_{Q_0} \right\|_{\vec{q}} \\ \lesssim \left\| \sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f; Q_j^k) \chi_{Q_j^k \cap \Omega_{k+1}^c} \right\|_{\vec{q}}.$$

We deduce from the disjointness of $\{Q_j^k \cap \Omega_{k+1}^c\}_{k \in \mathbb{N}_0, j \in J_k}$ and the definition of ω_λ that

$$\omega_{\lambda_n}(f;Q_j^k)\chi_{Q_j^k} \le M_{\lambda_n;Q_0}^{\#,d}f, \quad \sum_{k\in\mathbb{N}_0}\sum_{j\in J_k}\chi_{Q_j^k\cap\Omega_{k+1}^c} \le \chi_{Q_0}.$$

Then we obtain

$$\left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda_n}(f; Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{q}} \lesssim \left\| \left(M_{\lambda_n; Q_0}^{\#, d} f \right) \chi_{Q_0} \right\|_{\vec{q}}.$$

Next, we evaluate the second term. For, $0<\lambda<2^{-1},$ we have

$$|m_f(Q_0)| \le (f \cdot \chi_{Q_0})^* (\lambda |Q_0|) \le \left(\frac{1}{\lambda |Q_0|} \int_0^{\lambda |Q_0|} (f \cdot \chi_{Q_0})^* (t)^r dt\right)^{\frac{1}{r}} \lesssim \left(\frac{1}{|Q_0|} \int_{Q_0} |f(x)|^r dx\right)^{\frac{1}{r}}.$$

Thus, combining the two estimates gives the desired result.

By virtue of Theorem 3.2.11, we have the following norm estimate.

Corollary 3.2.12. Let $0 < \vec{q} < \infty$ and $0 < s < \infty$. Moreover, let 0 satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}, \quad 0 < s \le p.$$

Then, for $f \in L^0(\mathbb{R}^n)$, we have

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \left\|M^{\#,d}_{\lambda_n}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s}$$

So we turn to the proofs of Theorems 3.2.7 and 3.2.8.

Proof of Theorem 3.2.7. First, Since $0 < s \le \min\{q_1, \ldots, q_n, p\}$, we have

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \left\|M^{\#,d}_{\lambda_n}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s}$$

by Corollary 3.2.12. Conversely, combining Proposition 3.2.10, and embedding

$$\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) \hookrightarrow \mathcal{M}^p_s(\mathbb{R}^n),$$

we obtain

$$\left\|M_{\lambda_n}^{\#,d}f\right\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s} \lesssim \|f\|_{\mathcal{M}^p_{\vec{q}}} + \|f\|_{\mathcal{M}^p_s} \lesssim \|f\|_{\mathcal{M}^p_{\vec{q}}},$$

as desired.

Next, we give the proof of Theorem 3.2.8.

Proof of Theorem 3.2.8. Fix any $Q_0 \in \mathcal{Q}$. Then,

$$\begin{aligned} |Q_0|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \|f\chi_{Q_0}\|_{\vec{s}} &\lesssim |Q_0|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \left\| [f - m_f(2^\ell Q_0)]\chi_{Q_0} \right\|_{\vec{s}} \\ &+ |Q_0|^{\frac{1}{p}} |m_f(2^\ell Q_0)|. \end{aligned}$$

By the assumption, it follows that

$$|Q_0|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \|f\chi_{Q_0}\|_{\vec{s}} \lesssim \limsup_{\ell \to \infty} |Q_0|^{\frac{1}{p} - \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \left\| [f - m_f(2^\ell Q_0)]\chi_{Q_0} \right\|_{\vec{s}}.$$
 (3.3)

We use Theorem 3.2.5, and obtain

$$\left\| [f - m_f(2^\ell Q_0)] \chi_{Q_0} \right\|_{\vec{s}} \leq \left\| \left(M_\lambda^{\#,d} f \right) \chi_{Q_0} \right\|_{\vec{s}} + \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_\lambda(f; Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{s}}.$$
(3.4)

Here, we remark that the family $\{Q_j^k\}_{k\in\mathbb{N}_0,j\in J_k}\subset \mathcal{D}(2^\ell Q_0)$ is a sparse family generated by $2^\ell Q_0$. To evaluate the second term of (3.4), we have only to calculate the following two terms:

$$\mathbf{I} = \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k : Q_j^k \subset Q_0} \omega_\lambda(f; Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{s}},$$
$$\mathbf{II} = \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k : Q_j^k \supseteq Q_0} \omega_\lambda(f; Q_j^k) \chi_{Q_j^k} \right) \chi_{Q_0} \right\|_{\vec{s}}.$$

We can handle this with a similar argument to the proof of Theorem 3.2.11, that is, due to Lemma 3.2.4 and the boundedness of the Hardy–Littlewood maximal operator. Thus, we have

$$\mathbf{I} \lesssim \left\| \left(\sum_{k \in \mathbb{N}_0} \sum_{j \in J_k} \omega_{\lambda}(f; Q_j^k) \chi_{Q_j^k \cap \Omega_{k+1}^c} \right) \chi_{Q_0} \right\|_{\vec{s}}.$$

Note that the summation is taken over the cubes contained in Q_0 . Then, we have

$$\mathbf{I} \lesssim \left\| \left(M_{\lambda}^{\#,d} f \right) \chi_{Q_0} \right\|_{\vec{s}},$$

thanks to the disjointness of $\{Q_j^k \cap \Omega_{k+1}^c\}_{k \in \mathbb{N}_0, j \in J_k}$.

Meanwhile, by recalling that $Q_j^k \subset 2^\ell Q_0$ and the dyadic property, we can rewrite the summation of II as follows.

$$\Pi \leq \left\| \left(\sum_{m=1}^{\infty} \omega_{\lambda}(f; Q_0^{(m)}) \chi_{Q_0^{(m)}} \right) \chi_{Q_0} \right\|_{\vec{s}}.$$

Here, $Q_0^{(m)}$ denotes the dyadic *m*-th ancestor of Q_0 . Namely, $Q_0^{(m)}$ is a unique dyadic cube with respect to $2^{\ell}Q_0$ whose side length is $2^{m}\ell(Q_0)$ and containing Q_0 . Then, by Example 2.1.32, we see that

$$\begin{split} \mathrm{II} &\leq \|\chi_{Q_0}\|_{\vec{s}} \sum_{m=1}^{\infty} \omega_{\lambda}(f; Q_0^{(m)}) \\ &= |Q_0|^{\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \sum_{m=1}^{\infty} \left|Q_0^{(m)}\right|^{-\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \left\| \left(\omega_{\lambda}(f; Q_0^{(m)})\right) \chi_{Q_0^{(m)}} \right\|_{\vec{s}}. \end{split}$$

By virtue of the definition of the local sharp maximal operator $M_{\lambda}^{\#,d}$, we have

$$\begin{split} \mathrm{II} &\leq |Q_0|^{\frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j}\right)} \sum_{m=1}^\infty \left| Q_0^{(m)} \right|^{-\frac{1}{p}} \left\| M_\lambda^{\#,d} f \right\|_{\mathcal{M}^p_{\vec{s}}} \\ &= \left(\sum_{m=1}^\infty 2^{-\frac{m}{p}} \right) \left| Q_0 \right|^{-\frac{1}{p} + \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j} \right)} \left\| M_\lambda^{\#,d} f \right\|_{\mathcal{M}^p_{\vec{s}}} \sim \left| Q_0 \right|^{-\frac{1}{p} + \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j} \right)} \left\| M_\lambda^{\#,d} f \right\|_{\mathcal{M}^p_{\vec{s}}}. \end{split}$$

Thus, combining estimates I and II, we obtain

$$\left\| [f - m_f(2^{\ell}Q_0)]\chi_{Q_0} \right\|_{\vec{s}} \lesssim \left\| \left(M_{\lambda}^{\#,d}f \right) \chi_{Q_0} \right\|_{\vec{s}} + \left| Q_0 \right|^{-\frac{1}{p} + \frac{1}{n} \left(\sum_{j=1}^n \frac{1}{s_j} \right)} \left\| M_{\lambda}^{\#,d}f \right\|_{\mathcal{M}^p_{\vec{s}}}.$$
 (3.5)

Therefore, by (3.3) and (3.5), we obtain the desired result.

3.2.3 The relation to commutators and the sharp maximal function on mixed Morrey spaces

First, we check that the commutator $[b, I_{\alpha}]f$ is well defined for any $f \in \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$ and $b \in BMO(\mathbb{R}^{n})$.

Lemma 3.2.13. Let $0 < \alpha < n, 1 < \vec{q} < \infty, 1 < p < \frac{n}{\alpha}, 1 < r < \infty, and <math>\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$. Also, assume that

$$\frac{1}{r} = \frac{1}{p} - \frac{\alpha}{n}.$$

For any $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, $b \in BMO(\mathbb{R}^n)$, $x \in \mathbb{R}^n$ and $Q \in \mathcal{Q}$ containing x, we have

$$\int_{\mathbb{R}^n \setminus 2Q} \frac{|b(x) - b(y)|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y \lesssim |Q|^{-\frac{1}{r}} \, \|f\|_{\mathcal{M}^p_{\vec{q}}} \left(\|b\|_{\mathrm{BMO}} + |b(x) - b_Q| \right)$$

Proof. By the triangle inequality, it follows that

$$\begin{split} &\int_{\mathbb{R}^n \setminus 2Q} \frac{|b(x) - b(y)|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y \\ &\leq \int_{\mathbb{R}^n \setminus 2Q} \frac{|b(x) - b_Q|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y + \int_{\mathbb{R}^n \setminus 2Q} \frac{|b(y) - b_Q|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y \\ &\equiv \mathrm{I} + \mathrm{II}. \end{split}$$

Note that if $x \in Q$ and $y \in 2^{j+1}Q \setminus 2^jQ$ for each $j \ge 1$, then we have $|x-y| \lesssim 2^{j+1}\ell(Q)$.

We shall estimate I. By the above observation, we estimate

$$\begin{split} \mathbf{I} &= |b(x) - b_Q| \sum_{j=1}^{\infty} \int_{2^{j+1}Q \setminus 2^{j}Q} \frac{1}{|x-y|^{n-\alpha}} |f(y)| \mathrm{d}y \\ &\lesssim |b(x) - b_Q| \sum_{j=1}^{\infty} \int_{2^{j+1}Q \setminus 2^{j}Q} \frac{1}{(2^{j+1}\ell(Q))^{n-\alpha}} |f(y)| \mathrm{d}y \\ &\le |b(x) - b_Q| \sum_{j=1}^{\infty} |2^{j+1}Q|^{-1+\frac{\alpha}{n}} \int_{2^{j+1}Q} |f(y)| \mathrm{d}y. \end{split}$$

Since $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) \hookrightarrow \mathcal{M}^p_1(\mathbb{R}^n)$, we obtain

$$I \leq |b(x) - b_Q| \sum_{j=1}^{\infty} |2^{j+1}Q|^{-1+\frac{\alpha}{n}} |2^{j+1}\ell(Q)|^{1-\frac{1}{p}} ||f||_{\mathcal{M}_1^p}$$
$$= \left(\sum_{j=1}^{\infty} 2^{(j+1)(-\frac{1}{r})}\right) |b(x) - b_Q| \cdot |Q|^{-\frac{1}{r}} ||f||_{\mathcal{M}_{\vec{q}}^p}$$
$$\lesssim |b(x) - b_Q| \cdot |Q|^{-\frac{1}{r}} ||f||_{\mathcal{M}_{\vec{q}}^p}.$$

Here we use the convergence of the series $\sum_{j=1}^{\infty} 2^{(j+1)(-\frac{1}{r})}$.

Next, we consider the second term II. Using Hölder's inequality, we have

$$\begin{split} \mathrm{II} &= \sum_{j=1}^{\infty} \int_{2^{j+1}Q \setminus 2^{j}Q} \frac{|b(y) - b_Q|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y \\ &\lesssim \sum_{j=1}^{\infty} |2^{j+1}Q|^{-1 + \frac{\alpha}{n}} \int_{2^{j+1}Q} |b(y) - b_Q| |f(y)| \mathrm{d}y \\ &\leq \sum_{j=1}^{\infty} |2^{j+1}Q|^{\frac{\alpha}{n}} \left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |b(y) - b_Q|^{s'} \mathrm{d}y \right)^{\frac{1}{s'}} \left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |f(y)|^s \mathrm{d}y \right)^{\frac{1}{s}} \end{split}$$

for some s > 1. Since $|b_{2^{j+1}Q} - b_Q| \lesssim j ||b||_{BMO}$, we get

$$\left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |b(y) - b_Q|^{s'} dy\right)^{\frac{1}{s'}} \leq \left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |b(y) - b_{2^{j+1}Q}|^{s'} dy\right)^{\frac{1}{s'}} + |b_{2^{j+1}Q} - b_Q| \leq (1+j) \|b\|_{BMO}.$$
(3.6)

Furthermore, since $M^{(s)}$ is bounded on $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, we have

$$\begin{aligned} \left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |f(y)|^s \mathrm{d}y\right)^{\frac{1}{s}} &\leq \frac{\left\|\chi_{2^{j+1}Q}\right\|_{\mathcal{M}^p_{\vec{q}}}}{\left\|\chi_{2^{j+1}Q}\right\|_{\mathcal{M}^p_{\vec{q}}}} \inf_{x \in 2^{j+1}Q} M^{(s)} f(x) \\ &\leq \left\|\chi_{2^{j+1}Q}\right\|_{\mathcal{M}^p_{\vec{q}}}^{-1} \left\|M^{(s)}f\right\|_{\mathcal{M}^p_{\vec{q}}} \\ &\lesssim |2^{j+1}Q|^{-\frac{1}{p}} \|f\|_{\mathcal{M}^p_{\vec{q}}}.\end{aligned}$$

Thus,

$$\begin{aligned} \Pi &\lesssim \sum_{j=1}^{\infty} |2^{j+1}Q|^{\frac{\alpha}{n} - \frac{1}{p}} (1+j) \|b\|_{BMO} \|f\|_{\mathcal{M}^{p}_{\vec{q}}} \\ &= \left(\sum_{j=1}^{\infty} 2^{-\frac{1}{r}(j+1)} (1+j)\right) |Q|^{-\frac{1}{r}} \|b\|_{BMO} \|f\|_{\mathcal{M}^{p}_{\vec{q}}}. \end{aligned}$$

Since $\sum_{j=1}^{\infty} 2^{-\frac{1}{r}(j+1)}(1+j)$ converges, we obtain

$$\mathrm{II} \lesssim |Q|^{-\frac{1}{r}} \|b\|_{\mathrm{BMO}} \|f\|_{\mathcal{M}^p_{\vec{q}}}.$$

Combining these two estimates, we conclude

$$\int_{\mathbb{R}^n \setminus 2Q} \frac{|b(x) - b(y)|}{|x - y|^{n - \alpha}} |f(y)| \mathrm{d}y \lesssim (|b(x) - b_Q| + \|b\|_{\mathrm{BMO}}) |Q|^{-\frac{1}{r}} \|f\|_{\mathcal{M}^p_{\vec{q}}}$$

Remark 3.2.14. Let $f \in \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$, $b \in BMO(\mathbb{R}^{n})$, $x \in \mathbb{R}^{n}$, and $Q \in \mathcal{Q}$ containing x. Then, we can show that $f\chi_{Q} \in L^{\vec{q}}(\mathbb{R}^{n})$ and $(bf)\chi_{Q} \in L^{\vec{r}}(\mathbb{R}^{n})$ for any $1 < \vec{r} < \vec{q}$. In fact, by the definition of the mixed Morrey norm, it is clear that $f\chi_{Q} \in L^{\vec{q}}(\mathbb{R}^{n})$. On the other hand,

$$\|(bf)\chi_Q\|_{\vec{r}} \le \|(b-b_Q)f\chi_Q\|_{\vec{r}} + \|b_Qf\chi_Q\|_{\vec{r}}.$$

By Hölder's inequality, for \vec{s} satisfying $\frac{1}{\vec{s}} + \frac{1}{\vec{q}} = \frac{1}{\vec{r}}$,

$$\begin{split} \|(b-b_Q)f\chi_Q\|_{\vec{r}} &\leq \|(b-b_Q)\chi_Q\|_{\vec{s}} \|f\chi_Q\|_{\vec{q}} \\ &= \frac{\|\chi_Q\|_{\vec{s}}}{\|\chi_Q\|_{\vec{s}}} \|(b-b_Q)\chi_Q\|_{\vec{s}} \|f\chi_Q\|_{\vec{q}} \\ &\leq \|b\|_{\mathrm{BMO}_{\vec{s}}} |Q|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{s_j}} \times \|f\|_{\mathcal{M}^p_{\vec{q}}} |Q|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{q_j} - \frac{1}{p}} \\ &\sim |Q|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{r_j} - \frac{1}{p}} \|b\|_{\mathrm{BMO}} \|f\|_{\mathcal{M}^p_{\vec{q}}}, \end{split}$$

where

$$\|b\|_{\text{BMO}_{\tilde{s}}} \equiv \sup_{Q \in \mathcal{Q}} \frac{1}{\|\chi_Q\|_{\vec{s}}} \|(b - b_Q)\chi_Q\|_{\vec{s}}$$

and since mixed Lebesgue space is a ball Banach function space, we use the characterization of BMO via ball Banach function spaces [68]:

$$\|b\|_{\mathrm{BMO}_{\tilde{\mathbf{s}}}} \sim \|b\|_{\mathrm{BMO}}.$$

Note that other characterizations of BMO can be found in [67, 69] and [59, Theorem 4.11]. Furthermore,

$$\|b_Q f \chi_Q\|_{\vec{r}} \le |Q|^{\frac{1}{n}\sum_{j=1}^n \frac{1}{r_j} - \frac{1}{p}} |b_Q| \|f\|_{\mathcal{M}^p_{\vec{q}}}$$

Thus, $(bf)\chi_Q \in L^{\vec{r}}(\mathbb{R}^n)$. Since $|x-y|^{\alpha-n}$ is integrable on 2Q, we see that

$$|I_{\alpha}(f\chi_{2Q})(x)| \leq \int_{\mathbb{R}^n} \frac{|f(y)|\chi_{2Q}}{|x-y|^{n-\alpha}} \mathrm{d}y \leq ||f\chi_{2Q}||_{\vec{q}} \left\| |x-\cdot|^{\alpha-n}\chi_{2Q} \right\|_{\vec{q}'} < \infty$$

and

$$|I_{\alpha}(bf\chi_{2Q})| \leq \int_{\mathbb{R}^n} \frac{|b(y)f(y)|\chi_{2Q}}{|x-y|^{n-\alpha}} \mathrm{d}y \leq \|bf\chi_{2Q}\|_{\vec{r}} \left\| |x-\cdot|^{\alpha-n}\chi_{2Q} \right\|_{\vec{r}'} < \infty.$$

Hence, $I_{\alpha}(f\chi_{2Q})$ and $I_{\alpha}(bf\chi_{2Q})$ are well defined.

Using this fact and Lemma 3.2.13, we can justify the definition of $[b, I_{\alpha}]f$.

Finally, we evaluate the sharp maximal function of the commutator $[b, I_{\alpha}]f$. The following estimate is also important to show the main theorem.

Lemma 3.2.15. Let $0 < \alpha < n$ and $1 < \eta < \infty$. Then,

$$([b, I_{\alpha}]f)^{\#}(x) \lesssim \|b\|_{\text{BMO}} \left(M^{(\eta)}[I_{\alpha}f](x) + M^{(\eta)}_{\eta\alpha}f(x) \right)$$

for all $b \in BMO(\mathbb{R}^n)$, $f \in \mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)$ and $x \in \mathbb{R}^n$.

Note that similar estimates to Lemma 3.2.15 were proved in [8, 140]. Shirai showed this estimate for $f \in C_c^{\infty}(\mathbb{R}^n)$ [140, Lemma 4.2], while Arai and Nakai showed a similar estimate for the element of generalized Campanato spaces and generalized Morrey spaces [8, Proposition 5.2].

Proof of Lemma 3.2.15. Note that since

$$h^{\#}(x) \sim \sup_{x \in Q} \inf_{c \in \mathbb{C}} \frac{1}{|Q|} \int_{Q} |h(y) - c| \mathrm{d}y,$$

for any locally integrable function h in general, to prove this lemma, we only show that

$$\frac{1}{|Q|} \int_{Q} |[b, I_{\alpha}]f(y) - c| \, \mathrm{d}y \lesssim \|b\|_{\mathrm{BMO}} \left(M^{(\eta)}[I_{\alpha}f](x) + M^{(\eta)}_{\eta\alpha}f(x) \right)$$

for some $c \in \mathbb{C}$. Let $Q \equiv Q(x, \ell(Q))$. First, we decompose

$$f = f_1 + f_2 \equiv f\chi_{2Q} + f\chi_{\mathbb{R}^n \setminus 2Q}.$$

For, $y \in Q$, we define

$$F_{1}(y) \equiv (b(y) - b_{2Q})I_{\alpha}f(y),$$

$$F_{2}(y) \equiv I_{\alpha}[(b - b_{2Q})f_{1}](y),$$

$$F_{3}(y) \equiv I_{\alpha}[(b - b_{2Q})f_{2}](y) - C_{Q},$$

where $C_Q \equiv I_{\alpha}[(b - b_{2Q})f_2](x)$. Then,

$$[b, I_{\alpha}]f + C_Q = [(b - b_{2Q}), I_{\alpha}]f + C_Q = F_1 - F_2 - F_3.$$

Thus, we should evaluate

$$\frac{1}{|Q|} \int_{Q} |F_i(y)| \mathrm{d}y \quad (i = 1, 2, 3).$$

First, we estimate F_1 . By Hölder's inequality, we obtain

$$\frac{1}{|Q|} \int_{Q} |F_{1}(y)| \mathrm{d}y \lesssim \left(\frac{1}{|2Q|} \int_{2Q} |b(y) - b_{2Q}|^{\eta'} \mathrm{d}y\right)^{\frac{1}{\eta'}} \left(\frac{1}{|2Q|} \int_{2Q} |I_{\alpha}f(y)|^{\eta} \mathrm{d}y\right)^{\frac{1}{\eta}} \\
\leq \|b\|_{\mathrm{BMO}} M^{(\eta)}(I_{\alpha}f)(x).$$
(3.7)

Next, we estimate F_2 . By Hölder's inequality and the boundedness of the fractional integral operator, we get

$$\frac{1}{|Q|} \int_{Q} |F_2(y)| \mathrm{d}y \le \frac{1}{|Q|} \|\chi_Q\|_{r'} \|F_2\|_r \lesssim |Q|^{-1 + \frac{1}{r'}} \|(b - b_{2Q})f_1\|_{v_2}$$

where $1 < v < \eta$ satisfies $\frac{1}{r} = \frac{1}{v} - \frac{\alpha}{n}$. Let $\frac{1}{v} = \frac{1}{u} + \frac{1}{\eta}$. By virtue of Hölder's inequality again, we have

$$\frac{1}{|Q|} \int_{Q} |F_{2}(y)| \mathrm{d}y \lesssim |2Q|^{\frac{1}{v} - \frac{1}{r}} \left(\frac{1}{|2Q|} \int_{2Q} |b(y) - b_{2Q}|^{u} \mathrm{d}y \right)^{\frac{1}{u}} \left(\frac{1}{|2Q|} \int_{2Q} |f(y)|^{\eta} \mathrm{d}y \right)^{\frac{1}{\eta}} \\
\leq \|b\|_{\mathrm{BMO}} \left(\frac{1}{|2Q|^{1 - \frac{\eta\alpha}{n}}} \int_{2Q} |f(y)|^{\eta} \mathrm{d}y \right)^{\frac{1}{\eta}} \leq \|b\|_{\mathrm{BMO}} M_{\eta\alpha}^{(\eta)} f(x). \quad (3.8)$$

Finally, we estimate F_3 . If we write out fully the definition of I_{α} ,

$$F_{3}(y) = I_{\alpha}[(b - b_{2Q})f_{2}](y) - I_{\alpha}[(b - b_{2Q})f_{2}](x)$$

=
$$\int_{\mathbb{R}^{n}} \left(\frac{1}{|y - z|^{n - \alpha}} - \frac{1}{|x - z|^{n - \alpha}}\right) (b(z) - b_{2Q})f_{2}(z)dz.$$

Since $z \in \mathbb{R}^n \setminus 2Q$ and $y \in Q$, we get $|x - z| \lesssim |y - z|$. Hence,

$$\left| \frac{1}{|y-z|^{n-\alpha}} - \frac{1}{|x-z|^{n-\alpha}} \right| = (n-\alpha) \left| \int_{|x-z|}^{|y-z|} \frac{1}{t^{n+1-\alpha}} dt \right|$$
$$\leq \min(|y-z|, |x-z|)^{-n-1+\alpha} |x-y| \lesssim \frac{|x-y|}{|x-z|^{n+1-\alpha}}.$$

Therefore,

$$|F_{3}(y)| \lesssim \int_{\mathbb{R}^{n} \setminus 2Q} \frac{|x-y|}{|x-z|^{n+1-\alpha}} |b(z) - b_{2Q}| |f(z)| dz$$

= $\sum_{j=1}^{\infty} \int_{2^{j+1}Q \setminus 2^{j}Q} \frac{|x-y|}{|x-z|^{n+1-\alpha}} |b(z) - b_{2Q}| |f(z)| dz$
 $\lesssim \sum_{j=1}^{\infty} \frac{\ell(Q)}{(2^{j+1}\ell(Q))^{n-\alpha+1}} \int_{2^{j+1}Q \setminus 2^{j}Q} |b(z) - b_{2Q}| |f(z)| dz.$

Next, Hölder's inequality yields

$$\frac{\ell(Q)}{(2^{j+1}\ell(Q))^{n-\alpha+1}} \int_{2^{j+1}Q\setminus 2^{j}Q} |b(z) - b_{2Q}| |f(z)| dz
\leq \frac{\ell(Q)}{(2^{j+1}\ell(Q))^{n-\alpha+1}} \left(\int_{2^{j+1}Q} |b(z) - b_{2Q}|^{\eta'} dz \right)^{\frac{1}{\eta'}} \left(\int_{2^{j+1}Q} |f(z)|^{\eta} dz \right)^{\frac{1}{\eta}}
= \frac{1}{2^{j+1}} \left(\frac{1}{|2^{j+1}Q|} \int_{2^{j+1}Q} |b(z) - b_{2Q}|^{\eta'} dz \right)^{\frac{1}{\eta'}} \left(\frac{1}{|2^{j+1}Q|^{1-\frac{\eta\alpha}{n}}} \int_{2^{j+1}Q} |f(z)|^{\eta} dz \right)^{\frac{1}{\eta}}.$$

By the definition of M_{α} , we have

$$\left(\frac{1}{|2^{j+1}Q|^{1-\frac{\eta\alpha}{n}}}\int_{2^{j+1}Q}|f(z)|^{\eta}\mathrm{d}z\right)^{\frac{1}{\eta}} \le M_{\eta\alpha}^{(\eta)}f(x).$$

Moreover, in the same way as (3.6), we obtain

$$\left(\frac{1}{|2^{j+1}Q|}\int_{2^{j+1}Q}|b(z)-b_{2Q}|^{\eta'}\mathrm{d}z\right)^{\frac{1}{\eta'}} \lesssim (1+j)\|b\|_{\mathrm{BMO}}.$$

Since $\sum_{j=1}^{\infty} \frac{1+j}{2^{j+1}}$ converges, we conclude

$$|F_3(y)| \lesssim \|b\|_{\text{BMO}} \cdot M_{\eta\alpha}^{(\eta)} f(x).$$
(3.9)

Since estimate (3.9) is independent of y, we have

$$\frac{1}{|Q|} \int_{Q} |F_3(y)| \mathrm{d}y \lesssim ||b||_{\mathrm{BMO}} \cdot M_{\eta\alpha}^{(\eta)} f(x).$$
(3.10)

Combining estimates (3.7), (3.8), and (3.10) provides the desired result.

3.3Proof of the main theorem (Theorem 3.1.4)

Let us show Theorem 3.1.4.

Proof. (a) \Rightarrow (b): Let $1 < \eta < \min(s_1, \ldots, s_n, r)$ and $f \in \mathcal{M}^p_{\vec{a}}(\mathbb{R}^n)$. Put $s_0 \equiv$ $\min(s_1, \ldots, s_n) \text{ and } q_0 \equiv \min(q_1, \ldots, q_n). \text{ Since } f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) \hookrightarrow \mathcal{M}^p_{q_0}(\mathbb{R}^n), [b, I_\alpha] f \in \mathcal{M}^r_{s_0}(\mathbb{R}^n) = \mathcal{M}^r_{(s_0, \ldots, s_0)}(\mathbb{R}^n) \text{ by the result for classical Morrey spaces. Then, we}$ see that $M([b, I_{\alpha}]f) \in \mathcal{M}^{r}_{(s_{0}, \ldots, s_{0})}(\mathbb{R}^{n})$. Thus, the assumption of Corollary 3.2.6 is

times

satisfied. By virtue of Corollary 3.2.6 and Lemma 3.2.15, we have

$$\begin{split} \|[b, I_{\alpha}](f)\|_{\mathcal{M}_{\vec{s}}^{r}} \lesssim \left\| ([b, I_{\alpha}]f)^{\#} \right\|_{\mathcal{M}_{\vec{s}}^{r}} \lesssim \|b\|_{\mathrm{BMO}} \left\| M^{(\eta)}[I_{\alpha}f] + M^{(\eta)}_{\eta\alpha}f \right\|_{\mathcal{M}_{\vec{s}}^{r}} \\ & \leq \|b\|_{\mathrm{BMO}} \left\{ \|M^{(\eta)}[I_{\alpha}f]\|_{\mathcal{M}_{\vec{s}}^{r}} + \left\|M^{(\eta)}_{\eta\alpha}f\right\|_{\mathcal{M}_{\vec{s}}^{r}} \right\} \\ & \lesssim \|b\|_{\mathrm{BMO}} \left\{ \|I_{\alpha}(|f|)\|_{\mathcal{M}_{\vec{s}}^{r}} + \|I_{\eta\alpha}(|f|^{\eta})\|_{\mathcal{M}_{\vec{s}}^{\frac{n}{\eta}}}^{\frac{1}{\eta}} \right\}. \end{split}$$

Using Theorem 2.1.22, we conclude

$$\|[b, I_{\alpha}](f)\|_{\mathcal{M}^{r}_{\vec{s}}} \lesssim \|b\|_{\text{BMO}} \left\{ \|f\|_{\mathcal{M}^{p}_{\vec{q}}} + \||f|^{\eta}\|_{\mathcal{M}^{\frac{p}{\eta}}_{\vec{q}}}^{\frac{1}{\eta}} \right\} = \|b\|_{\text{BMO}} \|f\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

(b) \Rightarrow (c): It is clear since only the domain is restricted.

(c) \Rightarrow (d): Using the embedding $\mathcal{M}^p_{\vec{s}}(\mathbb{R}^n) \hookrightarrow \mathcal{M}^p_1(\mathbb{R}^n)$ and (c), we have

$$\|[b, I_{\alpha}]f\|_{\mathcal{M}_{1}^{p}} \leq \|[b, I_{\alpha}]f\|_{\mathcal{M}_{\vec{s}}^{p}} \lesssim \|f\|_{\widetilde{\mathcal{M}}_{\vec{q}}^{p}}$$

for $f \in \widetilde{\mathcal{M}}^p_{\vec{a}}(\mathbb{R}^n)$. Thus we obtain (d).

(d) \Rightarrow (a): We use the same method as Janson [70]. Choose $z_0 \in \mathbb{R}^n$ such that $|z_0| = 5$. Since $0 \notin Q(z_0, 2), |x|^{n-\alpha} \in C^{\infty}(Q(z_0, 2))$ for $x \in Q(z_0, 2)$. Hence, we choose a function $\varphi \in C^{\infty}(\mathbb{R}^n)$ with a π periodicity and satisfying $\varphi(x) = |x|^{n-\alpha}$ for all $x \in Q(z_0, 2)$. Then, we can expand this function into the absolutely convergent Fourier series on $Q(z_0, 2)$;

$$|x|^{n-\alpha}\chi_{Q(z_0,2)}(x) = \sum_{m \in \mathbb{Z}^n} a_m e^{2im \cdot x} \chi_{Q(z_0,2)}(x), \qquad (3.11)$$

with $\sum_{m \in \mathbb{Z}^n} |a_m| < \infty$. For any $x_0 \in \mathbb{R}^n$ and t > 0, let $Q \equiv Q(x_0, t)$ and $Q' \equiv Q(x_0 + z_0 t, t)$. Let

$$s(x) \equiv \operatorname{sgn}\left(\int_{Q'} (b(x) - b(y)) \mathrm{d}y\right).$$

If $x \in Q$ and $y \in Q'$, then $\frac{y-x}{t} \in Q(z_0, 2)$. Hence, we have

$$\begin{split} \int_{Q} |b(x) - b_{Q'}| \mathrm{d}x &= \int_{Q} (b(x) - b_{Q'}) \overline{s(x)} \mathrm{d}x \\ &= \frac{1}{|Q'|} \int_{Q} \overline{s(x)} \left(\int_{Q'} (b(x) - b(y)) \mathrm{d}y \right) \mathrm{d}x \\ &= \frac{t^{n-\alpha}}{t^n} \int_{Q} \overline{s(x)} \left(\int_{Q'} (b(x) - b(y)) |x - y|^{-n+\alpha} \left| \frac{x - y}{t} \right|^{n-\alpha} \mathrm{d}y \right) \mathrm{d}x. \end{split}$$

By (3.11) and the triangle inequality, we get

$$\begin{split} &\int_{Q} |b(x) - b_{Q'}| \mathrm{d}x \\ &= t^{-\alpha} \sum_{m \in \mathbb{Z}^{n}} \int_{Q} \overline{s(x)} \left(\int_{Q'} (b(x) - b(y)) |x - y|^{-n + \alpha} a_{m} e^{2im \cdot \frac{y}{t}} \mathrm{d}y \right) e^{-2im \cdot \frac{x}{t}} \mathrm{d}x \\ &\leq t^{-\alpha} \sum_{m \in \mathbb{Z}^{n}} \left| a_{m} \int_{\mathbb{R}^{n}} \overline{s(x)} [b, I_{\alpha}] (e^{2im \cdot \frac{i}{t}} \chi_{Q'})(x) \chi_{Q}(x) e^{-2im \cdot \frac{x}{t}} \mathrm{d}x \right| \\ &\leq t^{-\alpha} \sum_{m \in \mathbb{Z}^{n}} |a_{m}| \int_{Q} \left| [b, I_{\alpha}] (e^{2im \cdot \frac{i}{t}} \chi_{Q'})(x) \right| \mathrm{d}x \\ &\leq t^{-\alpha} \sum_{m \in \mathbb{Z}^{n}} |a_{m}| |Q|^{-\frac{1}{r}+1} \left\| [b, I_{\alpha}] (e^{2im \cdot \frac{i}{t}} \chi_{Q'}) \right\|_{\mathcal{M}_{q}^{r}} \\ &\leq t^{-\alpha} \sum_{m \in \mathbb{Z}^{n}} |a_{m}| \| [b, I_{\alpha}] \|_{\tilde{\mathcal{M}}_{q}^{p} \to \mathcal{M}_{1}^{r}} t^{\frac{n}{p}} \cdot t^{\frac{n}{r'}} \sim t^{n} \| [b, I_{\alpha}] \|_{\tilde{\mathcal{M}}_{q}^{p} \to \mathcal{M}_{1}^{r}} . \end{split}$$

Thus, we have

$$\frac{1}{|Q|} \int_{Q} |b(x) - b_Q| \mathrm{d}x \le \frac{2}{|Q|} \int_{Q} |b(x) - b_{Q'}| \mathrm{d}x \lesssim \|[b, I_\alpha]\|_{\tilde{\mathcal{M}}^p_{\vec{q}} \to \mathcal{M}^r_1}.$$

This implies that $b \in BMO(\mathbb{R}^n)$ since Q is an arbitrary cube.

Chapter 4

A characterization of Morrey spaces associated with Banach lattice in terms of pointwise multiplier spaces

4.1 Introduction and theorems

In [58], Ho defined vector-valued Morrey spaces. In this chapter, inspired these ideas, we will define Morrey spaces associated to general Banach lattices and establish that these spaces arise naturally as multiplier spaces from microlocal Besov spaces to Banach lattices. Recall that a *Banach (function) lattice* on \mathbb{R}^n is a Banach space $(E, \|\cdot\|_E)$ contained in $L^0(\mathbb{R}^n)$, the linear space of all measurable functions, such that, for all $f, g \in E$, the implication " $|f| \leq |g| \Rightarrow ||f||_E \leq ||g||_E$ " holds.

To define Morrey spaces associated to general Banach lattices, we assume the following:

Assumption 4.1.1. The Banach lattice $E(\mathbb{R}^n)$ is translation invariant, that is, let $E(\mathbb{R}^n)$ be a Banach lattice be such that

$$||f(\cdot - x)||_E = ||f||_E$$

for all $f \in E(\mathbb{R}^n)$ and $x \in \mathbb{R}^n$.

A direct consequence of the translation invariance is that $\|\chi_{3Q}\|_E \leq \|\chi_Q\|_E$ for all cubes Q, where 3Q denotes the triple of Q, that is, 3Q is a cube which is concentric to Q and has volume $3^n |Q|$. With a natural modification we can include the case of variable exponents but here for the sake of simplicity we do not do this. These assumptions are postulated so as to simplify matters. Nevertheless, as our examples show, we have translation invariant many function spaces.

Based on [58, Definition 2.6], we define $\mathcal{M}^p_E(\mathbb{R}^n)$ to be the set of all measurable functions f for which

$$\|f\|_{\mathcal{M}_E^p} \equiv \sup_Q |Q|^{\frac{1}{p}} \left(\frac{1}{\|\chi_Q\|_E} \|f\chi_Q\|_E\right)$$

is finite, where Q moves over all cubes whose edges are parallel to the coordinate axes. As we will show in Section 4.3, Morrey spaces associated to general Banach lattices realize mixed Morrey spaces [106], Morrey–Lorentz spaces [118] and Orlicz–Morrey spaces (of the third kind) [26].

From Assumption 4.1.1, we learn that the following dyadic Morrey norm is equivalent to the original norm

$$\|f\|_{\mathcal{M}^p_E} \sim \sup_{Q \in \mathcal{D}} |Q|^{\frac{1}{p}} \left(\frac{1}{\|\chi_Q\|_E} \|f\chi_Q\|_E\right),$$

where \mathcal{D} denotes the set of all dyadic cubes (see Section 1.3 (6)).

With the definition of dyadic cubes in mind, we formulate the definition of microlocal Besov spaces. Although we gave it in Section 2.2.2, we will adopt Theorem 2.2.10 ([74, Theorem 1]) with p = q = 1, s = 0, $K = [\alpha_2 + 1]$ and $L = [\alpha_1 + 1]$ as a definition of the microlocal Besov space (we write $\dot{B}_{11}^w(\mathbb{R}^n)$) in this chapter.

Definition 4.1.2. Let $w \equiv \{w_j\}_{j=-\infty}^{\infty} \subset \mathbb{R}$ be a positive sequence, and let $F(\mathbb{R}^n)$ be a Banach function lattice satisfying

$$\|\chi_{Q_{jm}}\|_F \lesssim w_j |Q_{j0}| \quad (j \in \mathbb{Z}, m \in \mathbb{Z}^n)$$

and

$$2^{-\alpha_1 j} w_j \le w_{j+1} \le 2^{\alpha_2 j} w_j \quad (j \in \mathbb{Z}),$$

where α_1 and α_2 are fixed parameters. One defines the microlocal homogeneous Besov space $\dot{B}_{11}^w(\mathbb{R}^n)$ by the set of all $f \in F(\mathbb{R}^n)$ for which it can be written as

$$f(x) = \sum_{j=-\infty}^{\infty} \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm}(x)$$

for almost every $x \in \mathbb{R}^n$, where for all $j \in \mathbb{Z}$ and $m \in \mathbb{Z}^n$, we have a collection $\{a_{jm}\}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ of C^{∞} -functions and a collection $\{\lambda_{jm}\}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ of complex constants satisfying

$$|\partial^{\alpha} a_{jm}| \le 2^{j|\alpha|} \chi_{3Q_{jm}}, \quad \int_{\mathbb{R}^n} x^{\beta} a_{jm}(x) \mathrm{d}x = 0, \quad \sum_{j=-\infty}^{\infty} 2^{-jn} w_j \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \right) < \infty$$

for all multiindices α with $|\alpha| \leq [\alpha_2 + 1]$ and for all multiindices β with $|\beta| \leq [\alpha_1 + 1]$. Then the *microlocal Besov norm* $||f||_{\dot{B}^w_{11}}$ is defined as the infimum of

$$\sum_{j=-\infty}^{\infty} 2^{-jn} w_j \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \right)$$

where $\Lambda = {\lambda_{jm}}_{j \in \mathbb{Z}, m \in \mathbb{Z}^n}$ moves over all possible expressions.

In the above, the convergence in $\mathcal{S}'(\mathbb{R}^n)$ is guaranteed thanks to the condition on Λ . The functions a_{jm} are called atoms. Note also that in terms of Definition 2.2.9, $\dot{B}_{11}^w(\mathbb{R}^n) = \dot{B}_{11}^{0,\text{mloc}}(\mathbb{R}^n, w).$

Finally, to state our main results, we turn to the definition of pointwise multipliers spaces. Given Banach spaces $E_1(\mathbb{R}^n)$ and $E_2(\mathbb{R}^n)$ of measurable functions defined on \mathbb{R}^n , we define PWM($E_1(\mathbb{R}^n), E_2(\mathbb{R}^n)$) as follows: A measurable function g is a *pointwise multiplier* from $E_1(\mathbb{R}^n)$ to $E_2(\mathbb{R}^n)$ if the pointwise product $f \cdot g$ belongs to $E_2(\mathbb{R}^n)$ for each $f \in E_1(\mathbb{R}^n)$ and there exists a constant M > 0 such that

$$\|f \cdot g\|_{E_2(\mathbb{R}^n)} \le M \|f\|_{E_1(\mathbb{R}^n)}.$$
(4.1)

One defines a norm on $\text{PWM}(E_1(\mathbb{R}^n), E_2(\mathbb{R}^n))$ by

 $||g||_{\text{PWM}(E_1, E_2)} \equiv \inf\{M > 0 : (4.1) \text{ holds for all } f \in E_1(\mathbb{R}^n)\}$

for $g \in \text{PWM}(E_1(\mathbb{R}^n), E_2(\mathbb{R}^n))$.

A simple example is the case of Lebesgue spaces.

Example 4.1.3. Let $1 \leq p_1, p_2, p_3 \leq \infty$ satisfy $\frac{1}{p_1} + \frac{1}{p_2} = \frac{1}{p_3}$. Then, by Hölder's inequality, we have $PWM(L^{p_1}(\mathbb{R}^n), L^{p_3}(\mathbb{R}^n)) = L^{p_2}(\mathbb{R}^n).$

We also refer to [83, 99] for the case where $E_1(\mathbb{R}^n)$ and $E_2(\mathbb{R}^n)$ are Morrey spaces.

Our main result in this chapter is the following assertion (This is the same one as Theorem 1.2.5).

Theorem 4.1.4. Let $E(\mathbb{R}^n)$ and $F(\mathbb{R}^n)$ be Banach lattices such that

$$\|\chi_{Q_{jm}}\|_F \lesssim \|\chi_{Q_{jm}}\|_E |Q_{j0}|^{-\frac{1}{p}} \quad (j \in \mathbb{Z}, m \in \mathbb{Z}^n).$$

Set

$$w_j \equiv \|\chi_{Q_{j0}}\|_E |Q_{j0}|^{-1-\frac{1}{p}} \quad (j \in \mathbb{Z}).$$

Then $\dot{B}^w_{11}(\mathbb{R}^n)$ is continuously embedded into $F(\mathbb{R}^n)$ and

$$\mathrm{PWM}(\dot{B}_{11}^w(\mathbb{R}^n), E(\mathbb{R}^n)) \approx \mathcal{M}_E^p(\mathbb{R}^n)$$

with equivalence of norms.

In connection with Theorem 4.1.4, we consider the result by Lemarié-Rieusset. Lemarié-Rieusset showed that Morrey spaces arises naturally when we consider the pointwise multipliers from $\dot{B}_{p1}^{s}(\mathbb{R}^{n})$ to $L^{p}(\mathbb{R}^{n})$ with $0 < s \leq \frac{n}{n}$.

Theorem 4.1.5 (cf. [82]). Let $1 \le p < \infty$ and $0 < s \le \frac{n}{p}$. Then $\text{PWM}(\dot{B}^s_{p1}(\mathbb{R}^n), L^p(\mathbb{R}^n)) \approx \mathcal{M}_p^{\frac{n}{s}}(\mathbb{R}^n)$

with equivalence of norms.

Lemarié-Rieusset obtained Theorem 4.1.5 for n = 3 and p = 2 [82, Lemma 6]. A passage to the general case is a minor modification. Here for the sake of completeness, we give a proof in Subsection 4.2.1.

Furthermore, we can generalize the result of Lemarié-Rieusset (Theorem 4.1.5) to multiplier spaces from homogeneous Besov spaces to Morrey spaces. This proof is also given in Subsection 4.2.1.

Theorem 4.1.6. Let $1 \le q \le p < \infty$ and $0 < s \le \frac{n}{q}$ satisfy $0 < \frac{s}{n} - \frac{1}{q} + \frac{1}{p} \le \frac{1}{p}$ Let $\frac{1}{\sigma} \equiv \frac{s}{n} - \frac{1}{q} + \frac{1}{p}.$

Then with equivalence norms

$$\mathrm{PWM}\left(\dot{B}_{q1}^{s}(\mathbb{R}^{n}), \mathcal{M}_{q}^{p}(\mathbb{R}^{n})\right) \approx \mathcal{M}_{\mathcal{M}_{q}^{p}}^{\sigma}(\mathbb{R}^{n}) = \mathcal{M}_{q}^{\sigma}(\mathbb{R}^{n}).$$

The structure of Chapter 4 is as follows: In Section 4.2, we prove our results. Subsection 4.2.1 is devoted to the proof of results for generalized 2-microlocal Besov spaces (Theorem 4.1.4). In Subsection 4.2.2, we prove the results for the pointwise multiplier spaces from classical Besov spaces to Lebesgue spaces (Theorem 4.1.5) and from Besov spaces to Morrey spaces (Theorem 4.1.6). Finally, we give the examples for our results in Section 4.3. We apply our results to Orlicz spaces (Subsection 4.3.1), Lorentz spaces (Subsection 4.3.2), mixed Lebesgue spaces (Subsection 4.3.3), and mixed Morrey spaces (Subsection 4.3.4), respectively.

4.2 Proofs of the main theorems

4.2.1 Pointwise multipliers from generalized 2-microlocal Besov spaces to Banach lattices (Theorem 4.1.4)

We will show the former half of Theorem 4.1.4 to check that $\text{PWM}(\dot{B}_{11}^w(\mathbb{R}^n), E(\mathbb{R}^n))$ is well defined.

Lemma 4.2.1. Let $F(\mathbb{R}^n)$ be a Banach function lattice satisfying

$$\|\chi_{Q_{jm}}\|_F \lesssim 2^{-jn} w_j \quad (j \in \mathbb{Z}, m \in \mathbb{Z}^n).$$

Then $\dot{B}^w_{11}(\mathbb{R}^n) \hookrightarrow F(\mathbb{R}^n)$.

Proof. Let $f \in \dot{B}^w_{11}(\mathbb{R}^n)$ be a decomposition as in Definition 4.1.2. By the triangle inequality, we have

$$\left\| \sum_{j=-\infty}^{\infty} \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right\|_F \leq \sum_{j=-\infty}^{\infty} \sum_{m \in \mathbb{Z}^n} \|\lambda_{jm} a_{jm}\|_F$$
$$\lesssim \sum_{j=-\infty}^{\infty} \sum_{m \in \mathbb{Z}^n} \|\lambda_{jm} \chi_{3Q_{jm}}\|_F$$
$$\lesssim \sum_{j=-\infty}^{\infty} 2^{-jn} w_j \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \right).$$

We prove the latter half of Theorem 4.1.4. Let $f \in \mathcal{M}^p_E(\mathbb{R}^n)$. Let also $g \in \dot{B}^w_{11}(\mathbb{R}^n)$, so that there exist a collection $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of C^{∞} -functions and a collection $\{\lambda_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of complex constants satisfying

$$g = \sum_{j=-\infty}^{\infty} \left(\sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right), \quad \sum_{j=-\infty}^{\infty} 2^{-jn} w_j \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \right) < \infty$$

and $|\partial^{\alpha} a_{jm}| \leq 2^{j|\alpha|} \chi_{3Q_{jm}}$ for all $j \in \mathbb{Z}$ and $m \in \mathbb{Z}^n$. Then

$$\begin{split} \|fa_{jm}\|_{E} &\leq \|f\chi_{3Q_{jm}}\|_{E} \\ &= |Q_{jm}|^{\frac{1}{p}} \frac{1}{\|\chi_{3Q_{jm}}\|_{E}} \|f\chi_{3Q_{jm}}\|_{E} \|\chi_{3Q_{j0}}\|_{E} |Q_{jm}|^{-\frac{1}{p}} \\ &\lesssim \|f\|_{\mathcal{M}_{E}^{p}} \|\chi_{Q_{j0}}\|_{E} |Q_{j0}|^{-\frac{1}{p}}. \end{split}$$

Consequently,

$$\|f \cdot g\|_E \leq \sum_{j=-\infty}^{\infty} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \|fa_{jm}\|_E \right)$$
$$\lesssim \|f\|_{\mathcal{M}_E^p} \sum_{j=-\infty}^{\infty} \|\chi_{Q_{j0}}\|_E |Q_{j0}|^{-\frac{1}{p}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}| \right).$$

If we take the infimum over all possible expressions of g, we obtain

$$||f \cdot g||_E \lesssim ||f||_{\mathcal{M}^p_E} ||g||_{\dot{B}^w_{11}}.$$

Thus, $f \in \text{PWM}(\dot{B}^w_{11}(\mathbb{R}^n), E(\mathbb{R}^n))$ and $\|f\|_{\text{PWM}(\dot{B}^w_{11}, E)} \lesssim \|f\|_{\mathcal{M}^p_E}$.

Conversely, we let $f \in \text{PWM}(\dot{B}_{11}^w(\mathbb{R}^n), E(\mathbb{R}^n))$. Choose a smooth function $\kappa \in C_c^\infty(\mathbb{R}^n)$ so that $\kappa(x) = 0$ for any $x \in \mathbb{R}^n \setminus 3Q_{00}$, that $\kappa(x) = 1$ for any $x \in Q_{00}$ and that

$$\int_{\mathbb{R}^n} x^\beta \kappa(x) \mathrm{d}x = 0$$

for any β with $|\beta| \leq [\alpha_1 + 1]$. Define $\kappa_{jm}(x) \equiv \kappa \left(2^j x - m\right)$ for each $j \in \mathbb{Z}$ and $m \in \mathbb{Z}^n$. Then since $2^{-jn} w_j \|f\|_{\text{PWM}(\dot{B}^w_{11}, E)} = \|\chi_{Q_{j0}}\|_E |Q_{j0}|^{-\frac{1}{p}} \|f\|_{\text{PWM}(\dot{B}^w_{11}, E)}$, we have

$$\|f\chi_{Q_{jm}}\|_{E} \leq \|f\kappa_{jm}\|_{E} \leq \|f\|_{\text{PWM}(\dot{B}_{11}^{w}, E)} \|\kappa_{jm}\|_{\dot{B}_{11}^{w}} \lesssim \|\chi_{Q_{j0}}\|_{E} |Q_{j0}|^{-\frac{1}{p}} \|f\|_{\text{PWM}(\dot{B}_{11}^{w}, E)}.$$
(4.2)

Thus, $f \in \mathcal{M}^p_E(\mathbb{R}^n)$.

4.2.2 Pointwise multipliers from Besov spaces to Lebesgue spaces and Morrey spaces (Theorems 4.1.5 and 4.1.6)

Let us prove Theorem 4.1.5.

Proof of Theorem 4.1.5. Let $f \in \text{PWM}(\dot{B}_{p1}^s(\mathbb{R}^n), L^p(\mathbb{R}^n))$. Then define κ_{jm} as before for all $m \in \mathbb{Z}^n$ and $j \in \mathbb{Z}$. Thus, by Example 2.2.7 and the same argument of (4.2),

$$||f||_{L^p(Q_{jm})} \lesssim 2^{js-j\frac{n}{p}} ||f||_{\mathrm{PWM}(\dot{B}^s_{p1},L^p)}$$

Consequently, $f \in \mathcal{M}_p^{\frac{n}{s}}(\mathbb{R}^n)$.

To show the opposite estimate, by Theorem 2.2.6, it suffices to show that

$$\left\| f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right\|_{L^p} \lesssim 2^{js} \|f\|_{\mathcal{M}_p^{\frac{n}{s}}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^p}$$

for all sequences $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of C^{∞} -functions satisfying $|\partial^{\alpha}a_{jm}| \leq 2^{j|\alpha|}\chi_{3Q_{jm}}$ with $|\alpha| \leq [s+1]$; once this is achieved, we have only to add this estimate over $j \in \mathbb{Z}$.

We calculate

$$\left\| f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right\|_{L^p} \lesssim \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^p \| f \chi_{3Q_{jm}} \|_{L^p}^p \right)^{\frac{1}{p}}$$
$$\leq \sup_{m \in \mathbb{Z}^n} \| f \chi_{3Q_{jm}} \|_{L^p} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^p \right)^{\frac{1}{p}}$$
$$\lesssim 2^{js} \| f \|_{\mathcal{M}_p^{\frac{n}{s}}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^p}.$$

Thus, the proof of Theorem 4.1.5 is complete.

We move on to the proof of Theorem 4.1.6.

Proof of Theorem 4.1.6. If we write out the norm of $\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q}}(\mathbb{R}^{n})$ using the dyadic Morrey norm, then we obtain

$$\|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q}}} \equiv \sup_{Q\in\mathcal{D}}\sup_{R\in\mathcal{D}}|Q|^{\frac{1}{\sigma}-\frac{1}{p}}|R|^{\frac{1}{p}-\frac{1}{q}}\|f\chi_{Q\cap R}\|_{L^{q}}.$$

Let Q, R be dyadic cubes. In order that $|Q|^{\frac{1}{\sigma}-\frac{1}{p}}|R|^{\frac{1}{p}-\frac{1}{q}}||f\chi_{Q\cap R}||_{L^q}$ is not zero, Q and R must intersect. Since

$$|Q|^{\frac{1}{\sigma}-\frac{1}{p}}|R|^{\frac{1}{p}-\frac{1}{q}} \le |Q \cap R|^{\frac{1}{\sigma}-\frac{1}{p}}|Q \cap R|^{\frac{1}{p}-\frac{1}{q}},$$

it follows that

$$\|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q}}} = \sup_{Q \in \mathcal{D}} |Q|^{\frac{1}{\sigma} - \frac{1}{p}} |Q|^{\frac{1}{p} - \frac{1}{q}} \|f\chi_{Q}\|_{L^{q}} = \sup_{Q \in \mathcal{D}} |Q|^{\frac{1}{\sigma} - \frac{1}{q}} \|f\chi_{Q}\|_{L^{q}} = \|f\|_{\mathcal{M}^{\sigma}_{q}}.$$

Let $f \in \text{PWM}\left(\dot{B}_{q1}^{s}(\mathbb{R}^{n}), \mathcal{M}_{q}^{p}(\mathbb{R}^{n})\right)$. As in the proof of Theorem 4.1.5, we obtain

$$\|f\chi_{Q_{jm}}\|_{\mathcal{M}^p_q} \lesssim 2^{js-j\frac{u}{q}} \|f\|_{\mathrm{PWM}(\dot{B}^s_{q1},\mathcal{M}^p_q)}.$$

Thus, $f \in \mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q}}(\mathbb{R}^{n}).$

To prove the opposite inclusion, it sufficies to show that

$$\|Q\|^{\frac{1}{p}-\frac{1}{q}} \left\| \left(f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right) \chi_Q \right\|_{L^q} \lesssim 2^{js} \|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^p_q}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^q}$$

for all cubes Q and sequences $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of C^{∞} -functions satisfying $|\partial^{\alpha}a_{jm}| \leq 2^{j|\alpha|}\chi_{3Q_{jm}}$ with $|\alpha| \leq [s+1]$; once this is achieved, we have only to take the supremum over all cubes and to add this estimate over $j \in \mathbb{Z}$.

We calculate

$$\begin{aligned} \|Q\|^{\frac{1}{p}-\frac{1}{q}} \left\| \left(f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right) \chi_Q \right\|_{L^q} &\lesssim \|Q\|^{\frac{1}{p}-\frac{1}{q}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^q \left\| (f\chi_{3Q_{jm}}) \chi_Q \right\|_{L^q} q \right)^{\frac{1}{q}} \\ &\leq \sup_{m \in \mathbb{Z}^n} \|Q\|^{\frac{1}{p}-\frac{1}{q}} \left\| (f\chi_{3Q_{jm}}) \chi_Q \right\|_{L^q} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^q \right)^{\frac{1}{q}} \\ &\leq \sup_{m \in \mathbb{Z}^n} \|f\chi_{3Q_{jm}}\|_{\mathcal{M}^p_q} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^q \right)^{\frac{1}{q}} \\ &\lesssim 2^{js} \|f\|_{\mathcal{M}^\sigma_{\mathcal{M}^p_q}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm}\chi_{Q_{jm}} \right\|_{L^q}. \end{aligned}$$

Thus, we obtain the desired result.

4.3 Examples

We will consider various special cases. We suppose that $E = L^{\Phi}(\mathbb{R}^n)$, the Orlicz space in Section 4.3.1, $E = L^{p,q}(\mathbb{R}^n)$, the Lorentz space in Section 4.3.2, $E = L^{\vec{q}}(\mathbb{R}^n)$, the mixed Lebesgue space in Section 4.3.3 and $E = \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, the mixed Morrey space in Section 4.3.4.

4.3.1 Pointwise multipliers from Besov spaces to Orlicz spaces

As an example of $E(\mathbb{R}^n)$, we first take up the case of $E(\mathbb{R}^n) = L^{\Phi}(\mathbb{R}^n)$. Then, it seems that the spaces $\mathcal{M}^p_E(\mathbb{R}^n)$ coincide Orlicz–Morrey spaces. However, there are at least three kinds of generalized Orlicz–Morrey spaces. In this time, we consider Orlicz–Morrey spaces of the third kind defined in [50].

Definition 4.3.1. Let $1 \leq p < \infty$. Suppose that Φ is a Young function. Also let Φ be bijective. Then, the Orlicz–Morrey space $\mathcal{M}_{p,\Phi}(\mathbb{R}^n)$ of the third kind is defined as the set of all measurable functions f for which the norm

$$\|f\|_{\mathcal{M}_{p,\Phi}} \equiv \sup_{Q \in \mathcal{Q}} |Q|^{\frac{1}{p}} \Phi^{-1}\left(\frac{1}{|Q|}\right) \|f\chi_Q\|_{L^{\Phi}}$$

is finite.

Generalized Orlicz–Morrey spaces of the first kind date back to 2004; see the paper [100] by Nakai, while the one of the second kind date back to 2012; see the paper [135] by Sawano, Sugano and Tanaka. See [26] for the definition of generalized Orlicz–Morrey spaces. In [101], Nakai showed the boundedness of the Hardy–Littlewood maximal operator for the Orlicz–Morrey spaces. As was shown in [39, Theorems 1.4 and 1.6], generalized Orlicz–Morrey spaces of the first kind and the one of the second kind are different. Deringoz, Guliyev, Hasanov, Noi, Samko and Sawano investigated the decomposition of Orlicz–Morrey spaces of the third kind [50]. See [27, 47] for vanishing generalized Orlicz–Morrey spaces of the third kind. We refer to [25] for the weighted Orlicz–Morrey spaces of the third kind. See [25] for the maximal operator and its commutator generated by BMO for the weighted Orlicz–Morrey spaces of the third kind. Finally, see [46] for the fractional integral operators together with commutators for the Orlicz–Morrey spaces of the third kind.

Thanks to Lemma 2.2.18, we can check the coincidence of $\mathcal{M}_{L^{\Phi}}^{p}(\mathbb{R}^{n})$ and $\mathcal{M}_{p,\Phi}(\mathbb{R}^{n})$. Indeed,

$$\|f\|_{\mathcal{M}^{p}_{L^{\Phi}}} = \sup_{Q \in \mathcal{Q}} |Q|^{\frac{1}{p}} \frac{1}{\|\chi_{Q}\|_{L^{\Phi}}} \|f\chi_{Q}\|_{L^{\Phi}} = \sup_{Q \in \mathcal{Q}} |Q|^{\frac{1}{p}} \Phi^{-1}\left(\frac{1}{|Q|}\right) \|f\chi_{Q}\|_{L^{\Phi}} = \|f\|_{\mathcal{M}_{p,\Phi}}.$$

From these observations, we can apply Theorem 4.1.4 to Orlicz spaces and obtain the following result.

Theorem 4.3.2. Let $1 \le p < \infty$ and Φ be as in Definition 4.3.1. Set

$$w_j = \Phi^{-1}\left(\frac{1}{|Q_{j0}|}\right) |Q_{j0}|^{-1-\frac{1}{p}} \quad (j \in \mathbb{Z}).$$

Then

$$\mathrm{PWM}(\dot{B}_{11}^w(\mathbb{R}^n), L^{\Phi}(\mathbb{R}^n)) \approx \mathcal{M}_{p,\Phi}(\mathbb{R}^n)$$

with equivalence of norms.

4.3.2 Pointwise multipliers from Besov spaces to Lorentz spaces

As an example of $E(\mathbb{R}^n)$, we take up the case of $E(\mathbb{R}^n) = L^{p,q}(\mathbb{R}^n)$. Then, we see that the space $\mathcal{M}^p_E(\mathbb{R}^n)$ coincides the Morrey–Lorentz space $\mathcal{M}^p_{q,r}(\mathbb{R}^n)$. We define

$$||f||_{\mathcal{M}^{p}_{q,r}} \equiv \sup_{Q} |Q|^{\frac{1}{p} - \frac{1}{q}} ||f\chi_{Q}||_{L^{q,r}}$$

for $0 < q \leq p < \infty$, $0 < r \leq \infty$. The Morrey–Lorentz space $\mathcal{M}_{q,r}^p(\mathbb{R}^n)$ denotes the set of all $f \in L^0(\mathbb{R}^n)$ for which $||f||_{\mathcal{M}_{q,r}^p}$ is finite. This space is introduced by Ragusa in [118]. Later, this space of characterizations and applications are investigated in [5, 36]. If $1 < q \leq p < \infty$ and $1 \leq r \leq \infty$, then $\mathcal{M}_{q,r}^p(\mathbb{R}^n)$ is a Banach space since above argument. The weak Morrey space $W\mathcal{M}_q^p(\mathbb{R}^n)$ denotes $\mathcal{M}_{q,\infty}^p(\mathbb{R}^n)$. We note that $||f||_{W\mathcal{M}_q^p} = \sup_{\lambda>0} \lambda ||\chi_{\{|f|>\lambda\}}||_{\mathcal{M}_q^p}$ for any $f \in L^0(\mathbb{R}^n)$.

When we replace E and F by $L^{q,r}(\mathbb{R}^n)$ and $L^{u,v}(\mathbb{R}^n)$ respectively in Theorem 4.1.4, we obtain the following result by using the embeddings of Lorentz spaces (Theorem 2.2.13).

Theorem 4.3.3. Let $1 < q \le p < \infty$, $1 \le r, v \le \infty$, $1 < u \le \infty$. Assume that

$$\frac{1}{q} = \frac{1}{p} + \frac{1}{u}.$$

Then $\dot{B}_{11}^{\frac{n}{p}+\frac{n}{q'}}(\mathbb{R}^n)$ is continuously embedded into $L^{u,v}(\mathbb{R}^n)$ and

$$\mathrm{PWM}(\dot{B}_{11}^{\frac{n}{p}+\frac{n}{q'}}(\mathbb{R}^n), L^{q,r}(\mathbb{R}^n)) \approx \mathcal{M}_{q,r}^p(\mathbb{R}^n)$$

with equivalence of norms. In particular,

$$\mathrm{PWM}(\dot{B}_{11}^{\frac{n}{p}+\frac{n}{q'}}(\mathbb{R}^n), \mathrm{W}L^q(\mathbb{R}^n)) \approx \mathrm{W}\mathcal{M}_q^p(\mathbb{R}^n)$$

with equivalence of norms.

We now extend Theorem 4.1.5 to Morrey-Lorentz spaces as follows.

Theorem 4.3.4. Let $1 , <math>1 \le q \le \infty$ and $0 < s \le \frac{n}{\min(p,q)}$. Assume

$$0 < \frac{s}{n} - \frac{1}{\min(p,q)} + \frac{1}{p} \le \frac{1}{p}.$$

Define $\sigma > 0$ by

$$\frac{1}{\sigma} \equiv \frac{s}{n} - \frac{1}{\min(p,q)} + \frac{1}{p}.$$

Then

$$\mathrm{PWM}(\dot{B}^s_{\min(p,q)1}(\mathbb{R}^n), L^{p,q}(\mathbb{R}^n)) \approx \mathcal{M}^{\sigma}_{p,q}(\mathbb{R}^n)$$

with equivalence of norms.

Theorem 4.3.5. Let $1 < q \le p < \infty$, $1 \le r \le \infty$ and $0 < s \le \frac{n}{\min(q, r)}$. Assume

$$0 < \frac{s}{n} - \frac{1}{\min(q, r)} + \frac{1}{p} \le \frac{1}{p}.$$

Define σ by

$$\frac{1}{\sigma} \equiv \frac{s}{n} - \frac{1}{\min(q, r)} + \frac{1}{p}.$$

Then

$$\mathrm{PWM}(\dot{B}^s_{\min(q,r)1}(\mathbb{R}^n), \mathcal{M}^p_{q,r}(\mathbb{R}^n)) \approx \mathcal{M}^{\sigma}_{\mathcal{M}^p_{q,r}}(\mathbb{R}^n) = \mathcal{M}^{\sigma}_{q,r}(\mathbb{R}^n)$$

with equivalence of norms.

Theorem 4.3.4 is included in Theorem 4.3.5. So we prove Theorem 4.3.5.

Proof of Theorem 4.3.5. As in the proof of Theorem 4.1.6, we obtain

$$\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q,r}}(\mathbb{R}^{n}) = \mathcal{M}^{\sigma}_{q,r}(\mathbb{R}^{n}).$$

Let $f \in \text{PWM}(\dot{B}^s_{\min(q,r)1}(\mathbb{R}^n), \mathcal{M}^p_{q,r}(\mathbb{R}^n))$. Then define κ_{jm} as before. Similar to the proof of Theorem 4.1.5, then we have

$$\|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q,r}}} \lesssim \|f\|_{\mathrm{PWM}(\dot{B}^{s}_{\min(q,r)1},\mathcal{M}^{p}_{q,r})}.$$

Consequently, $f \in \mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q,r}}(\mathbb{R}^{n}).$

To show the opposite estimate, it suffices to show that

$$|Q|^{\frac{1}{p}-\frac{1}{q}} \left\| \left(f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right) \chi_Q \right\|_{L^{q,r}} \lesssim 2^{js} \|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{q,r}}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^{\min(q,r)}}$$

for all sequences $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of C^{∞} -functions satisfying $|\partial^{\alpha}a_{jm}| \leq 2^{j|\alpha|}\chi_{3Q_{jm}}$ with $|\alpha| \leq [s+1]$; once this is achieved, we have only to add this estimate over $j \in \mathbb{Z}$.

Let r < q. By using Minkowski's inequality,

$$\begin{aligned} \|Q\|^{\frac{1}{p}-\frac{1}{q}} \left\| \left(f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right) \chi_Q \right\|_{L^{q,r}} &\lesssim \|Q\|^{\frac{1}{p}-\frac{1}{q}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^r \| (f\chi_{3Q_{jm}}) \chi_Q \|_{L^{q,r}} r \right)^{\frac{1}{r}} \\ &\leq \sup_{m \in \mathbb{Z}^n} \| f\chi_{3Q_{jm}} \|_{\mathcal{M}^p_{q,r}} \left(\sum_{m \in \mathbb{Z}^n} |\lambda_{jm}|^r \right)^{\frac{1}{r}} \\ &\lesssim 2^{js} \| f \|_{\mathcal{M}^\sigma_{\mathcal{M}^p_{q,r}}} \left\| \sum_{m \in \mathbb{Z}^n} \lambda_{jm} \chi_{Q_{jm}} \right\|_{L^r}. \end{aligned}$$

Conversely, let $q \leq r$. Note that, each cube Q_{jl} intersects 3^n cubes $3Q_{jm_{l,k}}, k = 1, 2, \ldots, 3^n$. Therefore, using Lemma 2.2.14, we have

$$\begin{aligned} &|Q|^{\frac{q}{p}-1} \left\| \left(f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right) \chi_Q \right\|_{L^{q,r}}^q \\ &\lesssim |Q|^{\frac{q}{p}-1} \sum_{l \in \mathbb{Z}^n} \left\| \left(f \sum_{k=1}^{3^n} \lambda_{jm_{l,k}} a_{jm_{l,k}} \right) \chi_Q \right\|_{L^{q,r}(Q_{jl})}^q \\ &\lesssim |Q|^{\frac{q}{p}-1} \sum_{l \in \mathbb{Z}^n} \sum_{k=1}^{3^n} |\lambda_{jm_{l,k}}|^q \left\| \left(|f| \sum_{k=1}^{3^n} |\chi_{3Q_{jm_{l,k}}}| \right) \chi_Q \right\|_{L^{q,r}(Q_{jl})}^q \\ &\lesssim \sup_{l \in \mathbb{Z}^n} \| f \chi_{Q_{jl}} \|_{\mathcal{M}^{q}_{q,r}}^q \left(\sum_{l \in \mathbb{Z}^n} |\lambda_{jl}|^q \right) \end{aligned}$$

Hence

$$\left\| f \sum_{m \in \mathbb{Z}^n} \lambda_{jm} a_{jm} \right\|_{\mathcal{M}^p_{q,r}} \lesssim 2^{js} \| f \|_{\mathcal{M}^\sigma_{\mathcal{M}^p_{q,r}}} \left\| \sum_{l \in \mathbb{Z}^n} \lambda_{jl} \chi_{Q_{jl}} \right\|_{L^q}$$

Thus, f is a pointwise multiplier from $\dot{B}^s_{\min(q,r)1}(\mathbb{R}^n)$ to $\mathcal{M}^{\sigma}_{q,r}(\mathbb{R}^n)$.

4.3.3 Pointwise multipliers from Besov spaces to mixed Lebesgue spaces

Applying Theorem 4.1.4 for mixed Lebesgue spaces, we obtain the following result:

Theorem 4.3.6. Let $1 \leq q_1, \ldots, q_n < \infty$, and suppose that p satisfies

$$\frac{1}{p} \le \frac{1}{n} \sum_{k=1}^{n} \frac{1}{q_k}.$$

Define r by

$$\frac{1}{r} \equiv \frac{1}{n} \sum_{k=1}^{n} \frac{1}{q_k} - \frac{1}{p}.$$

Then $\dot{B}_{11}^{\frac{n}{r'}}(\mathbb{R}^n)$ is continuously embedded into $L^r(\mathbb{R}^n)$ and

$$\operatorname{PWM}\left(\dot{B}_{11}^{\sum\limits_{k=1}^{n}\frac{1}{r_{k}}}(\mathbb{R}^{n}), L^{\vec{q}}(\mathbb{R}^{n})\right) = \mathcal{M}_{L^{\vec{q}}}^{p}(\mathbb{R}^{n}) = \mathcal{M}_{\vec{q}}^{p}(\mathbb{R}^{n}).$$

Proof. Simply observe that $E(\mathbb{R}^n) = L^{\vec{q}}(\mathbb{R}^n)$ and $F(\mathbb{R}^n) = L^r(\mathbb{R}^n)$ satisfy the condition of Theorem 4.1.4.

4.3.4Pointwise multipliers from Besov spaces to mixed Morrey spaces

As an example of $E(\mathbb{R}^n)$, we take up the case of $E(\mathbb{R}^n) = \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

Remark 4.3.7. In the definition of \mathcal{M}^p_E , let $E(\mathbb{R}^n) = \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then

$$\|f\|_{\mathcal{M}^{p}_{\mathcal{M}^{p}_{\vec{q}}}} = \sup_{Q} |Q|^{\frac{1}{p}} \left(\frac{1}{\|\chi_{Q}\|_{\mathcal{M}^{p}_{\vec{q}}}} \|f\chi_{Q}\|_{\mathcal{M}^{p}_{\vec{q}}} \right) = \sup_{Q} |Q|^{\frac{1}{p}} \left(\frac{1}{|Q|^{\frac{1}{p}}} \|f\chi_{Q}\|_{\mathcal{M}^{p}_{\vec{q}}} \right) = \|f\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

Thus, we see that

$$\mathcal{M}^p_{\mathcal{M}^p_{\vec{q}}}(\mathbb{R}^n) = \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n).$$

Keeping this remark in mind, we apply Theorem 4.1.4 for this space.

Theorem 4.3.8. Let $1 \le p < \infty$ and $1 \le \vec{q} < \infty$ satisfy

$$\sum_{j=1}^{n} \frac{1}{q_j} \ge \frac{n}{p}.$$

Then

$$\operatorname{PWM}\left(\dot{B}_{11}^{n}(\mathbb{R}^{n}), \mathcal{M}_{\vec{q}}^{p}(\mathbb{R}^{n})\right) = \mathcal{M}_{\mathcal{M}_{\vec{q}}^{p}}^{p}(\mathbb{R}^{n}) = \mathcal{M}_{\vec{q}}^{p}(\mathbb{R}^{n}).$$

Also we extend Theorem 4.1.5 to mixed Morrey spaces.

Theorem 4.3.9. Let $\vec{q} = (q_1, ..., q_n) \in [1, \infty)^n, 1 \le p < \infty$ and $0 < s \le \frac{n}{q_n}$. Suppose that σ is given by

$$\frac{1}{\sigma} \equiv \frac{s}{n} - \frac{1}{q_n} + \frac{1}{p}$$

Then with equivalence norms

$$\operatorname{PWM}\left(\dot{B}_{q_n1}^s(\mathbb{R}^n), \mathcal{M}_{\vec{q}}^p(\mathbb{R}^n)\right) \approx \mathcal{M}_{\mathcal{M}_{\vec{q}}^p}^{\sigma}(\mathbb{R}^n) = \mathcal{M}_{\vec{q}}^{\sigma}(\mathbb{R}^n).$$

Proof. Let $f \in \text{PWM}\left(\dot{B}^{s}_{q_{n}1}(\mathbb{R}^{n}), \mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})\right)$. As in the proof of Theorem 4.1.5, we obtain $\|f\chi_{Q_{jm}}\|_{\mathcal{M}^p_{\vec{q}}} \lesssim 2^{js-j\frac{n}{q_n}} \|f\|_{\mathrm{PWM}(\dot{B}^s_{q_n1},\mathcal{M}^p_{\vec{q}})}.$

Thus, $f \in \mathcal{M}^{\sigma}_{\mathcal{M}^{p}_{\vec{q}}}(\mathbb{R}^{n}).$

To establish the opposite inclusion, it suffices to show that

$$|Q|^{\frac{1}{p}-\frac{1}{n}\sum_{k=1}^{n}\frac{1}{q_{k}}}\left\|\left(f\sum_{m\in\mathbb{Z}^{n}}\lambda_{jm}a_{jm}\right)\chi_{Q}\right\|_{L^{\vec{q}}}\lesssim 2^{js}\|f\|_{\mathcal{M}^{\sigma}_{\mathcal{M}^{\vec{p}}_{\vec{q}}}}\left\|\sum_{m\in\mathbb{Z}^{n}}\lambda_{jm}\chi_{Q_{jm}}\right\|_{L^{q_{n}}}$$

for all cubes Q and sequences $\{a_{jm}\}_{j\in\mathbb{Z},m\in\mathbb{Z}^n}$ of C^{∞} -functions satisfying $|\partial^{\alpha}a_{jm}| \leq 2^{j|\alpha|}\chi_{3Q_{jm}}$ with $|\alpha| \leq [s+1]$; once this is achieved, again we have only to take the supremum over all cubes and to add this estimate over $j \in \mathbb{Z}$.

We calculate

$$\begin{aligned} |Q|^{\frac{1}{p}-\frac{1}{n}}\sum_{k=1}^{n}\frac{1}{q_{k}} \left\| \left(f\sum_{m\in\mathbb{Z}^{n}}\lambda_{jm}a_{jm} \right)\chi_{Q} \right\|_{L^{\vec{q}}} \\ &\lesssim |Q|^{\frac{1}{p}-\frac{1}{n}}\sum_{k=1}^{n}\frac{1}{q_{k}} \left(\sum_{m\in\mathbb{Z}^{n}}|\lambda_{jm}|^{q_{n}} \left\| \left(f\chi_{3}Q_{jm} \right)\chi_{Q} \right\|_{L^{q}} q_{n} \right)^{\frac{1}{q_{n}}} \\ &\leq \sup_{m\in\mathbb{Z}^{n}} |Q|^{\frac{1}{p}-\frac{1}{n}}\sum_{k=1}^{n}\frac{1}{q_{k}}} \left\| \left(f\chi_{3}Q_{jm} \right)\chi_{Q} \right\|_{L^{\vec{q}}} \left(\sum_{m\in\mathbb{Z}^{n}}|\lambda_{jm}|^{q_{n}} \right)^{\frac{1}{q_{n}}} \\ &\leq \sup_{m\in\mathbb{Z}^{n}} \| f\chi_{3}Q_{jm} \|_{\mathcal{M}_{\vec{q}}^{p}} \left(\sum_{m\in\mathbb{Z}^{n}}|\lambda_{jm}|^{q_{n}} \right)^{\frac{1}{q_{n}}} \\ &\lesssim 2^{js} \| f\|_{\mathcal{M}_{\mathcal{M}_{\vec{q}}^{p}}^{\sigma}} \left\| \sum_{m\in\mathbb{Z}^{n}}\lambda_{jm}\chi_{Q_{jm}} \right\|_{L^{q_{n}}}. \end{aligned}$$

Thus, f is a pointwise multiplier from $\dot{B}^s_{q_n 1}(\mathbb{R}^n)$ to $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

Chapter 5

Atomic decomposition for mixed Morrey spaces

5.1 Introduction and theorems

One of the characterizations of function spaces is to decompose an element of function spaces into good functions. The most important example is the Fourier series for L^2 functions. In addition, there are the Littlewood–Paley decomposition (cf. Subsection 2.2.1), the molecular decomposition, and the wavelet decomposition. In this thesis, we focus our interest on atomic decompositions, which were introduced by Coifman in 1974 to characterize the functions belonging to Hardy spaces. Here, an atom is a function which has a support on a cube (support condition), a norm estimate with respect to the cube (size condition), and moment condition. This decomposition is applied to characterize many function spaces.

The aim of this chapter is to develop a theory of decompositions for mixed Morrey spaces. Furthermore, we can extend these results to Hardy-mixed Morrey spaces. Our results are extension of the results for classical Morrey spaces and Hardy-Morrey spaces in [66].

First, we will prove the following boundedness of the maximal operator on mixed Lebesgue spaces.

Theorem 5.1.1. Assume that

 $1 \le t_k < \min\{q_1, \dots, q_k\} \le \infty \quad (k = 1, \dots, n).$

Define

$$M^{(\vec{t})}f(x) = \sup_{Q \in \mathcal{Q}} \frac{\chi_Q(x)}{\|\chi_Q\|_{L^{\vec{t}}}} \|f\chi_Q\|_{L^{\vec{t}}}$$

for a measurable function f. Then for all measurable functions f

$$\|M^{(t)}f\|_{L^{\vec{q}}} \lesssim \|f\|_{L^{\vec{q}}}$$

Based on this boundedness, we prove the following construction theorem. This is the same one as Theorem 1.2.6 in Chapter 1.

Theorem 5.1.2. Suppose that the parameters p, \vec{q}, s, \vec{t} satisfy

$$1
$$\frac{n}{p} \le \sum_{j=1}^n \frac{1}{q_j}, \quad \frac{n}{s} \le \sum_{j=1}^n \frac{1}{t_j}.$$$$

Assume that $\{a_j\}_{j=1}^{\infty} \subset \mathcal{M}_{\vec{t}}^s(\mathbb{R}^n), \{\lambda_j\}_{j=1}^{\infty} \subset [0,\infty), \text{ and } \{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n) \text{ fulfill}$

$$||a_j||_{\mathcal{M}^s_{\vec{t}}} \leq |Q_j|^{\frac{1}{s}}, \quad \operatorname{supp}(a_j) \subset Q_j, \quad \left\|\sum_{j=1}^\infty \lambda_j \chi_{Q_j}\right\|_{\mathcal{M}^p_{\vec{q}}} < \infty.$$

Then
$$f = \sum_{j=1}^{\infty} \lambda_j a_j$$
 converges in $\mathcal{S}'(\mathbb{R}^n) \cap L^{\vec{q}}_{\text{loc}}(\mathbb{R}^n)$ and satisfies

$$\|f\|_{\mathcal{M}^p_{\vec{q}}} \le C_{p,\vec{q},s,\vec{t}} \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

In connection with Theorem 5.1.2, we refer to [2, 40, 107] for more recent characterizations of the predual spaces of classical and mixed Morrey spaces.

Before we state the next results, we recall the some notation. For $d \ge 0$, denote by $\mathcal{P}_d(\mathbb{R}^n)$ the set of all polynomial functions with degree less than or equal to d, so that $\mathcal{P}(\mathbb{R}^n) \equiv \bigcup_{d=0}^{\infty} \mathcal{P}_d(\mathbb{R}^n)$. It is clear that $\mathcal{P}_{-1}(\mathbb{R}^n) = \{0\}$. The set $\mathcal{P}_K(\mathbb{R}^n)^{\perp}$ denotes the set of measurable function f for which $\langle \cdot \rangle^K f \in L^1(\mathbb{R}^n)$ and $\int_{\mathbb{R}^n} x^{\alpha} f(x) dx = 0$ for any $\alpha \in \mathbb{N}_0^n$ with $|\alpha| \le K$, where $\langle \cdot \rangle = (1 + |\cdot|^2)^{\frac{1}{2}}$. Such a function f is said to satisfy the moment condition of order K. In this case, one also writes $f \perp \mathcal{P}_K(\mathbb{R}^n)$.

The next assertion concerns the decomposition of functions in $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. This theorem is the same one as Theorem 1.2.7 in Chapter 1.

Theorem 5.1.3 (cf. [60]). Suppose that the real parameters p, \vec{q}, K satisfy

$$1$$

where $q_0 = \min(q_1, \ldots, q_n)$. Let $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then there exists a triplet $\{a_j\}_{j=1}^{\infty} \subset L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}^{\perp}_K(\mathbb{R}^n)$, $\{\lambda_j\}_{j=1}^{\infty} \subset [0, \infty)$, and $\{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n)$ such that $f = \sum_{j=1}^{\infty} \lambda_j a_j$ in

 $\mathcal{S}'(\mathbb{R}^n)$ and that, for any v > 0

$$|a_j| \le \chi_{Q_j}, \quad \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^v \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}} \le C_v \|f\|_{\mathcal{M}^p_{\vec{q}}}.$$
(5.1)

Here the constant $C_v > 0$ is independent of f.

We rephrase Theorems 5.1.2 and 5.1.3 in the case of mixed Lebesgue spaces.

Corollary 5.1.4. Suppose that the parameters \vec{q}, \vec{t} satisfy

$$1 < \max\{q_1, \dots, q_k\} < t_k < \infty \quad (k = 1, \dots, n).$$

Assume that $\{a_j\}_{j=1}^{\infty} \subset L^{\vec{t}}(\mathbb{R}^n), \{\lambda_j\}_{j=1}^{\infty} \subset [0,\infty), \text{ and } \{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n) \text{ fulfill}$

$$||a_j||_{L^{\vec{t}}} \le |Q_j|^{\frac{1}{n}\sum_{k=1}^{n}\frac{1}{t_k}}, \quad \operatorname{supp}(a_j) \subset Q_j, \quad \left\|\sum_{j=1}^{\infty}\lambda_j\chi_{Q_j}\right\|_{L^{\vec{q}}} < \infty.$$

Then $f = \sum_{j=1}^{\infty} \lambda_j a_j$ converges in $L^{\vec{q}}(\mathbb{R}^n)$ and satisfies

$$\|f\|_{L^{\vec{q}}} \le C_{p,q,s,t} \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{L^{\vec{q}}}.$$

Corollary 5.1.5 (cf. [60]). Let $1 < \vec{q} < \infty$ and $K \in \mathbb{N}_0$. Let $f \in L^{\vec{q}}(\mathbb{R}^n)$. Then there exists a triplet $\{a_j\}_{j=1}^{\infty} \subset L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}_K^{\perp}(\mathbb{R}^n), \{\lambda_j\}_{j=1}^{\infty} \subset [0,\infty), \text{ and } \{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n)$ such that $f = \sum_{j=1}^{\infty} \lambda_j a_j$ in $L^{\vec{q}}(\mathbb{R}^n)$ and that, for any v > 0

$$|a_j| \le \chi_{Q_j}, \quad \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^v \right)^{\frac{1}{v}} \right\|_{L^{\vec{q}}} \le C_v \|f\|_{L^{\vec{q}}}.$$

Here the constant $C_v > 0$ is independent of f.

Next, we generalize Theorems 5.1.2 and 5.1.3. Based on [126], we define Hardymixed Morrey spaces.

Definition 5.1.6. For $0 < \vec{q}, p < \infty$ satisfying $\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$, the Hardy-mixed Morrey space $H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ is defined as the set of any $f \in \mathcal{S}'(\mathbb{R}^n)$ for which the quasi-norm

$$\|f\|_{H\mathcal{M}^p_{\vec{q}}} = \left\|\sup_{t>0} |e^{t\Delta}f|\right\|_{\mathcal{M}^p_{\vec{q}}}$$
is finite, where $e^{t\Delta}f$ stands for the heat extension of f;

$$e^{t\Delta}f(x) = \left\langle \frac{1}{\sqrt{(4\pi t)^n}} \exp\left(-\frac{|x-\cdot|^2}{4t}\right), f \right\rangle \quad (t > 0, \ x \in \mathbb{R}^n).$$

See [150] for the equivalent norms of Hardy–Morrey spaces. We rephrase Theorem 5.1.2 and 5.1.3 in full generality in terms of Hardy-mixed Morrey spaces.

Theorem 5.1.7. Suppose that the parameters p, \vec{q}, s, \vec{t} satisfy

$$1
$$\frac{n}{p} \le \sum_{j=1}^n \frac{1}{q_j}, \quad \frac{n}{s} \le \sum_{j=1}^n \frac{1}{t_j}.$$
$$\vec{y} \equiv \min\{1, q_1, \dots, q_n\} \text{ and } d_q = \left[n\left(\frac{1}{1+1}-1\right)\right]. \text{ Assume that a transformation}$$$$

Write $v(\vec{q}) \equiv \min\{1, q_1, \dots, q_n\}$ and $d_q = \left\lfloor n\left(\frac{1}{v(\vec{q})} - 1\right) \right\rfloor$. Assume that a triple

$$(\{a_j\}_{j=1}^{\infty}, \{\lambda_j\}_{j=1}^{\infty}, \{Q_j\}_{j=1}^{\infty}) \in (\mathcal{M}_{\vec{t}}^s(\mathbb{R}^n) \cap \mathcal{P}_{d_q}^{\perp}(\mathbb{R}^n)) \times [0, \infty) \times \mathcal{Q}(\mathbb{R}^n)$$

fulfills

$$\|a_j\|_{\mathcal{M}^s_{\vec{t}}} \leq |Q_j|^{\frac{1}{s}}, \quad \operatorname{supp}(a_j) \subset Q_j, \quad \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}} < \infty.$$

Then $f = \sum_{j=1}^{\infty} \lambda_j a_j$ converges in $\mathcal{S}'(\mathbb{R}^n)$ and satisfies

$$\|f\|_{H\mathcal{M}^p_{\vec{q}}} \le C_{p,\vec{q},s,\vec{t}} \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Theorem 5.1.3 has the following counterpart.

Theorem 5.1.8 (cf. [60]). Suppose that the real parameters p, \vec{q}, K satisfy

$$0$$

where $q_0 = \min(1, q_1, \ldots, q_n)$. Let $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then there exists a triplet $\{a_j\}_{j=1}^{\infty} \subset L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}_K^{\perp}(\mathbb{R}^n)$, $\{\lambda_j\}_{j=1}^{\infty} \subset [0, \infty)$, and $\{Q_j\}_{j=1}^{\infty} \subset \mathcal{Q}(\mathbb{R}^n)$ such that $f = \sum_{j=1}^{\infty} \lambda_j a_j$ in $\mathcal{S}'(\mathbb{R}^n)$ and that, for any v > 0,

$$|a_j| \le \chi_{Q_j}, \quad \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^v \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}} \le C_v \|f\|_{H\mathcal{M}^p_{\vec{q}}}.$$
(5.2)

Here the constant C_v is a constant that is independent on v but not on f.

As we said in Subsection 1.2.3, although Theorems 5.1.3 and 5.1.8 are given as corollaries of the results in [60], we prove these theorems directly without using Hertz spaces which were used in [60].

Finally, we survey the classical results of this section. In fact, atomic decompositions are roughly classified into two types. One is the decomposition by smooth atoms, and the other is the decomposition by non-smooth atoms. Our results correspond to the non-smooth case. Concerning non-smooth results for Morrey spaces, Jia and Wang considered non-smooth atomic decompositions for Hardy-Morrey spaces in [72]. After that in [66], Iida, Sawano, and Tanaka investigated non-smooth atomic decompositions for Morrey spaces, which include results of Jia and Wang. So, our results for mixed Morrey spaces are extension of these results. At last, with respect to smooth results for Morrey spaces, we refer to [92, 131].

The remaining parts of this chapter is as follows. In Section 5.2, we establish the boundedness result for the maximal operator to prove Theorem 5.1.2. In Section 5.3, we give characterizations of Hardy-Morrey spaces. Section 5.4 is devoted to showing the main theorems. In Subsections 5.4.1 and 5.4.2, we consider the reconstruction theorems for mixed Morrey spaces (Theorem 5.1.2) and Hardy-mixed Morrey spaces (Theorem 5.1.7), respectively. Meanwhile in Subsection 5.4.3, we prove the decomposition theorems for mixed Morrey spaces (Theorem 5.1.3) and Hardy-mixed Morrey spaces (Theorem 5.1.8). As an application, we concern Olsen's inequality for mixed Morrey spaces in Section 5.5.

5.2 The boundedness of the maximal operator $M^{(t)}$ (Theorem 5.1.1)

In this section, we prove the boundedness property of the maximal operator $M^{(\vec{t})}$ on mixed Morrey spaces. This theorem is applied to the proof of Theorem 5.1.2. We invoke a result due to Bagby [10].

Lemma 5.2.1. Let $1 < q_1, \ldots, q_m < \infty$ and $1 . For <math>i = 1, 2, \ldots, m$, let (Ω_i, μ_i) be σ -finite measure spaces, and $\Omega = \Omega_1 \times \cdots \times \Omega_m$. For $f \in L^0(\mathbb{R}^n \times \Omega)$,

$$\int_{\mathbb{R}^n} \|Mf(x,\cdot)\|_{L^{(q_1,\ldots,q_m)}}^p \,\mathrm{d}x \lesssim \int_{\mathbb{R}^n} \|f(x,\cdot)\|_{L^{(q_1,\ldots,q_m)}}^p \,\mathrm{d}x.$$

The following lemma is used in the induction step.

Lemma 5.2.2. Let $\vec{q} = (q_1, q_2, \dots, q_n) \in (1, \infty)^n$ and let

$$t_n \in [1, \min\{q_1, q_2, \dots, q_n\}).$$

Then

$$\left\| M_n^{(t_n)} f \right\|_{L^{\vec{q}}} \lesssim \|f\|_{L^{\vec{q}}}$$

for all $f \in L^{\vec{q}}(\mathbb{R}^n)$.

For the proof we use the following notation for $h \in L^0(\mathbb{R}^n)$:

$$\|h\|_{L^{(q_1,\ldots,q_m)}}(x_{m+1},\ldots,x_n) \equiv \|\left[\|h\|_{L^{(q_1,\ldots,q_{m-1})}}\right]\|_{L^{(q_m)}}(x_{m+1},\ldots,x_n)$$

and when m = 1, we define

$$||h||_{L^{(q_1)}}(x_2,\ldots,x_n) \equiv \left(\int_{\mathbb{R}} |h(x_1,\ldots,x_n)|^{q_1} \mathrm{d}x_1\right)^{\frac{1}{q_1}}.$$

Proof. Thanks to Lemma 5.2.1, we obtain

$$\begin{split} \left\| M_n^{(t_n)} f \right\|_{L^{\vec{q}}}^{q_n} &= \int_{\mathbb{R}} \left\| M_n^{(t_n)} f(\cdot, x_n) \right\|_{L^{(q_1, \dots, q_{n-1})}}^{q_n} \mathrm{d}x_n \\ &= \int_{\mathbb{R}} \left\| M_n[|f|^{t_n}](\cdot, x_n) \right\|_{L^{\left(\frac{q_1}{t_n}, \dots, \frac{q_{n-1}}{t_n}\right)}}^{q_n} \mathrm{d}x_n \\ &\lesssim \int_{\mathbb{R}} \left\| [|f(\cdot, x_n)|^{t_n}] \right\|_{L^{\left(\frac{q_1}{t_n}, \dots, \frac{q_{n-1}}{t_n}\right)}}^{q_n} \mathrm{d}x_n = \|f\|_{L^{\vec{q}}}^{q_n}. \end{split}$$

Thus, we obtain the desired result.

Proof of Theorem 5.1.1. We start with a preliminary observation for maximal operators. Let $x \in \mathbb{R}^n$. Let $Q = I_1 \times \cdots \times I_n$ where each I_j is an interval in \mathbb{R} with the same length. Then,

$$\begin{split} & \frac{\chi_Q(x)}{\|\chi_Q\|_{L^{\vec{t}}}} \|f\chi_Q\|_{L^{\vec{t}}} = \frac{\bigotimes_{j=1}^n \chi_{I_j}(x)}{\prod\limits_{j=1}^n |I_j|^{\frac{1}{t_j}}} \left\|f\chi_{\prod\limits_{j=1}^n I_j}\right\|_{L^{\vec{t}}} \\ &= \frac{\bigotimes_{j=2}^n \chi_{I_j}(x_2, \dots, x_n)}{\prod\limits_{j=2}^n |I_j|^{\frac{1}{t_j}}} \\ & \times \left\| \left[\left(\frac{\chi_{I_1}(x_1)}{|I_1|} \int |f(y)|^{t_1} \chi_{I_1}(y_1) \mathrm{d}y_1\right)^{\frac{1}{t_1}} \right] \chi_{\prod\limits_{j=2}^n I_j} \right\|_{L^{(t_2,\dots,t_n)}} \\ &\leq \frac{\bigotimes_{j=2}^n \chi_{I_j}(x_2, \dots, x_n)}{\prod\limits_{j=2}^n |I_j|^{\frac{1}{t_j}}} \left\| \left[M^{(t_1)} f \right] \chi_{\prod\limits_{j=2}^n I_j} \right\|_{L^{(t_2,\dots,t_n)}}. \end{split}$$

Continuing this procedure, we have

$$\frac{\chi_Q(x)}{\|\chi_Q\|_{L^{\vec{t}}}} \|f\chi_Q\|_{L^{\vec{t}}} \le M_n^{(t_n)} \cdots M_1^{(t_1)}(f)(x).$$

Thus, it follows that

$$M^{(\vec{t})}f(x) \le M_n^{(t_n)} \cdots M_1^{(t_1)}(f)(x).$$

Therefore, it suffices to show that

$$\left\| M_n^{(t_n)} \cdots M_1^{(t_1)}(f) \right\|_{L^{\vec{q}}} \lesssim \|f\|_{L^{\vec{q}}}.$$
(5.3)

We proceed by induction on n. For n = 1, the result follows by the classical case of the boundedness of the Hardy–Littlewood maximal operator.

Suppose that the result holds for n = m - 1 with m > 1 in \mathbb{N} : assume that

$$\|M_{m-1}^{(t_{m-1})}\cdots M_1^{(t_1)}h\|_{L^{(q_1,\dots,q_{m-1})}} \lesssim \|h\|_{L^{(q_1,\dots,q_{m-1})}}$$

for $1 < t_k < \min\{q_1, \ldots, q_k\} < \infty$ for each $k = 1, \ldots, m-1$, and for $h \in L^0(\mathbb{R}^{m-1})$. Since $t_m < \min\{q_1, \ldots, q_m\}$, for $g \in L^0(\mathbb{R}^m)$, we have

$$\begin{split} \left\| M_m^{(t_m)} g \right\|_{L^{(q_1,\dots,q_m)}} &= \left\| \left[\left\| M_m^{(t_m)} g \right\|_{L^{(q_1,\dots,q_{m-1})}} \right] \right\|_{L^{(q_m)}} \\ &= \left\| \left[\left\| M_m[|g|^{t_m}] \right\|_{L^{\left(\frac{q_1}{t_m},\dots,\frac{q_{m-1}}{t_m}\right)}} \right] \right\|_{L^{\left(\frac{q_m}{t_m}\right)}} \\ &\lesssim \left\| \left[\left\| g \right\|_{L^{(q_1,\dots,q_{m-1})}} \right] \right\|_{L^{(q_m)}} &= \left\| g \right\|_{L^{(q_1,\dots,q_m)}}. \end{split}$$

Thus, by the induction assumption, letting $g = M_{m-1}^{(t_{m-1})} \cdots M_1^{(t_1)}(f)$, we obtain

$$\begin{split} \left\| M_m^{(t_m)} \cdots M_1^{(t_1)}(f) \right\|_{L^{(q_1,\dots,q_m)}} &= \left\| M_m^{(t_m)} [M_{m-1}^{(t_{m-1})} \cdots M_1^{(t_1)}(f)] \right\|_{L^{(q_1,\dots,q_m)}} \\ &\lesssim \left\| M_{m-1}^{(t_{m-1})} \cdots M_1^{(t_1)}(f) \right\|_{L^{(q_1,\dots,q_m)}} \\ &= \left\| \left\| M_{m-1}^{(t_{m-1})} \cdots M_1^{(t_1)}(f) \right\|_{L^{(q_1,\dots,q_{m-1})}} \right\|_{L^{(q_m)}} \\ &\lesssim \left\| \|f\|_{L^{(q_1,\dots,q_{m-1})}} \right\|_{L^{(q_m)}} \lesssim \|f\|_{L^{(q_1,\dots,q_m)}}. \end{split}$$

Hence, inequality (5.3) holds for any dimension n. We obtain the desired result.

One can show that the condition

$$t_k < \min\{q_1, q_2, \dots, q_k\}$$

is sharp.

Proposition 5.2.3. In Theorem 5.1.1, for each k = 1, 2, ..., n, the condition $t_k < \min\{q_1, q_2, ..., q_k\}$ can not be removed.

Proof. We induct on n. The base case n = 1 is clear since the Hardy–Littlewood maximal operator is bounded on $L^p(\mathbb{R})$ if and only if p > 1. Assume that the conclusion of Proposition 5.2.3 is true for n = m - 1 and that $M^{(t_1, t_2, \dots, t_m)}$ is bounded on $L^{(q_1, q_2, \dots, q_m)}(\mathbb{R}^m)$. Let $h \in L^{(t_1, t_2, \dots, t_{m-1})}(\mathbb{R}^{m-1})$ and $N \in \mathbb{N}$. Then

$$\chi_{[-N,N]^m} \left(M^{(t_1,t_2,\dots,t_{m-1})} \left[\chi_{[-N,N]^{m-1}} h \right] \otimes \chi_{[-N,N]} \right) \\ \leq M^{(t_1,t_2,\dots,t_m)} \left[(\chi_{[-N,N]^{m-1}} h) \otimes \chi_{[-N,N]} \right].$$

Consequently,

$$\begin{aligned} (2N)^{\frac{1}{q_m}} \left\| \chi_{[-N,N]^{m-1}} M^{(t_1,t_2,\dots,t_{m-1})} \left[\chi_{[-N,N]^{m-1}} h \right] \right\|_{L^{(q_1,q_2,\dots,q_{m-1})}} \\ &= \left\| \chi_{[-N,N]^m} M^{(t_1,t_2,\dots,t_{m-1})} \left[\chi_{[-N,N]^{m-1}} h \right] \otimes \chi_{[-N,N]} \right\|_{L^{(q_1,q_2,\dots,q_m)}} \\ &\leq \left\| M^{(t_1,t_2,\dots,t_{m-1})} \left[\chi_{[-N,N]^{m-1}} h \right] \otimes \chi_{[-N,N]} \right\|_{L^{(q_1,q_2,\dots,q_m)}} \\ &\leq C \left\| \left(\chi_{[-N,N]^{m-1}} h \right) \otimes \chi_{[-N,N]} \right\|_{L^{(q_1,q_2,\dots,q_m)}} \\ &\leq C (2N)^{\frac{1}{q_m}} \left\| h \right\|_{L^{(q_1,q_2,\dots,q_{m-1})}}. \end{aligned}$$

So, we are led to

$$\left\|\chi_{[-N,N]^{m-1}}M^{(t_1,t_2,\dots,t_{m-1})}[\chi_{[-N,N]^{m-1}}h]\right\|_{L^{(q_1,q_2,\dots,q_{m-1})}} \le C\|h\|_{L^{(q_1,q_2,\dots,q_{m-1})}}.$$

Letting $N \to \infty$, we obtain

$$\left\| M^{(t_1,t_2,\dots,t_{m-1})}h \right\|_{L^{(q_1,q_2,\dots,q_{m-1})}} \le C \|h\|_{L^{(q_1,q_2,\dots,q_{m-1})}}$$

By the induction assumption, we have $t_k < \min\{q_1, q_2, \ldots, q_k\}$ for all $k = 1, 2, \ldots, m-1$. If we start from the inequality

$$\chi_{[-N,N]^m} \left(\chi_{[-N,N]} \otimes M^{(t_1,t_2,\dots,t_{m-1})} \left[\chi_{[-N,N]^{m-1}} h \right] \right) \\ \leq M^{(t_1,t_2,\dots,t_m)} \left[\chi_{[-N,N]} \otimes \left(\chi_{[-N,N]^{m-1}} h \right) \right],$$

and argue similarly, we obtain

$$\left\| M^{(t_2,t_3,\ldots,t_m)}h \right\|_{L^{(q_2,q_3,\ldots,q_m)}} \le C \|h\|_{L^{(q_2,q_3,\ldots,q_m)}}.$$

Thus $t_m < \min(q_2, q_3, \ldots, q_m)$ by the induction assumption. It remains to show that $t_m < q_1$. To this end, we consider the function of the form:

$$f(x_1, x_2, \dots, x_m) = \sum_{j=-\infty}^{\infty} \chi_{([jN, (j+1)N] \times [-N, N]^{m-1})}(x_1, x_2, \dots, x_m) h_j(x_m),$$

where $h_j \in L^{q_m}(\mathbb{R})$. Then for all (x_1, x_2, \ldots, x_m)

$$\chi_{(\mathbb{R}\times[-N,N]^{m-1})}(x_1,x_2,\ldots,x_m)M^{(\bar{t})}f(x_1,x_2,\ldots,x_m)$$

$$\geq \sum_{j=-\infty}^{\infty} \chi_{([jN,(j+1)N]\times[-N,N]^{m-1})}(x_1,x_2,\ldots,x_m)M^{(t_m)}\left[\chi_{[-N,N]}h_j\right](x_m).$$

We abbreviate

$$H_m(x_m) \equiv M^{(t_m)} \left[\chi_{[-N,N]} h_j \right] (x_m).$$

Hence, we obtain

$$\left\| \sum_{j=-\infty}^{\infty} \chi_{([jN,(j+1)N] \times [-N,N]^{m-1})} M^{(t_m)} \left[\chi_{[-N,N]} h_j \right] \right\|_{L^{\vec{q}}}$$

$$= \left\| \left(\sum_{j=-\infty}^{\infty} \int_{\mathbb{R}} \chi_{[jN,(j+1)N]} \left(H_m(\cdot_m) \right)^{q_1} dx_1 \right)^{\frac{1}{q_1}} \chi_{[-N,N]^{m-1}} \right\|_{L^{(q_2,\dots,q_m)}}$$

$$\sim (2N)^{\frac{1}{q_1}+\dots+\frac{1}{q_{m-1}}} \left\| \left(\sum_{j=-\infty}^{\infty} \left(H_m(\cdot_m) \right)^{q_1} \right)^{\frac{1}{q_1}} \right\|_{L^{q_m}}.$$

In the same way, we deduce

$$\left\| \chi_{(\mathbb{R} \times [-N,N]^{m-1})} M^{(\vec{t})} f \right\|_{L^{\vec{q}}} \\ \lesssim (2N)^{\frac{1}{q_1} + \dots + \frac{1}{q_{m-1}}} \left\| \left(\sum_{j=-\infty}^{\infty} \left(|\chi_{[-N,N]} h_j(\cdot_m)| \right)^{q_1} \right)^{\frac{1}{q_1}} \right\|_{L^{q_m}},$$

since $M^{(\vec{t})}$ is bounded. Thus, letting $N \to \infty$, we obtain

$$\left\| \{ M^{(t_m)} h_j \}_{j=-\infty}^{\infty} \right\|_{L^{q_m}(\ell^{q_1})} \le \left\| \{ h_j \}_{j=-\infty}^{\infty} \right\|_{L^{q_m}(\ell^{q_1})}$$

This forces $q_1 > t_m$.

5.3 Some observations of Hardy-mixed Morrey spaces

In this section, we consider the characterizations of Hardy-Morrey spaces. Concerning $\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$ and $\mathcal{H}\mathcal{M}^{p}_{\vec{q}}(\mathbb{R}^{n})$ when $\vec{q} > 1$, we have the following assertion:

Proposition 5.3.1. Let $1 and <math>1 < \vec{q} < \infty$ satisfy

$$\frac{n}{p} \le \sum_{j=1}^{n} \frac{1}{q_j}.$$

- 1. If $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, then $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.
- 2. If $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, then f can be represented by a locally integrable function and the representative belongs to $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

Proposition 5.3.1 was investigated in [66, 72] when $q_j = q$ for all j = 1, ..., n. To prove Proposition 5.3.1, we need the description of the predual spaces of mixed Morrey spaces (see Subsection 2.1.4).

Proof of Proposition 5.3.1.

1. Denote by $B(R) = \{x \in \mathbb{R}^n : |x| < R\}$ for R > 0. Since

$$||f||_{L^1(B(R))} \le CR^{-\frac{n}{p}+n} ||f||_{\mathcal{M}^p_{\vec{n}}}$$

we have $f \in \mathcal{S}'(\mathbb{R}^n)$. As is described in [29], we have a pointwise estimate $|e^{t\Delta}f| \leq Mf$, where M denotes the Hardy–Littlewood maximal operator. Since M is shown to be bounded in [19], we have $f \in H\mathcal{M}^p_{\overline{a}}(\mathbb{R}^n)$.

2. Let $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then $\{e^{t\Delta}f\}_{t>0}$ is a bounded set of $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, which admits a separable predual as we have seen in Lemma 2.1.30. Therefore, there exists a sequence $\{t_j\}_{j=1}^{\infty}$ decreasing to 0 such that $\{e^{t_j\Delta}f\}_{j=1}^{\infty}$ converges to a function gin the weak-* topology of $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Meanwhile, it can be shown that $\lim_{t\downarrow 0} e^{t\Delta}f = f$ in the topology of $\mathcal{S}'(\mathbb{R}^n)$ [126]. Since the weak-* topology of $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ is stronger than the topology of $\mathcal{S}'(\mathbb{R}^n)$, it follows that $f = g \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

Furthermore, Hardy-mixed Morrey spaces admit a characterization by using the grand maximal operator. To formulate the result, we recall the following two fundamental notions.

1. Topologize $\mathcal{S}(\mathbb{R}^n)$ by norms $\{p_N\}_{N\in\mathbb{N}}$ given by

$$p_N(\varphi) \equiv \sum_{|\alpha| \le N} \sup_{x \in \mathbb{R}^n} (1 + |x|)^N |\partial^{\alpha} \varphi(x)|$$

for each $N \in \mathbb{N}$. Define $\mathcal{F}_N \equiv \{\varphi \in \mathcal{S}(\mathbb{R}^n) : p_N(\varphi) \leq 1\}.$

2. Let $f \in \mathcal{S}'(\mathbb{R}^n)$. The grand maximal operator $\mathcal{M}f$ is given by

$$\mathcal{M}f(x) \equiv \sup\{|t^{-n}\psi(t^{-1}\cdot) * f(x)| : t > 0, \ \psi \in \mathcal{F}_N\} \quad (x \in \mathbb{R}^n),$$
(5.4)

where we choose and fix a large integer N.

The following proposition can be proved.

Proposition 5.3.2. Let $0 < \vec{q} < \infty, 0 < p < \infty$, and $\frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$. Then

$$\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \sim \|f\|_{H\mathcal{M}^p_{\vec{q}}}$$

for all $f \in \mathcal{S}'(\mathbb{R}^n)$.

When $p \leq 1$ and $q_1 = q_2 = \cdots = q_n$, this proposition represents the result for classical Morrey spaces which is in [72].

Meanwhile the proof of Proposition 5.3.2 is similar to Hardy spaces with variable exponents [21, 102]. We content ourselves with stating two fundamental estimates (5.5) and (5.6).

We define the (discrete) maximal function with respect to $e^{t\Delta}$ by

$$M_{\text{heat}}f(x) \equiv \sup_{j \in \mathbb{Z}} |e^{2^j \Delta} f(x)| \quad (x \in \mathbb{R}^n).$$

Suppose that we are given an integer $K \gg 1$. We write

$$M_{\text{heat}}^* f(x) \equiv \sup_{j \in \mathbb{Z}} \left(\sup_{y \in \mathbb{R}^n} \frac{|e^{2^j \Delta} f(y)|}{(1+4^j |x-y|^2)^K} \right) \quad (x \in \mathbb{R}^n).$$

The next lemma connects M_{heat}^* with M_{heat} in terms of the usual Hardy–Littlewood maximal operator M.

Lemma 5.3.3 ([102, Lemma 3.2], [125, §4]). For $0 < \theta < 1$, there exists K_{θ} so that for all $K \geq K_{\theta}$, we have

$$M_{\text{heat}}^* f(x) \le CM^{(\theta)} [M_{\text{heat}} f](x) = CM \left[\sup_{k \in \mathbb{Z}} |e^{2^k \Delta} f|^{\theta} \right] (x)^{\frac{1}{\theta}} \quad (x \in \mathbb{R}^n)$$
(5.5)

for any $f \in \mathcal{S}'(\mathbb{R}^n)$, where $M^{(\theta)}$ is the powered maximal operator given by

$$M^{(\theta)}g(x) \equiv M[|g|^{\theta}](x)^{\frac{1}{\theta}} \quad (x \in \mathbb{R}^n)$$

for measurable functions g.

In the course of the proof of [102, Theorem 3.3], it can be shown that

$$\mathcal{M}f(x) \sim \sup_{\tau \in \mathcal{F}_N, \, j \in \mathbb{Z}} |\tau^j * f(x)| \lesssim M_{\text{heat}}^* f(x)$$
 (5.6)

once we fix an integer $K \gg 1$ and $N \gg 1$.

With the fundamental pointwise estimates (5.5) and (5.6), Proposition 5.3.2 can be proved with ease.

Proof of Proposition 5.3.2. Take $\theta \in (0,1)$ with $\theta < \min\{q_1, \ldots, q_n, p\}$. Then, using the pointwise estimates (5.5) and (5.6) and the boundedness of the Hardy–Littlewood maximal operator, we have

$$\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|M^*_{\text{heat}}f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|M^{(\theta)}[M_{\text{heat}}f]\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|M_{\text{heat}}f\|_{\mathcal{M}^p_{\vec{q}}}$$

Since it is known that $\sup_{t>0} |e^{t\Delta}f(x)| \lesssim \mathcal{M}f(x)$ (see [141, p. 98]), we obtain

$$\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|M_{\text{heat}}f\|_{\mathcal{M}^p_{\vec{q}}} \le \left\|\sup_{t>0} |e^{t\Delta}f|\right\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}.$$

This is the desired result.

5.4 Proofs of the main theorems (Theorems 5.1.2, 5.1.3, 5.1.7, and 5.1.8)

5.4.1 Reconstruction of mixed Morrey spaces by atoms (Theorem 5.1.2)

Before we show the main theorems, we observe the relation to the maximal operator $M^{(\vec{t}')}$ and blocks.

Example 5.4.1. Suppose that $1 \leq t'_k < \min(q'_1, q'_2, \ldots, q'_k) < \infty$. If we let κ be the operator norm of the maximal operator $M^{(\vec{t}')}$ on $L^{\vec{q}'}(\mathbb{R}^n)$, whose finiteness is guaranteed by Theorem 5.1.1, then we obtain $\kappa^{-1}\chi_Q M^{(\vec{t}')}g$ is a (p', \vec{q}') -block modulo a multiplicative constant for any (p', \vec{q}') -block g. Indeed, it is supported on a cube Q and it satisfies

$$\left\|\kappa^{-1}\chi_Q M^{(\vec{t}')}g\right\|_{L^{\vec{q}'}} \le \|\chi_Q g\|_{L^{\vec{q}'}} = \|g\|_{L^{\vec{q}'}} \le |Q|^{\frac{1}{n}\sum_{j=1}^{n}\frac{1}{q_j'} - \frac{1}{p'}}.$$

Keeping in mind this observation, we turn to the proof.

Proof of Theorem 5.1.2. By decomposing Q_j suitably, we may suppose each Q_j is dyadic.

To prove this theorem, we resort to the duality. For the time being, we assume that there exists $N \in \mathbb{N}$ such that $\lambda_j = 0$ whenever $j \geq N$. Let us assume in addition that a_j are non-negative. Fix a non-negative (p', \vec{q}') -block $g \in \mathcal{H}_{\vec{q}'}^{p'}(\mathbb{R}^n)$ with the associated cube Q.

Assume first that each Q_j contains Q as a proper subset. If we group j's such that Q_j are identical, we can assume that Q_j is the *j*-th dyadic parent of Q for each $j \in \mathbb{N}$. Then by the Hölder inequality [11]

$$\int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x = \sum_{j=1}^\infty \lambda_j \int_Q a_j(x)g(x) \, \mathrm{d}x \le \sum_{j=1}^\infty \lambda_j \|a_j\|_{L^{\vec{q}}(Q)} \|g\|_{L^{\vec{q}'}(Q)}$$

from $f = \sum_{j=1}^{\infty} \lambda_j a_j$. Due to the size condition of a_j and g, we obtain

$$\begin{split} \int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x &\leq \sum_{j=1}^\infty \lambda_j |Q|^{\frac{1}{n} \sum_{j=1}^n \frac{1}{q_j} - \frac{1}{s}} |Q_j|^{\frac{1}{s}} |Q|^{\frac{1}{n} \sum_{j=1}^n \frac{1}{q_j'} - \frac{1}{p'}} \\ &\leq \sum_{j=1}^\infty \lambda_j |Q|^{\frac{1}{p} - \frac{1}{s}} |Q_j|^{\frac{1}{s}}. \end{split}$$

Note that

$$\left\|\sum_{j=1}^{\infty}\lambda_{j}\chi_{Q_{j}}\right\|_{\mathcal{M}^{p}_{\vec{q}}} \geq \left\|\lambda_{j_{0}}\chi_{Q_{j_{0}}}\right\|_{\mathcal{M}^{p}_{\vec{q}}} = |Q_{j_{0}}|^{\frac{1}{p}}\lambda_{j_{0}}$$

for each j_0 . Consequently, it follows from the condition p < s that

$$\int_{\mathbb{R}^n} f(x)g(x) \,\mathrm{d}x \le \sum_{j=1}^\infty |Q|^{\frac{1}{p}-\frac{1}{s}} |Q_j|^{\frac{1}{s}-\frac{1}{p}} \left\| \sum_{j=1}^\infty \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}} \le C \left\| \sum_{j=1}^\infty \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Conversely assume that Q contains each Q_j . Then by the Hölder inequality

$$\int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x = \sum_{j=1}^\infty \lambda_j \int_{Q_j} a_j(x)g(x) \, \mathrm{d}x \le \sum_{j=1}^\infty \lambda_j \|a_j\|_{L^{\vec{t}}(Q_j)} \|g\|_{L^{\vec{t}'}(Q_j)}.$$

Thanks to the condition of a_j , we obtain

$$\int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x \le \sum_{j=1}^\infty \lambda_j |Q_j|^{\frac{1}{n} \sum_{j=1}^n \frac{1}{t_j} - \frac{1}{s}} |Q_j|^{\frac{1}{s}} \|g\|_{L^{\bar{t}'}(Q_j)}.$$

Thus, in terms of the maximal operator $M^{(\vec{t}')}$ defined in Theorem 5.1.1, we obtain

$$\begin{split} \int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x &\leq \sum_{j=1}^\infty \lambda_j |Q_j| \times \inf_{y \in Q_j} M^{(\vec{t}')}g(y) \\ &\leq \int_{\mathbb{R}^n} \left(\sum_{j=1}^\infty \lambda_j \chi_{Q_j}(y) \right) M^{(\vec{t}')}g(y) \, \mathrm{d}y \\ &\leq \int_{\mathbb{R}^n} \left(\sum_{j=1}^\infty \lambda_j \chi_{Q_j}(y) \right) \chi_Q(y) M^{(\vec{t}')}g(y) \, \mathrm{d}y \end{split}$$

Hence, by Example 5.4.1, we obtain

$$\int_{\mathbb{R}^n} f(x)g(x) \, \mathrm{d}x \le \kappa \left\| \sum_{j=1}^\infty \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

This is the desired result. Finally, we can remove the assumption that $\lambda_j = 0$ for large j. Thus, the proof is complete.

5.4.2 Reconstruction of Hardy-mixed Morrey spaces by atoms (Theorem 5.1.7)

Before we move on the proof of Theorem 5.1.7, we prepare the estimate on the grand maximal operator \mathcal{M} . Recall again that the grand maximal operator \mathcal{M} was given by

$$\mathcal{M}f(x) = \sup\{|\varphi_t * f(x)| : \varphi \in \mathcal{F}_N, t > 0\} \quad (x \in \mathbb{R}^n).$$

Then we know that

$$\mathcal{M}a_j(x) \lesssim \chi_{3Q_j}(x) Ma_j(x) + (M\chi_{Q_j}(x))^{\frac{n+d_q+1}{n}},$$
(5.7)

where $d_q = \left[n\left(\frac{1}{v(\vec{q})} - 1\right)\right]$ and $v(\vec{q}) = \min(1, q_1, \dots, q_n)$. See [102, (5.2)] for more details. The first term can be controlled by an argument similar to Theorem 5.1.2. The second term can be handled by using the Fefferman–Stein maximal inequality for mixed Morrey spaces (see Proposition 2.1.24).

Let us show Theorem 5.1.7. Using Proposition 5.3.2 and (5.7), we have

$$\begin{split} \|f\|_{H\mathcal{M}^{p}_{\vec{q}}} &\sim \|\mathcal{M}f\|_{\mathcal{M}^{p}_{\vec{q}}} \\ &\leq \left\|\sum_{j=1}^{\infty} \lambda_{j} \mathcal{M}a_{j}\right\|_{\mathcal{M}^{p}_{\vec{q}}} \\ &\lesssim \left\|\sum_{j=1}^{\infty} \lambda_{j} \left(\chi_{3Q_{j}} Ma_{j} + (M\chi_{Q_{j}})^{\frac{n+d_{q}+1}{n}}\right)\right\|_{\mathcal{M}^{p}_{\vec{q}}} \\ &\lesssim \left\|\sum_{j=1}^{\infty} \lambda_{j} \chi_{3Q_{j}} Ma_{j}\right\|_{\mathcal{M}^{p}_{\vec{q}}} + \left\|\sum_{j=1}^{\infty} \lambda_{j} (M\chi_{Q_{j}})^{\frac{n+d_{q}+1}{n}}\right\|_{\mathcal{M}^{p}_{\vec{q}}} \equiv I_{1} + I_{2}. \end{split}$$

First, we consider I_1 . The proof is similar to Theorem 5.1.2. For the sake of completeness, we supply the proof. Thanks to decomposing Q_j suitably, we may suppose each Q_j is dyadic. We will use duality again. We assume that there exists $N \in \mathbb{N}$ such that $\lambda_j = 0$ whenever $j \ge N$. Let $r = \frac{p}{v(\vec{q})}$ and $\vec{w} = \frac{\vec{q}}{v(\vec{q})}$, so that $r, \vec{w} > 1$. Then,

$$\begin{split} \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{3Q_j} M a_j \right\|_{\mathcal{M}^p_{\vec{q}}} &\leq \left\| \left(\sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j} M a_j \right]^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}} \\ &= \left\| \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j} M a_j \right]^{v(\vec{q})} \right\|_{\mathcal{M}^r_{\vec{w}}}^{\frac{1}{v(\vec{q})}}. \end{split}$$

Fix a non-negative (r', \vec{w}') -block $g \in \mathcal{H}_{\vec{w}'}^{r'}(\mathbb{R}^n)$ with the associated cube Q. Assume first that each Q_j contains Q as a proper subset. If we group j's such that Q_j are identical, we can assume that Q_j is the *j*-th dyadic parent of Q for each $j \in \mathbb{N}$. Then, we calculate

$$\int_{\mathbb{R}^n} \sum_{j=1}^\infty \left[\lambda_j \chi_{3Q_j}(x) M a_j(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x = \sum_{j=1}^\infty \lambda_j^{v(\vec{q})} \int_Q \left[M a_j(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x.$$

Using the Hölder inequality twice, we obtain

$$\begin{split} \int_{\mathbb{R}^{n}} \sum_{j=1}^{\infty} \left[\lambda_{j} \chi_{3Q_{j}}(x) M a_{j}(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left\| [Ma_{j}]^{v(\vec{q})} \right\|_{L^{\vec{w}}(Q)} \|g\|_{L^{\vec{w}'}(Q)} \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \|Ma_{j}\|_{L^{\vec{q}}(Q)}^{v(\vec{q})} \|g\|_{L^{\vec{w}'}(Q)} \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left[\|Ma_{j}\|_{L^{\vec{t}}(Q)} |Q|^{\frac{1}{n}} \sum_{j=1}^{n} \left(\frac{1}{q_{j}} - \frac{1}{t_{j}} \right) \right]^{v(\vec{q})} \|g\|_{L^{\vec{w}'}(Q)}. \end{split}$$

Using the boundedness of the Hardy–Littlewood maximal operator on $\mathcal{M}^s_{\vec{t}}(\mathbb{R}^n),$ we have

$$\begin{split} \int_{\mathbb{R}^{n}} \sum_{j=1}^{\infty} \left[\lambda_{j} \chi_{3Q_{j}}(x) M a_{j}(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left[\left\| M a_{j} \right\|_{L^{\vec{t}}(Q)} \left| Q \right|^{\frac{1}{n} \sum_{j=1}^{n} \left(\frac{1}{q_{j}} - \frac{1}{t_{j}} \right)} \right]^{v(\vec{q})} \left\| g \right\|_{L^{\vec{w}'}(Q)} \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left| Q \right|^{\frac{1}{n} \sum_{j=1}^{n} \frac{1}{w_{j}}} \left[\left| Q \right|^{-\frac{1}{s}} \left\| M a_{j} \right\|_{\mathcal{M}^{s}_{\vec{t}}} \right]^{v(\vec{q})} \left\| g \right\|_{L^{\vec{w}'}(Q)} \\ &\lesssim \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left| Q \right|^{\frac{1}{n} \sum_{j=1}^{n} \frac{1}{w_{j}}} \left[\left| Q \right|^{-\frac{1}{s}} \left\| a_{j} \right\|_{\mathcal{M}^{s}_{\vec{t}}} \right]^{v(\vec{q})} \left\| g \right\|_{L^{\vec{w}'}(Q)}. \end{split}$$

Thus, using the size condition of a_j and g, we obtain

$$\begin{split} &\left(\int_{\mathbb{R}^{n}}\sum_{j=1}^{\infty}\left[\lambda_{j}\chi_{3Q_{j}}(x)Ma_{j}(x)\right]^{v(\vec{q})}g(x)\mathrm{d}x\right)^{\frac{1}{v(\vec{q})}}\\ &\lesssim \left(\sum_{j=1}^{\infty}\lambda_{j}^{v(\vec{q})}|Q|^{\frac{1}{n}}\sum_{j=1}^{n}\frac{1}{w_{j}}\left[|Q|^{-\frac{1}{s}}|Q_{j}|^{\frac{1}{s}}\right]^{v(\vec{q})}|Q|^{\frac{1}{n}}\sum_{j=1}^{n}\frac{1}{w_{j}'}^{-\frac{1}{r'}}\right)^{\frac{1}{v(\vec{q})}}\\ &= |Q|^{\frac{1}{p}-\frac{1}{s}}\left(\sum_{j=1}^{\infty}\left[\lambda_{j}|Q_{j}|^{\frac{1}{s}}\right]^{v(\vec{q})}\right)^{\frac{1}{v(\vec{q})}}. \end{split}$$

Note that

$$\left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}} \ge \left\| \lambda_{j_0} \chi_{Q_{j_0}} \right\|_{\mathcal{M}^p_{\vec{q}}} = \lambda_{j_0} |Q_{j_0}|^{\frac{1}{p}}$$

for each $j_0 \in \mathbb{N}$. Thus,

$$\left(\int_{\mathbb{R}^n} \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j}(x) M a_j(x)\right]^{v(\vec{q})} g(x) \mathrm{d}x\right)^{\frac{1}{v(\vec{q})}}$$
$$\lesssim \sum_{k=1}^{\infty} |Q|^{\frac{1}{p} - \frac{1}{s}} |Q_k|^{\frac{1}{s} - \frac{1}{p}} \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}}$$
$$\sim \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Conversely assume that Q contains each Q_j . Then by the Hölder inequality,

$$\begin{split} \int_{\mathbb{R}^n} \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j}(x) M a_j(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \\ &= \sum_{j=1}^{\infty} \lambda_j^{v(\vec{q})} \int_{3Q_j} \left[M a_j(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \\ &\leq \sum_{j=1}^{\infty} \lambda_j^{v(\vec{q})} \left\| \left[M a_j \right]^{v(\vec{q})} \right\|_{L^{\vec{\tau}}(3Q_j)} \|g\|_{L^{\vec{\tau}'}(3Q_j)} \quad \left(\vec{\tau} = \frac{\vec{t}}{v(\vec{q})} \right) \\ &\leq \sum_{j=1}^{\infty} \lambda_j^{v(\vec{q})} \|M a_j\|_{L^{\vec{t}}(3Q_j)}^{v(\vec{q})} \|g\|_{L^{\vec{\tau}'}(3Q_j)}. \end{split}$$

Additionally, by virtue of the boundedness of the Hardy–Littlewood maximal operator on $L^{\vec{t}}(\mathbb{R}^n),$ we have

$$\int_{\mathbb{R}^n} \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j}(x) M a_j(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \lesssim \sum_{j=1}^{\infty} \lambda_j^{v(\vec{q})} \|a_j\|_{L^{\vec{t}}}^{v(\vec{q})} \|g\|_{L^{\vec{t}'}(3Q_j)}.$$

Considering the condition of a_j , we obtain

$$\begin{split} &\int_{\mathbb{R}^{n}} \sum_{j=1}^{\infty} \left[\lambda_{j} \chi_{3Q_{j}}(x) M a_{j}(x) \right]^{v(\vec{q})} g(x) \mathrm{d}x \\ &\lesssim \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left[\left| Q_{j} \right|^{\frac{1}{n} \sum_{j=1}^{n} \frac{1}{t_{j}} - \frac{1}{s}} \left\| a_{j} \right\|_{\mathcal{M}^{s}_{\vec{t}}} \right]^{v(\vec{q})} \left\| g \right\|_{L^{\vec{\tau}'}(3Q_{j})} \\ &\leq \sum_{j=1}^{\infty} \lambda_{j}^{v(\vec{q})} \left[\left| Q_{j} \right|^{\frac{1}{n} \sum_{j=1}^{n} \frac{1}{t_{j}}} \right]^{v(\vec{q})} \left\| g \right\|_{L^{\vec{\tau}'}(3Q_{j})}. \end{split}$$

Thus, in terms of the maximal operator $M^{(\vec{t'})}$ defined in Theorem 5.1.1, we obtain

$$\begin{split} &\left(\int_{\mathbb{R}^n} \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j}(x) M a_j(x)\right]^{v(\vec{q})} g(x) \mathrm{d}x\right)^{\frac{1}{v(\vec{q})}} \\ &\leq \left(\sum_{j=1}^{\infty} \lambda_j^{v(\vec{q})} |Q_j| \times \inf_{y \in Q_j} M^{(\vec{\tau}\,')} g(y)\right)^{\frac{1}{v(\vec{q})}} \\ &\leq \left(\int_{\mathbb{R}^n} \left(\sum_{j=1}^{\infty} \left(\lambda_j \chi_{Q_j}(y)\right)^{v(\vec{q})}\right) M^{(\vec{\tau}\,')} g(y) \, \mathrm{d}y\right)^{\frac{1}{v(\vec{q})}} \\ &\leq \left(\int_{\mathbb{R}^n} \left(\sum_{j=1}^{\infty} \left(\lambda_j \chi_{Q_j}(y)\right)^{v(\vec{q})}\right) \chi_Q(y) M^{(\vec{\tau}\,')} g(y) \, \mathrm{d}y\right)^{\frac{1}{v(\vec{q})}} \end{split}$$

As in Example 5.4.1, $\kappa^{-1}\chi_Q M^{(\vec{\tau}\,')}g$ is a $(r', \vec{w}\,')$ -block as long as κ is the operator norm of $M^{(\vec{\tau}\,')}$ on $L^{\vec{q}\,'}(\mathbb{R}^n)$. Hence, we obtain

.

$$\left(\int_{\mathbb{R}^n} \sum_{j=1}^{\infty} \left[\lambda_j \chi_{3Q_j}(x) M a_j(x)\right]^{v(\vec{q})} g(x) \mathrm{d}x\right)^{\frac{1}{v(\vec{q})}}$$
$$\lesssim \kappa^{\frac{1}{v(\vec{q})}} \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})}\right) \right\|_{\mathcal{M}^r_{\vec{w}}}^{\frac{1}{v(\vec{q})}} = \kappa^{\frac{1}{v(\vec{q})}} \left\| \left(\sum_{j=1}^{\infty} (\lambda_j \chi_{Q_j})^{v(\vec{q})}\right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}^p_{\vec{q}}}$$

Next, we consider I_2 . Put

$$u = \frac{n + d_q + 1}{n}p, \quad \vec{v} = \frac{n + d_q + 1}{n}\vec{q}.$$

Then, by Proposition 2.1.24 and the embedding $\ell^{v(\vec{q})} \hookrightarrow \ell^1$, we have

$$I_{2} = \left\| \left[\sum_{j=1}^{\infty} \lambda_{j} (M\chi_{Q_{j}})^{\frac{n+d_{q}+1}{n}} \right]^{\frac{n}{n+d_{q}+1}} \right\|_{\mathcal{M}_{\vec{v}}^{u}}^{\frac{n+d_{q}+1}{n}} \lesssim \left\| \sum_{j=1}^{\infty} \lambda_{j} \chi_{Q_{j}} \right\|_{\mathcal{M}_{\vec{q}}^{p}}$$
$$\leq \left\| \left(\sum_{j=1}^{\infty} (\lambda_{j} \chi_{Q_{j}})^{v(\vec{q})} \right)^{\frac{1}{v(\vec{q})}} \right\|_{\mathcal{M}_{\vec{q}}^{p}}$$

Thus, we obtain the desired result.

5.4.3 Atomic decomposition for mixed Morrey spaces and Hardymixed Morrey spaces (Theorems 5.1.3 and 5.1.8)

Theorem 5.1.3 is included in Theorem 5.1.8 so that we concentrate in the proof of Theorem 5.1.8.

We invoke the following lemma. We refer to [66, Lemma 3.2] and [141, p.101–105].

Lemma 5.4.2. Let $f \in \mathcal{S}'(\mathbb{R}^n)$, $K \in \{0, 1, 2, ...\}$ and $j \in \mathbb{Z}$. Then, there are collections of cubes $\{Q_{j,k}^*\}_{k \in K_j}$ and functions $\{\eta_{j,k}\}_{k \in K_j} \subset C_c^{\infty}(\mathbb{R}^n)$, which are all indexed be a set K_j for every j, and a decomposition

$$f = g_j + b_j, \quad b_j = \sum_{k \in K_j} b_{j,k},$$

such that

(i) Define $\mathcal{O}_j \equiv \{y \in \mathbb{R}^n : \mathcal{M}f(y) > 2^j\}$ and consider its Whitney decomposition. Then, the cubes $\{200Q_{j,k}^*\}_{k \in K_j}$ have the bounded intersection property, and

$$\mathcal{O}_j = \bigcup_{k \in K_j} Q_{j,k}^* = \bigcup_{k \in K_j} 200Q_{j,k}^*.$$
(5.8)

(ii) Consider the partition of unity with respect to the collections of cubes $\{Q_{j,k}^*\}_{k \in K_j}$. Denote it by $\{\eta_{j,k}\}_{k \in K_j}$. Then each function $\eta_{j,k}$ is supported in $Q_{j,k}^*$ and

$$\sum_{k \in K_j} \eta_{j,k} = \chi_{\{y \in \mathbb{R}^n : \mathcal{M}f(y) > 2^j\}}, \quad 0 \le \eta_{j,k} \le 1.$$

(iii) The distribution g_i satisfies the inequality:

$$\mathcal{M}g_{j}(x) \leq C\left(\mathcal{M}f(x)\chi_{\mathcal{O}_{j}^{c}}(x) + 2^{j}\sum_{k \in K_{j}} \frac{\ell_{j,k}^{n+K+1}}{(\ell_{j,k} + |x - x_{j,k}|)^{n+K+1}}\right)$$

for all $x \in \mathbb{R}^n$.

(iv) Each distribution $b_{j,k}$ is given by $b_{j,k} = (f - c_{j,k})\eta_{j,k}$ with a certain polynomial $c_{j,k} \in \mathcal{P}_K(\mathbb{R}^n)$ satisfying

$$\int_{\mathbb{R}^n} b_{j,k}(x) q(x) \mathrm{d}x = 0$$

for all $q \in \mathcal{P}_K(\mathbb{R}^n)$, and

$$\mathcal{M}b_{j,k}(x) \le C\left(\mathcal{M}f(x)\chi_{Q_{j,k}^*}(x) + 2^j \cdot \frac{\ell_{j,k}^{n+K+1}}{|x - x_{j,k}|^{n+K+1}}\chi_{\mathbb{R}^n \setminus Q_{j,k}^*}\right)$$
(5.9)

for all $x \in \mathbb{R}^n$.

In the above $x_{j,k}$ and $\ell_{j,k}$ denote the center and the side-length of $Q_{j,k}^*$, respectively, and the implicit constants are dependent only on n.

Observe that (5.8) together with the bounded overlapping property, that is, every point is contained in at most a fixed number (we denote it by N) of the $\{Q_{j,k}^*\}$, yields

$$\chi_{\mathcal{O}_{j}}(x) \leq \sum_{k \in K_{j}} \chi_{Q_{j,k}^{*}}(x) \leq \sum_{k \in K_{j}} \chi_{200Q_{j,k}^{*}}(x) \leq N\chi_{\mathcal{O}_{j}}(x) \quad (x \in \mathbb{R}^{n}).$$
(5.10)

For the proof of Theorem 5.1.8, we need the following embedding.

Lemma 5.4.3. Let $0 < \vec{q} < \infty, 0 < p < \infty, \frac{n}{p} \leq \sum_{j=1}^{n} \frac{1}{q_j}$ and $q_0 = \min(q_1, \dots, q_n)$. Choose τ to be $1 - \frac{q_0}{p} < \tau < 1$. Then $\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) \hookrightarrow \mathcal{M}^p_{q_0}(\mathbb{R}^n) \hookrightarrow L^{q_0}(M\chi_{B(1)}^{\tau})(\mathbb{R}^n),$

where $L^q(w)$ denotes the weighted Lebesgue space with respect to the measure $w \cdot dx$ for a non-negative measurable function w.

Proof. The first embedding follows from Proposition 2.1.17. We shall show the second embedding. Let $f \in \mathcal{M}_{q_0}^p(\mathbb{R}^n)$. Then, we have

$$\begin{split} \|f\|_{L^{q_0}(M\chi_{B(1)}^{\tau})} &\leq \|f\|_{L^{q_0}(M\chi_{B(1)}^{\tau}\cdot\chi_{B(1)})} + \sum_{k=1}^{\infty} \|f\|_{L^{q_0}(M\chi_{B(1)}^{\tau}\cdot\chi_{B(2^k)\setminus B(2^{k-1})})} \\ &\lesssim \sum_{k=0}^{\infty} 2^{-\frac{nk\tau}{q_0}} \|f\|_{L^{q_0}(B(2^k))} \\ &\lesssim \sum_{k=0}^{\infty} 2^{-\frac{nk\tau}{q_0} + \frac{nk}{q_0} - \frac{nk}{p}} \|f\|_{\mathcal{M}^{p}_{q_0}}. \end{split}$$

Since $1 - \frac{q_0}{p} < \tau < 1, -\frac{n\tau}{q_0} + \frac{n}{q_0} - \frac{n}{p} < 0.$ Thus, $f \in L^{q_0}(M\chi_{B(1)}^{\tau})(\mathbb{R}^n).$

Recall that a non-negative measurable function w is A_1 -weight if w satisfies

$$Mw(x) \le Cw(x) \quad (x \in \mathbb{R}^n)$$

Lemma 5.4.4. Let $\varphi \in S(\mathbb{R}^n)$. Keep to the same notation as Lemmas 5.4.2 and 5.4.3. Then we have

$$|\langle b_j, \varphi \rangle| \le C \|\chi_{\mathcal{O}_j} \mathcal{M}f\|_{L^{q_0}(M\chi_{B(1)})}$$
(5.11)

and

$$|\langle g_j, \varphi \rangle| \le C \left\| \min(\mathcal{M}f, 2^j) \right\|_{L^{q_0}\left(M\chi_{B(1)}^{\tau}\right)}, \qquad (5.12)$$

where the constants in (5.11) and (5.12) depend on φ but not on j or k.

Proof. By the subadditivity of \mathcal{M} given by (5.4), we have

$$|\langle b_j, \varphi \rangle| \le C \inf_{x \in B(1)} \mathcal{M}b_j(x) \le C \inf_{x \in B(1)} \sum_{k \in K_j} \mathcal{M}b_{j,k}(x).$$

Observe also that

$$CM\chi_Q(x) \ge \frac{|Q|}{|Q| + |x - x_Q|^n},$$

if Q is a cube centered at x_Q . Take a and θ satisfying $q_0 > a$ and $(\theta - \tau) \left(\frac{q_0}{a}\right)' > 1$. It follows from (5.9) and Hölder's inequality that

$$\begin{split} &\left(\inf_{x\in B(1)}\sum_{k\in K_{j}}\mathcal{M}b_{j,k}(x)\right)^{a}\\ \lesssim &\int_{\mathbb{R}^{n}}\left(\sum_{k\in K_{j}}\mathcal{M}b_{j,k}(x)\right)^{a}\chi_{B(1)}(x)\mathrm{d}x\\ \lesssim &\int_{\mathbb{R}^{n}}\left(\mathcal{M}f(x)\chi_{\mathcal{O}_{j}}(x)+2^{j}\sum_{k\in K_{j}}M\chi_{Q_{j,k}^{*}}(x)^{\frac{n+K+1}{n}}\right)^{a}M\chi_{B(1)}(x)^{\theta}\mathrm{d}x\\ \lesssim &\left\|\left((\mathcal{M}f)\chi_{\mathcal{O}_{j}}+2^{j}\sum_{k\in K_{j}}M\chi_{Q_{j,k}^{*}}^{\frac{n+K+1}{n}}\right)^{a}\right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})}\left\|M\chi_{B(1)}^{\theta-\tau}\right\|_{L^{\left(\frac{q_{0}}{a}\right)'}}. \end{split}$$

Using the triangle inequality and the Fefferman–Stein weighted vector-valued inequality for A_1 -weights (see [6]), we obtain

$$\begin{split} \left(\inf_{x\in B(1)}\sum_{k\in K_{j}}\mathcal{M}b_{j,k}(x)\right)^{a} \\ \lesssim \left\|\left((\mathcal{M}f)\chi_{\mathcal{O}_{j}}\right)^{a}\right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} + \left\|\left(2^{j}\sum_{k\in K_{j}}M\chi_{Q_{j,k}^{*}}^{\frac{n+K+1}{n}}\right)^{a}\right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} \\ \lesssim \left\|\left((\mathcal{M}f)\chi_{\mathcal{O}_{j}}\right)^{a}\right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} + \left\|\left(2^{j}\sum_{k\in K_{j}}\chi_{Q_{j,k}^{*}}\right)^{a}\right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} \\ \lesssim \left\|(\mathcal{M}f)\chi_{\mathcal{O}_{j}}\right\|_{L^{q_{0}}(M\chi_{B(1)}^{\tau})}^{a} \cdot \end{split}$$

Thus, (5.11) is proved.

In the same way we can prove (5.12). In fact, using the Fefferman–Stein inequality

for A_1 -weighted Lebesgue spaces, we obtain

$$\begin{split} \left(\inf_{x\in B(1)}\mathcal{M}g_{j}(x)\right)^{a} \\ \lesssim \left\| \left((\mathcal{M}f)\,\chi_{\mathcal{O}_{j}^{c}} + 2^{j}\sum_{k\in K_{j}}\frac{\ell_{j,k}^{n+K+1}}{(\ell_{j,k}+|\cdot-x_{j,k}|)^{n+d+1}} \right)^{a} \right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} \\ \lesssim \left\| \left((\mathcal{M}f)\chi_{\mathcal{O}_{j}^{c}} \right)^{a} \right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} + \left\| \left(2^{j}\sum_{k\in K_{j}}M\chi_{Q_{j,k}^{*}}\frac{n+K+1}{n} \right)^{a} \right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} \\ \lesssim \left\| \left((\mathcal{M}f)\chi_{\mathcal{O}_{j}^{c}} \right)^{a} \right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} + \left\| \left(2^{j}\chi_{\mathcal{O}_{j}} \right)^{a} \right\|_{L^{\frac{q_{0}}{a}}(M\chi_{B(1)}^{\tau})} \\ \lesssim \left\| (\mathcal{M}f)\chi_{\mathcal{O}_{j}^{c}} + 2^{j}\chi_{\mathcal{O}_{j}} \right\|_{L^{q_{0}}(M\chi_{B(1)}^{\tau})}^{a} \\ = \left\| \min(\mathcal{M}f, 2^{j}) \right\|_{L^{q_{0}}(M\chi_{B(1)}^{\tau})}^{a} \end{split}$$

Thus, (5.12) is proven.

Lemma 5.4.5. In the notation of Lemmas 5.4.2 and 5.4.3, in the topology of $\mathcal{S}'(\mathbb{R}^n)$, we have $g_j \to 0$ as $j \to -\infty$ and $b_j \to 0$ as $j \to \infty$. In particular,

$$f = \sum_{j=-\infty}^{\infty} (g_{j+1} - g_j)$$

in the topology of $\mathcal{S}'(\mathbb{R}^n)$.

Proof. By Lemma 5.4.3, since

$$\|\mathcal{M}f\|_{L^{q_0}(M\chi_{B(1)})} \lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \sim \|f\|_{H\mathcal{M}^p_{\vec{q}}} < \infty$$

for $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$, we have $\mathcal{M}f \in L^{q_0}(M\chi_{B(1)}^{\tau})$.

Let us show that $b_j \to 0$ as $j \to \infty$. Once this is proved, then we have $f = \lim_{j \to \infty} g_j$ in $\mathcal{S}'(\mathbb{R}^n)$. Let us choose a test function $\varphi \in \mathcal{S}(\mathbb{R}^n)$. Then by (5.11) we have

$$|\langle b_j, \varphi \rangle| \lesssim \|\chi_{\mathcal{O}_j} \mathcal{M} f\|_{L^{q_0}(M\chi_{B(1)}^{\tau})}).$$

Hence it follows that $b_j \to 0$ as $j \to \infty$. Likewise by (5.12),

$$|\langle g_j, \varphi \rangle| \lesssim \|\min(\mathcal{M}f, 2^j)\|_{L^{q_0}(M\chi_{B(1)}\tau)}).$$

Thus, $g_j \to 0$ as $j \to -\infty$. Consequently, we have $f = \lim_{j \to \infty} (g_j - g_{-j}) = \sum_{j = -\infty}^{\infty} (g_{j+1} - g_j)$ in $\mathcal{S}'(\mathbb{R}^n)$.

To prove Theorem 5.1.8, we first assume $f \in L^1_{\text{loc}}(\mathbb{R}^n)$. After that, we show this theorem for all $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$.

Lemma 5.4.6. Theorem 5.1.8 holds for $f \in L^1_{loc}(\mathbb{R}^n)$.

Proof. Assume that $f \in L^1_{loc}(\mathbb{R}^n)$. For each $j \in \mathbb{Z}$, consider the level set \mathcal{O}_j as in Lemma 5.4.2. Then it follows immediately from the definition that

$$\mathcal{O}_{j+1} \subset \mathcal{O}_j.$$

If we invoke Lemma 5.4.2, then f can be decomposed;

$$f = g_j + b_j, \quad b_j = \sum_k b_{j,k}, \quad b_{j,k} = (f - c_{j,k})\eta_{j,k}$$

where each $b_{j,k}$ is supported in a cube $Q_{j,k}^*$ as is described in Lemma 5.4.2.

We know that

$$f = \sum_{j=-\infty}^{\infty} (g_{j+1} - g_j),$$

in $\mathcal{S}'(\mathbb{R}^n)$. Here, going through the same argument as the one in [141, p.108–109], we have an expression;

$$f = \sum_{j,k} A_{j,k}, \quad g_{j+1} - g_j = \sum_k A_{j,k} \quad (j \in \mathbb{Z})$$

in the sense of distributions, where each $A_{j,k}$ satisfies the pointwise estimate $|A_{j,k}(x)| \leq C_0 2^j \chi_{Q_{j,k}^*}(x)$, and belongs to $\mathcal{P}_K(\mathbb{R}^n)^{\perp}$. With these observations in mind, let us set

$$a_{j,k} \equiv \frac{A_{j,k}}{C_0 2^j}, \quad \kappa_{j,k} \equiv C_0 2^j.$$

Then we automatically obtain that each $a_{j,k}$ belongs to $\mathcal{P}_K(\mathbb{R}^n)^{\perp}$ and satisfies that

 $|a_{j,k}| \le \chi_{Q_{j,k}^*},$

and that $f = \sum_{j,k} \kappa_{j,k} a_{j,k}$ in the topology of $\mathcal{S}'(\mathbb{R}^n)$, once we prove the estimate of coefficients. Put $\Lambda = \{(j,k) : j \in \mathbb{Z}, k \in K_j\}$ and fix a bijection $\mu : \Lambda \to \mathbb{N}$. Then, we put

$$Q_{\mu(j,k)} = Q_{j,k}^*, \quad \lambda_{\mu(j,k)} = \lambda_{j,k}^*, \quad a_{\mu(j,k)} = a_{j,k}^*.$$

To establish (5.2) we need to estimate

$$\alpha \equiv \left\| \left(\sum_{(j,k)\in\Lambda}^{\infty} |\lambda_{\mu(j,k)} \chi_{Q_{\mu(j,k)}}|^{v} \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

Since $\{(\kappa_{j,k}, Q_{j,k}^*)\}_{j,k} = \{(\lambda_{\mu(j,k)}, Q_{\mu(j,k)})\}_{(j,k)\in\Lambda}$, we have

$$\alpha = \left\| \left(\sum_{j=-\infty}^{\infty} \sum_{k \in K_j} |\kappa_{j,k} \chi_{Q_{j,k}^*}|^v \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

1.11

If we insert the definition of $\kappa_{j,k}$ into the definition of α , then we have

$$\alpha = C_0 \left\| \left(\sum_{j=-\infty}^{\infty} \sum_{k \in K_j} |2^j \chi_{Q_{j,k}^*}|^v \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}} = C_0 \left\| \left(\sum_{j=-\infty}^{\infty} 2^{jv} \sum_{k \in K_j} \chi_{Q_{j,k}^*} \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^p_{\vec{q}}}.$$

Thus, by (5.10), we obtain

$$\alpha \le C \left\| \left(\sum_{j=-\infty}^{\infty} \left(2^{j} \chi_{\mathcal{O}_{j}} \right)^{v} \right)^{\frac{1}{v}} \right\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

Since $\mathcal{O}_j \supset \mathcal{O}_{j+1}$ for each $j \in \mathbb{Z}$, for $x \in \mathbb{R}^n$ we have

$$\sum_{j=-\infty}^{\infty} \left(2^j \chi_{\mathcal{O}_j}(x)\right)^v \le \sum_{j=-\infty}^{\lfloor \log_2 \mathcal{M}f(x) \rfloor + 1} \left(2^j \chi_{\mathcal{O}_j}(x)\right)^v \sim 2^{\lfloor \log_2 \mathcal{M}f(x) \rfloor + 1} \sim \mathcal{M}f(x).$$

Hence, we conclude $\alpha \leq C \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \sim \|f\|_{H\mathcal{M}^p_{\vec{q}}}$. This is the desired result.

Note that by Lemma 5.4.6, for $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n) \cap L^1_{\text{loc}}(\mathbb{R}^n)$, there exists a decomposition:

$$f = \sum_{j=1}^{\infty} \lambda_j a_j,$$

where $a_j \in \mathcal{P}_K^{\perp}(\mathbb{R}^n), \, \lambda_j \ge 0$ and

$$|a_j| \le \chi_{Q_j}, \quad \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}.$$

By replacing each Q_j by a dyadic cube suitably, we have a decomposition:

$$f = \sum_{Q' \in \mathcal{D}} \lambda_{Q'} a_{Q'},$$

where $a_{Q'} \in \mathcal{P}_K^{\perp}(\mathbb{R}^n), \, \lambda_{Q'} \ge 0$ and

$$|a_{Q'}| \le \chi_{3Q'}, \quad \left\| \sum_{Q' \in \mathcal{D}}^{\infty} \lambda_{Q'} \chi_{3Q'} \right\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}.$$

Here, \mathcal{D} denotes the set of all dyadic cubes in \mathbb{R}^n .

Let us prove Theorem 5.1.8.

Proof of Theorem 5.1.8. Let $f \in H\mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$. Then we consider the decomposition:

$$e^{t\Delta}f = \sum_{Q \in \mathcal{D}} \lambda_Q^t a_Q^t$$

in the topology of $\mathcal{S}'(\mathbb{R}^n),$ where $a_Q^t\in\mathcal{P}_K^{\perp}(\mathbb{R}^n),\,\lambda_Q^t\geq 0$ and

$$|a_Q^t| \le \chi_{3Q}, \quad \left\| \sum_{Q \in \mathcal{D}} \lambda_Q^t \chi_{3Q} \right\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|\mathcal{M}[e^{t\Delta}f]\|_{\mathcal{M}^p_{\vec{q}}} \lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}.$$

Due to the weak-* compactness of the unit ball of $L^{\infty}(\mathbb{R}^n)$, there exists a sequence $\{t_l\}_{l=1}^{\infty}$ that converges to 0 such that

$$\lambda_Q = \lim_{l \to \infty} \lambda_Q^{t_l}, \quad a_Q = \lim_{l \to \infty} a_Q^{t_l}$$

exist for all $Q \in \mathcal{D}$ in the sense that

$$\lim_{l \to \infty} \int_{\mathbb{R}^n} a_Q^{t_l}(x)\varphi(x) \mathrm{d}x = \int_{\mathbb{R}^n} a_Q(x)\varphi(x) \mathrm{d}x$$

for all $\varphi \in L^1(\mathbb{R}^n)$. We claim

$$f = \sum_{Q \in \mathcal{D}} \lambda_Q a_Q$$

in the topology of $\mathcal{S}'(\mathbb{R}^n)$. Let $\varphi \in \mathcal{S}(\mathbb{R}^n)$ be a test function. Then we have

$$\langle f, \varphi \rangle = \lim_{l \to \infty} \langle e^{t_l \Delta} f, \varphi \rangle = \lim_{l \to \infty} \sum_{Q \in \mathcal{D}} \lambda_Q^{t_l} \int_{\mathbb{R}^n} a_Q^{t_l}(x) \varphi(x) \mathrm{d}x$$

from the definition of the convergence in $\mathcal{S}'(\mathbb{R}^n)$. Once we fix m, we have

$$|\lambda_Q^{t_l}| \lesssim \frac{\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}}{\|\chi_{[0,2^{-m})^n}\|_{\mathcal{M}^p_{\vec{q}}}}$$
(5.13)

and

$$\left|\int_{\mathbb{R}^n} a_Q^{t_l}(x)\varphi(x)\mathrm{d}x\right| \leq \int_{3Q} |\varphi(x)|\mathrm{d}x.$$

Since

$$\sum_{Q \in \mathcal{D}_m} \frac{\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}}{\|\chi_{[0,2^{-m})^n}\|_{\mathcal{M}^p_{\vec{q}}}} \int_{3Q} |\varphi(x)| \mathrm{d}x = 3^n \frac{\|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}}{\|\chi_{[0,2^{-m})^n}\|_{\mathcal{M}^p_{\vec{q}}}} \|\varphi\|_{L^1} < \infty,$$

we are in the position of using the Fubini theorem to have

$$\sum_{m\in\mathbb{Z}}\int_{\mathbb{R}^n}\left(\sum_{Q\in\mathcal{D}_m}\lambda_Q^{t_l}a_Q^{t_l}(x)\right)\varphi(x)\mathrm{d}x=\sum_{m\in\mathbb{Z}}\sum_{Q\in\mathcal{D}_m}\lambda_Q^{t_l}\int_{\mathbb{R}^n}a_Q^{t_l}(x)\varphi(x)\mathrm{d}x.$$

With this in mind, let us set

$$a_{m,l} \equiv \sum_{Q \in \mathcal{D}_m} \lambda_Q^{t_l} \int_{\mathbb{R}^n} a_Q^{t_l}(x) \varphi(x) \mathrm{d}x$$

for each $m \in \mathbb{Z}$ and $l \in \mathbb{N}$. Then we have

$$|a_{m,l}| \le C2^{\frac{nm}{p}} \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}} \|\varphi\|_1 \quad (m \in \mathbb{Z})$$
(5.14)

thanks to (5.13).

Let $m \in \mathbb{Z}$. Then we have

$$\begin{aligned} a_{m,l} &= \sum_{Q \in \mathcal{D}_m} \lambda_Q^{t_l} \int_{3Q} a_Q^{t_l}(x) \varphi(x) \mathrm{d}x \\ &= \sum_{Q \in \mathcal{D}_m} \lambda_Q^{t_l} \int_{3Q} a_Q^{t_l}(x) \left(\varphi(x) - \sum_{|\beta| \le K} \frac{1}{\beta!} \partial^\beta \varphi(c(Q))(x - c(Q))^\beta \right) \mathrm{d}x \end{aligned}$$

since $a_Q^{t_l} \in \mathcal{P}_K^{\perp}(\mathbb{R}^n)$. Thus, by the mean-value theorem, we have

$$|a_{m,l}| \le C(\varphi) \sum_{Q \in \mathcal{D}_m} |\lambda_Q^{t_l}| \ell(Q)^{n+K+1} \sup_{y \in 3Q} \frac{1}{1+|y|^{n+1}}.$$
(5.15)

Here $C(\varphi)$ is a constant depending on φ .

Meanwhile, for each $\tilde{m} \in \mathbb{Z}^n$, we have

$$\left\|\sum_{Q\in\mathcal{D}_m,|c(Q)-\tilde{m}|\leq n}\lambda_Q^{t_l}\chi_Q\right\|_{\mathcal{M}^p_{q_0}}\lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}},$$

which implies

$$\left\|\sum_{Q\in\mathcal{D}_m,|c(Q)-\tilde{m}|\leq n}\lambda_Q^{t_l}\chi_Q\right\|_{L^{q_0}}\lesssim \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}$$

or equivalently

$$\left(\sum_{Q\in\mathcal{D}_m,|c(Q)-\tilde{m}|\leq n} 2^{-mn} |\lambda_Q^{t_l}|^{q_0}\right)^{\frac{1}{q_0}} \lesssim \|\mathcal{M}f\|_{\mathcal{M}_{q}^{p}}$$

Since $\ell^{q_0}(\mathbb{Z}^n) \hookrightarrow \ell^1(\mathbb{Z}^n)$,

$$\sum_{Q \in \mathcal{D}_m, |c(Q) - \tilde{m}| \le n} |\lambda_Q^{t_l}| \lesssim 2^{\frac{mn}{q_0}} \|\mathcal{M}f\|_{\mathcal{M}^p_{\vec{q}}}.$$

Combining this estimate with (5.15), we obtain

$$|a_{m,l}| \lesssim \sum_{\tilde{m} \in \mathbb{Z}^n} \sum_{Q \in \mathcal{D}_m, |c(Q) - \tilde{m}| \le n} |\lambda_Q^{t_l}| \ell(Q)^{n+K+1} \sup_{y \in 3Q} \frac{1}{1 + |y|^{n+1}}$$
$$\sim \sum_{\tilde{m} \in \mathbb{Z}^n} \sum_{Q \in \mathcal{D}_m, |c(Q) - \tilde{m}| \le n} \frac{|\lambda_Q^{t_l}| \ell(Q)^{n+K+1}}{1 + |\tilde{m}|^{n+1}}$$
$$\lesssim 2^{\frac{mn}{q_0} - (n+K+1)m} \|\mathcal{M}f\|_{\mathcal{M}_{\vec{q}}^p}.$$
(5.16)

Since $K + 1 > n\left(\frac{1}{q_0} - 1\right)$, we obtain

$$n+K+1 > \frac{n}{q_0}.$$

Thus by (5.14) and (5.16), we obtain

$$|a_{m,l}| \lesssim \min(2^{\frac{mn}{q_0} - (n+K+1)m}, 2^{\frac{mn}{p}}).$$

Since

$$\sum_{m=-\infty}^{\infty} \min(2^{\frac{mn}{q_0} - (n+K+1)m}, 2^{\frac{mn}{p}}) \lesssim 1,$$

we are in the position of using the Lebesgue convergence theorem to have

$$\lim_{l \to \infty} \sum_{m = -\infty}^{\infty} a_{m,l} = \sum_{m = -\infty}^{\infty} \left(\lim_{l \to \infty} a_{m,l} \right).$$

That is,

$$\langle f,\varphi\rangle = \lim_{l\to\infty} \langle e^{t_l\Delta}f,\varphi\rangle = \sum_{m=-\infty}^\infty \left(\lim_{l\to\infty}\sum_{Q\in\mathcal{D}_m}\lambda_Q^{t_l}\int_{\mathbb{R}^n}a_Q^{t_l}(x)\varphi(x)\mathrm{d}x\right).$$

Hence, using Fubini's theorem again, we obtain

$$\begin{split} \langle f, \varphi \rangle &= \sum_{m=-\infty}^{\infty} \left(\lim_{l \to \infty} \int_{\mathbb{R}^n} \left(\sum_{Q \in \mathcal{D}_m} \lambda_Q^{t_l} a_Q^{t_l}(x) \right) \varphi(x) \mathrm{d}x \right) \\ &= \sum_{m=-\infty}^{\infty} \sum_{Q \in \mathcal{D}_m} \lim_{l \to \infty} \left(\int_{\mathbb{R}^n} \lambda_Q^{t_l} a_Q^{t_l}(x) \varphi(x) \mathrm{d}x \right) \\ &= \sum_{m=-\infty}^{\infty} \sum_{Q \in \mathcal{D}_m} \int_{\mathbb{R}^n} \lambda_Q a_Q(x) \varphi(x) \mathrm{d}x = \left\langle \sum_{Q \in \mathcal{D}} \lambda_Q a_Q, \varphi \right\rangle. \end{split}$$

Consequently, we obtain the desired result.

5.5 Application : Olsen's inequality on mixed Morrey spaces

As an application of Theorem 5.1.2, we can prove the following Olsen inequality about the fractional integral operator I_{α} . Based upon Proposition 2.1.22, we can prove the following result.

Theorem 5.5.1. Suppose that the parameters α , p, \vec{q} , p^* , \vec{q}^* , s, \vec{t} satisfy

$$1 < p, p^*, s < \infty, \quad 1 < \vec{q}, \vec{q}^*, \vec{t} < \infty,$$
$$\frac{n}{p} \le \sum_{j=1}^n \frac{1}{q_j}, \quad \frac{n}{p^*} \le \sum_{j=1}^n \frac{1}{q_j^*}, \quad \frac{n}{s} \le \sum_{j=1}^n \frac{1}{t_j},$$
$$\max\{t_1, \dots, t_j\} < q_j^*, \quad \frac{1}{p} > \frac{\alpha}{n}, \quad \frac{1}{p^*} \le \frac{\alpha}{n},$$

for each $j = 1, 2, \ldots, n$, and that

$$\frac{1}{s} = \frac{1}{p^*} + \frac{1}{p} - \frac{\alpha}{n}, \quad \frac{t_j}{s} = \frac{q_j}{p} \quad (j = 1, 2, \dots, n).$$

Then for all $f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n)$ and $g \in \mathcal{M}^{p^*}_{\vec{q}^*}(\mathbb{R}^n)$

$$\|g \cdot I_{\alpha}f\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \cdot \|f\|_{\mathcal{M}^{p}_{\vec{q}}},$$

where the constant C is independent of f and g.

This result recaptures [134, Proposition 1.8]. Note that a detailed calculation in [133, p.6] shows that Theorem 5.5.1 is not just a combination of Proposition 2.1.22 and Lemma 5.5.2.

Lemma 5.5.2. Suppose that the parameters $p, \vec{q}, p^*, \vec{q}^*, s, \vec{t}$ satisfy

$$1 < p, p^*, s < \infty, \quad 1 < \vec{q}, \vec{q}^*, t < \infty,$$
$$\frac{n}{p} \le \sum_{j=1}^n \frac{1}{q_j}, \quad \frac{n}{p^*} \le \sum_{j=1}^n \frac{1}{q_j^*}, \quad \frac{n}{s} \le \sum_{j=1}^n \frac{1}{t_j}.$$

Assume

$$\frac{1}{s} = \frac{1}{p^*} + \frac{1}{p}, \quad \frac{1}{t_j} = \frac{1}{q_j^*} + \frac{1}{q_j}.$$

Then

$$\|f \cdot g\|_{\mathcal{M}^s_{\vec{t}}} \le \|f\|_{\mathcal{M}^p_{\vec{q}}} \|g\|_{\mathcal{M}^{p^*}_{\vec{q}^*}} \quad (f \in \mathcal{M}^p_{\vec{q}}(\mathbb{R}^n), g \in \mathcal{M}^{p^*}_{\vec{q}^*}(\mathbb{R}^n)).$$

We can prove this lemma easily by using Hölder's inequality. So we omit the proof. We write $\infty' = 1$ and $s' = \frac{s}{s-1}$ for $1 < s < \infty$. We have the following proposition: **Proposition 5.5.3.** In addition to the assumption in Theorem 5.5.1, suppose that $u \in (1, \infty]$ satisfies $u' < \min\{q_1, q_2, \ldots, q_n, p\}$. Let $\Omega \in L^s(\mathbb{S}^{n-1})$ be homogeneous of degree zero, that is, Ω satisfies, for any $\lambda > 0$, $\Omega(\lambda x) = \Omega(x)$. Then,

$$\left\|g \cdot I_{\Omega,\alpha}(f)\right\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \left\|g\right\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \left\|\Omega\right\|_{L^{u}(\mathbb{S}^{n-1})} \left\|f\right\|_{\mathcal{M}^{p}_{\vec{q}}},$$

where

$$I_{\Omega,\alpha}f(x) \equiv \int_{\mathbb{R}^n} \frac{\Omega(x-y)}{|x-y|^{n-\alpha}} f(y) \mathrm{d}y.$$

Proposition 5.5.3 is a direct consequence of Theorem 5.5.1, the next lemma and the boundedness of the Hardy–Littlewood maximal operator M.

Lemma 5.5.4. [64] If $1 < u \le \infty$, then we have

$$|I_{\Omega,\alpha}f(x)| \le C \, \|\Omega\|_{L^u(\mathbb{S}^{n-1})} \, |I_\alpha F(x)|,$$

where $F(x) \equiv M\left(|f|^{u'}\right)(x)^{\frac{1}{u'}}$.

First, we prove two lemmas. We invoke an estimate from [30, Lemma 2.2] and [31, Lemma 2.1].

Lemma 5.5.5. There exists a constant depending only on n and α such that, for every cube Q, we have $I_{\alpha}\chi_Q(x) \ge C\ell(Q)^{\alpha}\chi_Q(x)$ for all $x \in Q$.

To prove the next estimate, we use Proposition 2.1.22. We invoke another estimate from [66, Lemma 4.2].

Lemma 5.5.6. Let K = 0, 1, 2, ... Suppose that A is an $L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}_K(\mathbb{R}^n)^{\perp}$ -function supported on a cube Q. Then,

$$|I_{\alpha}A(x)| \le C_{\alpha,K} ||A||_{L^{\infty}} \ell(Q)^{\alpha} \sum_{k=1}^{\infty} \frac{1}{2^{k(n+K+1-\alpha)}} \chi_{2^{k}Q}(x) \quad (x \in \mathbb{R}^{n}).$$

Now we prove Theorem 5.5.1. We may assume that $f \in L_c^{\infty}(\mathbb{R}^n)$ is a positive measurable function in view of the positivity of the integral kernel. We decompose faccording to Theorem 5.1.3 with $K > \alpha - \frac{n}{p^*} - 1$; $f = \sum_{j=1}^{\infty} \lambda_j a_j$, where $\{Q_j\}_{j=1}^{\infty} \subset \mathcal{D}(\mathbb{R}^n)$, $\{a_j\}_{j=1}^{\infty} \subset L^{\infty}(\mathbb{R}^n) \cap \mathcal{P}_K(\mathbb{R}^n)^{\perp}$ and $\{\lambda_j\}_{j=1}^{\infty} \subset [0, \infty)$ fulfill (5.1). Then by Lemma 5.5.6, we obtain

$$|g(x)I_{\alpha}f(x)| \le C \sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \frac{\lambda_j}{2^{k(n+K+1-\alpha)}} \left(\ell(Q_j)^{\alpha} |g(x)| \chi_{2^k Q_j}(x) \right).$$

Therefore, we conclude

$$\|g \cdot I_{\alpha}f\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \left\| \sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \frac{\lambda_{j}\ell (2^{k}Q_{j})^{\alpha-\frac{n}{p^{*}}}}{2^{k(n+K+1)}} \cdot \frac{\ell (2^{k}Q_{j})^{\frac{n}{p^{*}}}}{\|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}}} |g|\chi_{2^{k}Q_{j}}\right\|_{\mathcal{M}^{s}_{\vec{t}}}.$$

For each $(j, k) \in \mathbb{N} \times \mathbb{N}$, write

$$\kappa_{jk} \equiv \frac{\lambda_j \ell (2^k Q_j)^{\alpha - \frac{n}{p^*}}}{2^{k(n+K+1)}}, \quad b_{jk} \equiv \frac{\ell (2^k Q_j)^{\frac{n}{p^*}}}{\|g\|_{\mathcal{M}_{\vec{q}^*}^{p^*}}} |g|\chi_{2^k Q_j}|$$

Then,

$$\sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \frac{\lambda_j \ell (2^k Q_j)^{\alpha - \frac{n}{p^*}}}{2^{k(n+K+1)}} \cdot \frac{\ell (2^k Q_j)^{\frac{n}{p^*}}}{\|g\|_{\mathcal{M}_{\vec{q}^*}^{p^*}}} |g|\chi_{2^k Q_j} = \sum_{j,k=1}^{\infty} \kappa_{jk} b_{jk},$$

each b_{jk} is supported on a cube $2^k Q_j$ and

$$\|b_{jk}\|_{\mathcal{M}^{p^*}_{\vec{q}^*}} \le \ell (2^k Q_j)^{\frac{n}{p^*}}.$$

Observe that $\chi_{2^kQ_j} \leq 2^{kn}M\chi_{Q_j}$. Hence, if we choose $1 < \theta$ so that

$$K > \alpha - \frac{n}{p^*} - 1 + \theta n - n,$$

then we have

$$\begin{aligned} \left\| \sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \kappa_{jk} \chi_{2^{k}Q_{j}} \right\|_{\mathcal{M}_{t}^{s}} &= \left\| \sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \frac{\lambda_{j} \ell(2^{k}Q_{j})^{\alpha - \frac{n}{p^{*}}}}{2^{k(n+K+1)}} \chi_{2^{k}Q_{j}} \right\|_{\mathcal{M}_{t}^{s}} \\ &= \left\| \sum_{j=1}^{\infty} \lambda_{j} \ell(Q_{j})^{\alpha - \frac{n}{p^{*}}} (M\chi_{Q_{j}})^{\theta} \right\|_{\mathcal{M}_{t}^{s}} \\ &\leq C \left\| \sum_{j=1}^{\infty} \left(M \left[\lambda_{j}^{\frac{1}{\theta}} \ell(Q_{j})^{\frac{1}{\theta}(\alpha - \frac{n}{p^{*}})} \chi_{Q_{j}} \right] \right)^{\theta} \right\|_{\mathcal{M}_{t}^{s}} \\ &\leq C \left(\left\| \left\{ \sum_{j=1}^{\infty} \left(M \left[\lambda_{j}^{\frac{1}{\theta}} \ell(Q_{j})^{\frac{1}{\theta}(\alpha - \frac{n}{p^{*}})} \chi_{Q_{j}} \right] \right)^{\theta} \right\}^{\frac{1}{\theta}} \right\|_{\mathcal{M}_{\theta_{q}}^{\theta}} \end{aligned}$$

Thanks to Proposition 2.1.24 with

$$f_j = \lambda_j^{\frac{1}{\theta}} \ell(Q_j)^{\frac{1}{\theta}(\alpha - \frac{n}{p^*})} \chi_{Q_j},$$

we can remove the maximal operator and we obtain

$$\left\|g \cdot I_{\alpha}f\right\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \left\|g\right\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \left\|\sum_{j=1}^{\infty} \lambda_{j}\ell(Q_{j})^{\alpha-\frac{n}{p^{*}}}\chi_{Q_{j}}\right\|_{\mathcal{M}^{s}_{\vec{t}}}$$

We distinguish two cases here.

1. If
$$\alpha = \frac{n}{p^*}$$
, then $p = s$ and $\vec{q} = \vec{t}$. Thus, we can use (5.1).

2. If $\alpha > \frac{n}{p^*}$, then, by Proposition 2.1.22 and Lemma 5.5.5, we obtain

$$\left\| \sum_{j=1}^{\infty} \lambda_j \ell(Q_j)^{\alpha - \frac{n}{p^*}} \chi_{Q_j} \right\|_{\mathcal{M}^s_{t}} \leq C \left\| I_{\alpha - \frac{n}{p^*}} \left[\sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right] \right\|_{\mathcal{M}^s_{t}} \leq C \left\| \sum_{j=1}^{\infty} \lambda_j \chi_{Q_j} \right\|_{\mathcal{M}^p_{q}}.$$

Thus, we are still in the position of using (5.1).

Consequently, we obtain

$$\left\|\sum_{j=1}^{\infty}\sum_{k=1}^{\infty}\kappa_{jk}\chi_{2^{k}Q_{j}}\right\|_{\mathcal{M}^{s}_{\vec{t}}} \lesssim \|f\|_{\mathcal{M}^{p}_{\vec{q}}} < \infty.$$

$$(5.17)$$

Observe also that $p^* > s$ and that $\vec{q}^* > \vec{t}$. Thus, by Theorem 5.1.2 and (5.17), it follows that

$$\|g \cdot I_{\alpha}f\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \left\| \sum_{j=1}^{\infty} \sum_{k=1}^{\infty} \kappa_{jk} \chi_{2^{k}Q_{j}} \right\|_{\mathcal{M}^{s}_{\vec{t}}} \leq C \|g\|_{\mathcal{M}^{p^{*}}_{\vec{q}^{*}}} \|f\|_{\mathcal{M}^{p}_{\vec{q}}}.$$

Bibliography

- [1] D. R. Adams, A note on Riesz potentials, Duke Math. J. 42 (1975), 765–778.
- [2] D. R. Adams and J. Xiao, Nonlinear potential analysis on Morrey spaces and their capacities, Indiana Univ. Math. J. 53 (6) (2004) 1629–1663.
- [3] A. Akbulut, R. V Guliyev, S. Celik, and M. N. Omarova, Fractional integral associated with Schrödinger operator on vanishing generalized Morrey spaces, J. Math. Inequal. 12 (2018), no. 3, 789–805.
- [4] A. Almeida, J. Hasanov, and S. Samko, Maximal and potential operators in variable exponent Morrey spaces, Georgian Math. J. 15 (2) (2008), 195–208.
- [5] M. F. de Almeida and L. S. M. Lima, Nonlinear boundary problem for Harmonic functions in higher dimensional Euclidean half-spaces, available at http://arxiv.org/abs/1807.04122v1.
- [6] K. F. Anderson and R. T. John, Weighted inequalities for vector-valued maximal functions and singular integrals, Studia Mathematica 69 (1980), 19–31.
- [7] N. Antonic and I. Ivec, On the Hörmander-Mihlin theorem for mixed-norm Lebesgue spaces, J. Math. Anal. Appl. 433 (2016), no. 1, 176–199.
- [8] R. Arai and E. Nakai, Commutators of Calderón-Zygmund and generalized fractional integral operators on generalized Morrey spaces, Rev. Mat. Complut. 31 (2018), 287–331.
- [9] M. Ariño and B. Muckenhoupt, Maximal functions on classical spaces and Hardy's inequality with weights for nonincreasing functions, Trans. Amer. Math. Soc. 320 (1990), 727–735.
- [10] R. J. Bagby, An extended inequality for the maximal function, Proc. Amer. Math. Soc. 48 (1975), 419–422.
- [11] A. Benedek and R. Panzone, The spaces L^P , with mixed norm, Duke Math. J. **28** (1961), 301–324.
- [12] C. Bennett and R. Sharpley, Interpolation of Operators, Academic Press, 1988.
- [13] O. V. Besov, On a family of function spaces. Embedding theorems and extensions, Dokl. Acad. Nauk SSSR 126 (1959), 1163–1165.

- [14] Z. W. Birnbaum and W. Orlicz, Über die Varallgemeinerung des Begriffes der zueinander konjugierten Potenzen, Studia Math. 3 (1931), 1–67.
- [15] A. Blozinski, Multivariate rearrangements and Banach function spaces with mixed norms, Trans. Amer. Math. Soc. 263 (1981), 149–167.
- [16] J.-M. Bony, Second microlocalization and propagation of singularities for semilinear hyperbolic equations, Hyperbolic equations and related topics (Katata/Kyoto, 1984), 11–49, Academic Press, Boston, MA, 1986.
- [17] A. P. Calderón, Intermediate spaces and interpolation, the complex method, Studia Math. 24 (1964), 113–190.
- [18] S. Chanillo, A note on commutators, Indiana Univ. Math. J. 31 (1982), No.1, 7–16.
- [19] F. Chiarenza and M. Frasca, Morrey spaces and Hardy-Littlewood maximal function, Rend. Mat. 7 (1987), 273–279.
- [20] R. R. Coifman, R. Rochberg, and G. Weiss, Factorization theorems for Hardy spaces in several variables, Ann. of Math. (2) 103 (1976), no. 3, 611–635.
- [21] D. Cruz-Uribe and L-A. D. Wang, Variable Hardy spaces, Indiana Univ. Math. J. 63 (2014), no. 2, 447–493.
- [22] L. Cui and Q. Yang, On the generalized Morrey spaces, Siberian Math J. 46 (2005), 133–141.
- [23] G. Di Fazio, D. I. Hakim, and Y. Sawano, *Elliptic equations with discontinuous coefficients in generalized Morrey spaces*, Eur. J. Math. 3 (2017), no.3, 728–762.
- [24] G. Di Fazio and M. A. Ragusa, Commutators and Morrey spaces, Boll. Un. Mat. Ital. A(7) 5 (1991), no.3, 323–332.
- [25] F. Deringoz, V. S. Guliyev and G. Hasanov, Maximal operator and its commutators on generalized weighted Orlicz-Morrey spaces, Tokyo J. Math. 41 (2018), no. 2, 347–369.
- [26] F. Deringoz, V. S. Guliyev and S. Samko, Boundedness of the maximal and singular operators on generalized Orlicz-Morrey spaces, Operator theory, operator algebras and applications, 139–158, Oper. Theory Adv. Appl., 242, Birkhäuser/Springer, Basel, 2014.
- [27] F. Deringoz, V. S. Guliyev, and S. Samko, Boundedness of the maximal operator and its commutators on vanishing generalized Orlicz-Morrey spaces, Ann. Acad. Sci. Fenn. Math. 40 (2015), no. 2, 535–549.
- [28] H. Dong and T. Phan, Mixed-norm L_p-estimates for non-stationary Stokes systems with singular VMO coefficients and applications, arXiv:1805.04143v2.

- [29] J. Duoandikoetxea, Fourier Analysis, Translated and revised from the 1995 Spanish original by D. Cruz-Uribe. Graduate Studies in Mathematics, 29. American Mathematical Society, Providence, RI, 2001.
- [30] Eridani, H. Gunawan and M. I. Utoyo, A characterization for fractional integral operators on generalized Morrey spaces, Anal. Theory Appl. 28 (2012), No. 3, 263–267.
- [31] Eridani, H. Gunawan, E. Nakai and Y. Sawano, Characterizations for the generalized fractional integral operators on Morrey spaces, Math. Ineq. and Appl. 17 (2014), no.2, 761–777.
- [32] A. Eroglu, M. N. Omarova, and M. A. Shemsiyye, *Elliptic equations with measurable coefficients in generalized weighted Morrey spaces*, Proc. Inst. Math. Mech. Natl. Acad. Sci. Azerb. 43 (2017), no.2, 197–213.
- [33] C. Fefferman and E. Stein, Some maximal inequalities, Amer. J. Math, 93 (1971), 107–115.
- [34] C. Fefferman and E. Stein, H^p spaces of several variables, Acta Math. 128 (1972), 137–193.
- [35] D. L. Fernandez, Vector-valued singular integral operators on L^p-spaces with mixed norms and applications, Pacific J. Math. **129** (1987), no. 2, 257–275.
- [36] L. C. F. Ferreira, On a bilinear estimate in weak-Morrey spaces and uniqueness for Navier-Stokes equations, J. Math. Pures Appl. 105 (2016), 228–247.
- [37] J. Fournier, Mixed norms and rearrangements: Sobolev's inequality and Littlewood's inequality, Ann. Math. Pura Appl. 148 (1987), 51–76.
- [38] N. Fujii, A proof of the Fefferman-Stein-Strömberg inequality for the sharp maximal functions, Proc. Amer. Soc. 106 (2) (1991), 371–377.
- [39] S. Gala, Y. Sawano and H. Tanaka, A remark on two generalized Orlicz-Morrey spaces, J. Approx. Theory 198 (2015), 1–9.
- [40] A. Gogatishvili and R. Mustafayev, New pre-dual space of Morrey space, J. Math. Anal. Appl. 397 (2013), no. 2, 678–692.
- [41] L. Grafakos, *Classical Fourier Analysis*, Graduate Texts in Mathematics 249, Springer, New York, 2008.
- [42] W. Grey and G. Sinnamon, The inclusion problem for mixed norm spaces, Trans. Amer. Math. Soc. 368 (2016), no. 12, 8715–8736.
- [43] V. S. Guliyev, Integral operators on function spaces on the homogeneous groups and on domains in \mathbb{R}^n (Russian), Doctor degree dissertation, Mat. Inst. Steklov, Moscow (1994).
- [44] V. S. Guliyev, Boundedness of the maximal, potential and singular operators in the generalized Morrey spaces, J. Inequal. Appl. 2009, Art.ID 503948, 20pp.

- [45] V. S. Guliyev, S. S. Aliyev, and T. Karaman, Boundedness of sublinear operators and commutators on generalized Morrey spaces, Abstr. Appl. Anal. 71 (2011), 327–355.
- [46] V. S. Guliyev and F. Deringoz, On the Riesz potential and its commutators on generalized Orlicz-Morrey spaces, J. Funct. Spaces 2014, Art. ID 617414, 11 pp.
- [47] V. S. Guliyev, F. Deringoz and J. J. Hasanov, (Φ, Ψ)-admissible potential operators and their commutators on vanishing Orlicz-Morrey spaces, Collect. Math. 67 (2016), no. 1, 133–153.
- [48] V. Guliyev, J. Hasanov, and S. Samko, Boundedness of the maximal, potential and singular integral operators in the generalized variable exponent Morrey spaces $M^{p(\cdot),\theta(\cdot),\omega(\cdot)}(\Omega)$, J. Math. Sci. **170** (4) (2010), 1–21.
- [49] V. Guliyev, J. Hasanov, and S. Samko, Boundedness of the maximal, potential and singular operators in the generalized variable exponent Morrey spaces, Math. Scand. 107 (2010), 285–304.
- [50] V. S. Guliyev, S. G. Hasanov, Y. Sawano and T. Noi, Non-smooth atomic decompositions for generalized Orlicz-Morrey spaces of the third kind, Acta Appl. Math. 145 (2016), 133–174.
- [51] V. S. Guliyev, S. Muradova, M. N. Omarova, and L. Softova, Gradient estimates for parabolic equations in generalized weighted Morrey spaces, Acta Math. Sin. (Engl. Ser.) 32 (2016), no.8, 911–924.
- [52] V. S. Guliyev and M. N. Omarova, Parabolic oblique derivative problem with discontinuous coefficients in generalized weighted Morrey spaces, Open Math. 14 (2016), 49–61.
- [53] D. I. Hakim and Y. Sawano, Interpolation of generalized Morrey spaces, Rev. Mat. Complut. 29 (2016), no.2, 487–514.
- [54] G. H. Hardy and J. E. Littlewood, A maximal theorem with function theoretic applications, Acta. Math, 54 (1930), 81–116.
- [55] N. Hatano, T. Nogayama and Y. Sawano, A remark on the paper "Vector-valued operators with singular kernel and Triebel-Lizorkin block spaces with variable exponents" by Kwok Pun Ho, to appear in Kyoto J. Math.
- [56] K.-P. Ho, Atomic decomposition of Hardy spaces and characterization of BMO via Banach function spaces, Anal. Math. 38 (2012), no. 3, 173–185.
- [57] K.-P. Ho, Strong maximal operator on mixed-norm spaces, Ann. Univ. Ferrara 62 (2016), 275–291.
- [58] K.-P. Ho, Vector-valued operators with singular kernel and Triebel-Lizorkin block spaces with variable exponents, Kyoto J. Math., 56 (2016), no. 1, 97–124.
- [59] K.-P. Ho, Mixed norm Lebesgue spaces with variable exponents and applications, Riv. Mat. Univ. Parma 9 (2018), 21–44.

- [60] K.-P. Ho, Y. Sawano, D. Yang and S. Yang, Hardy Spaces for Ball Quasi-Banach Function Spaces, Dissertationes Math. 525 (2017), 1–102.
- [61] R. A. Hunt, On L(p,q) spaces, Ensign. Math. 1 (1966), 249–276.
- [62] L. Huang and D. Yang, On function spaces with mixed norms—A survey, arXiv:1908.03291v1.
- [63] S. Igari, Interpolation of operators in Lebesgue spaces with mixed norm and its applications to Fourier analysis, Tohoku Math. J. 38 (1986), 469–490.
- [64] T. Iida, Weighted inequalities on Morrey spaces for linear and multilinear fractional integrals with homogeneous kernels, to appear in Taiwanese Math. J. 18 (2014), no. 1, 147–185.
- [65] T. Iida, Y. Komori-Furuya, and E. Sato, The Adams inequality on weighted Morrey spaces, Tokyo J. Math. 34 (2011), no. 2, 535–545.
- [66] T. Iida, Y. Sawano and H. Tanaka, Atomic decomposition for Morrey spaces, Z. Anal. Anwend. 33 (2014), no. 2, 149–170.
- [67] M. Izuki and Y. Sawano, Variable Lebesgue norm estimates for BMO functions, Czechoslovak Math. J. 62 (137) (2012), 717–727.
- [68] M. Izuki and Y. Sawano, Characterization of BMO via ball Banach function spaces, Vestn. St.-Peterbg. Univ. Mat. Mekh. Astron. 4 (62) (2017), 78–86.
- [69] M. Izuki, Y. Sawano, and Y. Tsutsui, Variable Lebesgue norm estimates for BMO functions. II, Anal. Math. 40 (3) (2014), 215–230.
- [70] S. Janson, Mean oscillation and commutators of singular integral operators, Ark. Mat. 16 (2) (1978), 263–270.
- [71] B. Jawerth, A. Torchinsky, Local sharp maximal functions, J. Approx. Theory 43 (1965), 231–270.
- [72] H. Jia and H. Wang, Decomposition of Hardy-Morrey spaces, J. Math. Anal. Appl. 354 (2009), no. 1, 99–110.
- [73] E. A. Kalita, *Dual Morrey spaces*, Dokl. Akad. Nauk **361** (1998), no. 4, 447–449.
- [74] H. Kempka, Atomic, molecular and wavelet decomposition of generalized 2microlocal Besov spaces, J. Funct. Spaces Appl. 8 (2010), 129–165.
- [75] H. Kita, On maximal functions in Orlicz spaces, Proc. Amer. Math. Soc. 124 (10) (1996), 3019–3025.
- [76] H. Kita, On a converse inequality for maximal functions in Orlicz spaces, Studia Math. 118 (1996), 1–10.
- [77] H. Kita, On Hardy-Littlewood maximal functions in Orlicz spaces, Math. Nachr. 183 (1997), 135–155.

- [78] V. M. Kokilashvili, A. Meskhi, H. Rafeiro, Estimates for nondivergence elliptic equations with VMO coefficients in generalized Morrey spaces, Complex Var. Elliptic Equ. 59 (2014), no.8, 1169–1184.
- [79] Y. Komori and T. Mizuhara, Factorization of functions in H¹(Rⁿ) and generalized Morrey spaces, Math. Nachr. 279 (2006), no.5–6, 619–624.
- [80] Y. Komori and S. Shirai, Weighted Morrey spaces and a singular integral operator, Math. Nachr. 282 (2) (2009), 219–231.
- [81] H. Kozono and M. Yamazaki, Semilinear heat equations and the Navier-Stokes equation with distributions in new function spaces as initial data, Comm Partial Differential Equations, 19 (1994), 959–1014.
- [82] P. G. Lemarié-Rieusset, The Navier-Stokes equations in the critical Morrey-Campanato space, Rev. Mat. Iberoam. 23 (2007), no. 3, 897–930.
- [83] P. G. Lemarié-Rieusset, Multipliers and Morrey spaces, Potential Anal. 38 (2013), no. 3, 741–752.
- [84] A. K. Lerner, A simple proof of the A₂ conjecture, Int. Math. Res. Not. 2013 (14) (2013), 3159–3170.
- [85] A. K. Lerner, F. Nazarov, Intuitive dyadic calculus: the basics, Expo. Math. 37 (2019), no. 3, 225–265.
- [86] L. Z. Liu, Interior estimates in Morrey spaces for solutions of elliptic equations and weighted boundedness for commutators of singular integral operators, Acta Math. Scientia. 25 (2005), 89–94.
- [87] R. L. Long, The spaces generated by blocks, Sci. Sinica Ser. A 27 (1984), no. 1, 16–26.
- [88] G. G. Lorentz, Some new function spaces, Ann. of Math. (2nd Ser.) 51 (1950), no.1, 37–55.
- [89] G. G. Lorentz, On the theory of spaces Λ , Pacific. J. Math. 1 (1951), 411–429.
- [90] L. Maligranda, Orlicz spaces and interpolation, Seminários de Mateática [Seminars in Mathematics], 5. Universidade Estadual de Campinas, Departamento de Matemática, Campinas, 1989. iii+206 pp.
- [91] L. Maligranda, Calderón-Lozanovskii construction for mixed norm spaces, Acta Math. Hungar. 103 (4) (2004), 279–302.
- [92] A. Mazzucato, Decomposition of Besov-Morrey spaces. Harmonic analysis at Mount Holyoke (South Hadley, MA, 2001), 279–294, Contemp. Math., 320, Amer. Math. Soc., Providence, RI, 2003.
- [93] T. Mizuhara, Boundedness of some classical operators on generalized Morrey spaces, Harmonic analysis (Sendai 1990), 183–189, ICM-90 Satell. Conf. Proc., Springer, Tokyo, 1991.

- [94] Y. Mizuta, E. Nakai, T. Ohno and T. Shimomura, *Riesz potentials and Sobolev embeddings on Morrey spaces of variable exponents*, Complex Var. Elliptic Equ. 56 (2011), no. 7-9, 671–695.
- [95] Y. Mizuta and T. Shimomura, Continuity properties for Riesz potentials of functions in Morrey spaces of variable exponent, Math. Inequal. Appl. 13 (2010), No. 1, 99–122.
- [96] S. Moritoh and T. Yamada, Two-microlocal Besov spaces and wavelets, Rev. Mat. Iberoamericana 20 (2004), no. 1, 277–283.
- [97] C. B. Morrey Jr., On the solutions of quasi-linear elliptic partial differential equations, Trans. Amer. Math. Soc. 43 (1938), no. 1, 126–166.
- [98] E. Nakai, Hardy-Littlewood maximal operator, singular integral operators and the Riesz potentials on generalized Morrey spaces, Math. Nachr. 166 (1994), 95–103.
- [99] E. Nakai, A characterization of pointwise multipliers on the Morrey spaces, Sci. Math. 3 (2000), 445–454.
- [100] E. Nakai, Generalized fractional integrals on Orlicz-Morrey spaces, Banach and function spaces, 323–333, Yokohama Publ., Yokohama, 2004.
- [101] E. Nakai, Orlicz-Morrey spaces and the Hardy-Littlewood maximal function, Studia Math. 188 (2008), 193–221.
- [102] E. Nakai and Y. Sawano, Hardy spaces with variable exponents and generalized Campanato spaces, J. Funct. Anal. 262 (2012), 3665–3748.
- [103] S. Nakamura, Generalized weighted Morrey spaces and classical operators, Math. Nachr. 289 (2016), no.17–18, 2235–2262.
- [104] S. Nakamura and Y. Sawano, The singular integral operator and its commutator on weighted Morrey spaces, Collect. Math. 68 (2017), 145–174.
- [105] H. Nakano, Modulared Semi-ordered Linear Spaces, Maruzen (Tokyo), 1950.
- [106] T. Nogayama, Mixed Morrey spaces, Positivity 23 (4) (2019), 961–1000.
- [107] T. Nogayama, Boundedness of the commutators of fractional integral operators on mixed Morrey spaces, Integral Transforms Spec. Funct. 30 (2019), no. 10, 790-816.
- [108] T. Nogayama, T. Ono, D. Salim and Y. Sawano, Atomic decomposition for mixed Morrey spaces, J. Geom. Anal. (2020), DOI:10.1007/s12220-020-00513-z
- [109] W. Orlicz, Über eien gewisse Klasse von Räumen vom Typus B, Bull. int'l de l'Acad. Pol. série A 8 (1932), 207–220.
- [110] W. Orlicz, *Über Räumen* (L^M) , Bull. int'l de l'Acad. Pol. série A **10** (1936), 93–107.

- [111] W. Orlicz, Some classes of modular spaces, Studia Math. 26 (1966), 165–192.
- [112] T. Phan, Well-posedness for the Navier-Stokes equations in critical mixed-norm Lebesgue spaces, J. Evol. Equ. 20 (2020), 553–576.
- [113] T. Phan, Liouville type theorems for 3D stationary Navier-Stokes equations in weighted mixed-norm Lebesgue spaces, Dyn. Partial Differ. Equ. 17 (2020), no. 3, 229–243.
- [114] J. Peetre, On the theory of $\mathcal{L}_{p,\lambda}$ spaces, J. Funct. Anal. 4 (1969), 71–87.
- [115] J. Peetre, Sur le espaces de Besov., C. R. Acad. Sci. Paris Sér. A-B 264 (1967), 281–283.
- [116] J. Peetre, New Thoughts on Besov spaces, Duke Univ. Math. Ser. 1, Math. Dept., Duke Univ., Durham, NC, 1976.
- [117] L. Pick, A. Kufner, J. Oldřich, and S. Fučík, Function spaces. Vol. 1. Second revised and extended edition, De Gruyter Series in Nonlinear Analysis and Applications, 14. Walter de Gruyter & Co., Berlin, 2013. xvi+479 pp.
- [118] M. A. Ragusa, Embeddings for Morrey-Lorentz Spaces, J. Optim. Theory Appl., 154 (2012), 491–499.
- [119] M. M. Rao and Z. D. Ren, Theory of Orlicz Spaces, Dekker 1991.
- [120] M. Rosenthal and H. J. Schmeisser, On the Boundedness of Singular Integral in Morrey spaces and its Preduals, J. Fourier Anal. Appl. 22 (2016), no. 2, 462–490.
- [121] M. Rosenthal and H. Triebel, Calderon-Zygmund operators in Morrey spaces, Rev. Mat. Complut. 27 (2014), no. 1, 1–11.
- [122] M. Rosenthal and H. Triebel, Morrey spaces, their duals and preduals, Rev. Mat. Complut. 28 (2015), 1–30.
- [123] N. Samko, On a Muckenhoupt-type condition for Morrey spaces, Mediterr. J. Math. 10 (2013), 941–951.
- [124] Y. Sawano, Morrey spaces with non-doubling measures. II, J. Indones. Math. Soc. 14 (2008), no. 2, 121–152.
- [125] Y. Sawano, A note on Besov-Morrey spaces and Triebel-Lizorkin-Morrey spaces, Acta Math. Sin. (Engl. Ser.) 25 (2009), 1223–1242.
- [126] Y. Sawano, *Theory of Besov spaces*, Development in Mathematics 56 (2018).
- [127] Y. Sawano, A thought on generalized Morrey spaces, J. Indones. Math. Soc. 25 (2019), no. 3, 210–281.
- [128] Y. Sawano, Homogeneous Besov spaces, Kyoto J. Math. 60 (2020), no. 1, 1–43.
- [129] Y. Sawano and S. Shirai, Compact commutators on Morrey spaces with nondoubling measures, Georgian Math. J. 15 (2) (2008), 353–376.

- [130] Y. Sawano and H. Tanaka, Morrey spaces for non-doubling measures, Acta Math. Sinica, 21 (2005), no. 6, 1535–1544.
- [131] Y. Sawano and H. Tanaka, Decompositions of Besov-Morrey spaces and Triebel-Lizorkin-Morrey spaces, Math. Z. 257 (2007), no. 4, 871–905.
- [132] Y. Sawano and H. Tanaka, Sharp maximal inequalities and commutators on Morrey spaces with non-doubling measures, Taiwanese J. Math. 11 (4) (2007), 1091– 1112.
- [133] Y. Sawano, S. Sugano and H. Tanaka, A note on generalized fractional integral operators on generalized Morrey spaces, Bound. Value Probl. 2009, Art. ID 835865, 18 pp.
- [134] Y. Sawano, S. Sugano and H. Tanaka, Generalized fractional integral operators and fractional maximal operators in the framework of Morrey spaces, Trans. Amer. Math. Soc. 363 (2011), no. 12, 6481–6503.
- [135] Y. Sawano, S. Sugano and H. Tanaka, Orlicz-Morrey spaces and fractional operators, Potential Anal. 36 (2012), no. 4, 517–556.
- [136] E. Sawyer, Boundedness of classical operators on classical Lorentz spaces, Studia Math. 96 (1990), 145–158.
- [137] A. Scapellato and M. A. Ragusa, Mixed Morrey spaces and their applications to partial differential equations, Nonlinear Analysis, 151 (2017), 51–65.
- [138] S. G. Shi, Z. Fu, and F. Zhao, Estimates for operators on weighted Morrey spaces and their applications to nondivergence elliptic equations, J. Inequal. Appl. 2013, 2013;390.
- [139] S. G. Shi and F. Zhao, Necessary and sufficient conditions for boundedness of commutators of the general fractional integral operators on weighted Morrey spaces, Abstr. Appl. Anal. 2012, Art.ID 929381, 14pp.
- [140] S. Shirai, Necessary and sufficient conditions for boundedness of commutators of fractional integral operators on classical Morrey spaces, Hokkaido Math. J. 35
 (3) (2006), 683–696.
- [141] E. M. Stein, Harmonic analysis: real-variable methods, orthogonality, and oscillatory integrals, Princeton Mathematical Series, vol. 43, Princeton University Press, Princeton, NJ, 1993.
- [142] A. Stefanov and R. H. Torres, Calderón-Zygmund operators on mixed Lebesgue spaces and applications to null forms, J. London Math. Soc. (2) 70 (2004), no. 2, 447–462.
- [143] E. M. Stein and G. Weiss Introduction to FOURIER ANALYSIS ON EU-CLIDEAN SPACES, Prinston (1975).
- [144] B. Stöckert, Ungleichungen von Plancherel-Polya-Nikol'skij typ in gewichteten L_n^{Ω} Räumen mit gemischten Norm, Math. Nach. **86** (1978), 19–32.
- [145] M. H. Taibleson, G. Weiss, Certain function spaces connected with almost everywhere convergence of Fourier series, Conference on harmonic analysis in honor of Antoni Zygmund, Vol. I, II (Chicago, Ill., 1981), 95–113, Wadsworth Math. Ser., Wadsworth, Belmont, CA, 1983.
- [146] L. Tang and J. Xu, Some properties of Morrey type Besov-Triebel spaces, Math. Nachr. 278 (2005), 904–917.
- [147] H. Triebel, *Theory of Function Spaces*, Birkhäuser (1983).
- [148] H. Triebel, Theory of Function Spaces II, Birkhäuser (1992).
- [149] H. Triebel, Fractal and Spectra, Birkhäuser (1997).
- [150] W. Yuan, W. Sickel and D. Yang, Morrey and Campanato Meet Besov, Lizorkin and Triebel, Lecture Notes in Mathematics, 2005, Springer-Verlag, Berlin, 2010, xi+281 pp.
- [151] X. F. Ye and T. F. Wang, Two-weighted norm inequality on weighted Morrey spaces, Turkish J. Math. 38 (2014), no. 3, 426–435.
- [152] C. T. Zorko, Morrey space, Proc. Amer. Math. Soc. 98 (1986), 586–592.